As filed with the Securities and Exchange Commission on March 19, 2021.

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

AMENDMENT NO. 2

TO

FORM S-1

REGISTRATION STATEMENT

Under

The Securities Act of 1933

Frontier Group Holdings, Inc.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization) 4512 (Primary Standard Industrial Classification Code Number) 46-3681866 (I.R.S. Employer Identification Number)

4545 Airport Way Denver, CO 80239 (720) 374-4200

(Address, including zip code, and telephone number, including area code, of registrant's principal executive offices)

Barry L. Biffle President and Chief Executive Officer Frontier Group Holdings, Inc. 4545 Airport Way Denver, CO 80239 (720) 374-4200

(Name, address, including zip code, and telephone number, including area code, of agent for service)

Copies to:

Anthony J. Richmond Miles P. Jennings Brian D. Paulson Latham & Watkins LLP 140 Scott Drive Menlo Park, CA 94025 Telephone: (650) 328-4600 Facsimile: (650) 463-2600 Howard M. Diamond Senior Vice President, General Counsel & Secretary Frontier Group Holdings, Inc. 4545 Airport Way Denver, CO 80239 Telephone: (720) 374-4200

Alan F. Denenberg Stephen Salmon Davis Polk & Wardwell LLP 1600 El Camino Real Menlo Park, CA 94025 Telephone: (650) 752-2010 Facsimile: (650) 752-2115

Approximate date of commencement of proposed sale to the public: As soon as practicable after this Registration Statement becomes effective.

If any of the securities being registered on this Form are to be offered on a delayed or continuous basis pursuant to Rule 415 under the Securities Act of 1933, as amended, check the following box.

If this Form is filed to register additional securities for an offering pursuant to Rule 462(b) under the Securities Act, please check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

If this Form is a post-effective amendment filed pursuant to Rule 462(c) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

If this Form is a post-effective amendment filed pursuant to Rule 462(d) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer	Accelerated filer	
Non-accelerated filer	Smaller reporting company	

Emerging growth company

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 7(a)(2)(B) of the Securities Act.

CALCULATION OF REGISTRATION FEE

	PROPOSED	
	MAXIMUM	
TITLE OF EACH CLASS OF	AGGREGATE	AMOUNT OF
SECURITIES TO BE REGISTERED	OFFERING PRICE(1)	REGISTRATION FEE(2)(3)
Common Stock, \$0.001 par value per share	\$100,000,000	\$10,910

(1) Estimated solely for the purpose of calculating the amount of the registration fee in accordance with Rule 457(o) under the Securities Act of 1933, as amended. Includes the aggregate offering price of additional shares that the underwriters have the option to purchase.

(2) Calculated pursuant to Rule 457(o) based on an estimate of the proposed maximum aggregate offering price.
 (3) Previously paid.

The Registrant hereby amends this Registration Statement on such date or dates as may be necessary to delay its effective date until the Registrant shall file a further amendment which specifically states that this Registration Statement shall thereafter become effective in accordance with Section 8(a) of the Securities Act of 1933, as amended, or until the Registration Statement shall become effective on such date as the Commission, acting pursuant to said Section 8(a), may determine.

EXPLANATORY NOTE

This Amendment No. 2 (the "Amendment No. 2") to the Registration Statement on Form S-1 (File No. 333-254004) of Frontier Group Holdings, Inc. (the "Registration Statement") is being filed as an exhibits-only filing. Accordingly, this Amendment No. 2 consists only of the facing page, this explanatory note, Part II of the Registration Statement, the signature page to the Registration Statement and the filed exhibits. The remainder of the Registration Statement is unchanged and has therefore been omitted.

PART II

Item 13. Other Expenses of Issuance and Distribution

The following table sets forth the costs and expenses, other than underwriting discounts, payable in connection with the sale and distribution of the securities being registered. All amounts are estimated except the SEC registration fee and the FINRA filing fee. All the expenses below will be paid by Frontier Group Holdings, Inc.

Item	Amount
SEC Registration fee	*
FINRA filing fee	*
Initial Nasdaq Stock Market listing fee	*
Legal fees and expenses	*
Accounting fees and expenses	*
Printing and engraving expenses	*
Transfer Agent and Registrar fees	*
Blue Sky fees and expenses	*
Miscellaneous fees and expenses	*
Total	*

* To be completed by amendment.

Item 14. Indemnification of Directors and Officers

Frontier Group Holdings, Inc., Inc. is a Delaware corporation. Section 145 of the Delaware General Corporation Law authorizes a court to award, or a corporation's board of directors to grant, indemnity to directors and officers in terms sufficiently broad to permit such indemnification under certain circumstances for liabilities, including reimbursement for expenses incurred, arising under the Securities Act of 1933, as amended. Our amended and restated certificate of incorporation to be in effect immediately prior to the consummation of this offering compels indemnification of our directors and officers and permits indemnification of our employees and other agents, in each case to the maximum extent permitted by the Delaware General Corporation Law, and our amended and restated bylaws to be in effect immediately prior to the consummation of this offering provide for indemnification of our directors, officers, employees and other agents to the maximum extent permitted by the Delaware General Corporation Law. In addition, we have entered into indemnification agreements with our directors, officers and some employees containing provisions which are in some respects broader than the specific indemnification provisions contained in the Delaware General Corporation Law. The indemnification agreements may require us, among other things, to indemnify our directors against certain liabilities that may arise by reason of their status or service as directors and to advance their expenses incurred as a result of any proceeding against them as to which they could be indemnified. Reference is also made to Section 8 of the underwriting agreement to be filed as Exhibit 1.1 hereto, which provides for indemnification by the underwriter of our officers and directors against certain liabilities.

Item 15. Recent Sales of Unregistered Securities

Since January 1, 2018, we have granted equity awards for an aggregate of 116,440 shares of our common stock to employees and directors under our 2014 Equity Incentive Plan, which includes 2,416 shares that were subsequently forfeited and 262 shares that were subsequently repurchased.

The sales of the above securities were deemed to be exempt from registration under the Securities Act in reliance upon Rule 701 promulgated under Section 3(b) of the Securities Act as transactions by an issuer not involving any public offering or pursuant to benefit plans and contracts relating to compensation as provided

under Rule 701. The recipients of the securities in each of these transactions represented their intentions to acquire the securities for investment only and not with a view to or for sale in connection with any distribution thereof, and appropriate legends were placed upon the stock certificates issued in these transactions. All recipients had adequate access, through their relationships with us, to information about us.

Between May 19, 2020 and January 15, 2021, we issued to the United States Department of Treasury (the "Treasury") warrants to purchase an aggregate of 75,807 shares of common stock at an exercise price of \$241.72 per share in connection with our Payroll Support Program Agreement and the loan we received from the Treasury. Additionally, by operation of the Payroll Support Program Extension Agreement entered into with the Treasury on January 15, 2021, we issued additional warrants to the Treasury to purchase an aggregate of 2,716 shares of common stock at an exercise price of \$442.69 per share based on \$140 million of funding pursuant to the so-called "PSP2" program.

The offers, sales, and issuances of the securities described above were deemed to be exempt from registration under Section 4(a)(2) of the Securities Act as a transaction by an issuer not involving a public offering. The recipients of securities in each of these transactions acquired the securities for investment only and not with a view to or for sale in connection with any distribution thereof and appropriate legends were affixed to the securities issued in these transactions.

There were no underwriters employed in connection with any of the transactions set forth in Item 15.

EXHIBIT INDEX

		Incorporated By Reference			
Exhibit Number	Exhibit Description	Form	Date	Number	Filed Herewith
1.1*	Form of Underwriting Agreement.				
3.1	Amended and Restated Certificate of Incorporation, currently in effect.				Х
3.2	Form of Amended and Restated Certificate of Incorporation, to be in effect immediately prior to the consummation of this offering.				Х
3.3	Amended and Restated Bylaws, currently in effect.	S-1	3/8/2021	3.3	
3.4	Form of Amended and Restated Bylaws, to be in effect immediately prior to the consummation of this offering.				Х
4.1	Reference is made to exhibits 3.1, 3.2, 3.3 and 3.4.				
4.2	Form of Common Stock Certificate Award.	S-1	3/8/2021	4.2	
4.3	Registration Rights Agreement, to be in effect immediately prior to the consummation of this offering, by and among Frontier Group Holdings, Inc. and Indigo Frontier Holdings Company, LLC.				Х
5.1*	Opinion of Latham & Watkins LLP.				
10.1	<u>Airport Use and Lease Agreement, dated as of February 1, 2021, by and between</u> <u>Frontier Airlines, Inc. and the City and County of Denver.</u>	S-1	3/8/2021	10.1	
10.2(a)#	2014 Equity Incentive Plan.	S-1	3/8/2021	10.2(a)	
10.2(b)#	Form of Stock Option Grant Notice and Stock Option Agreement under the 2014 Equity Incentive Plan.	S-1	3/8/2021	10.2(b)	
10.2(c)#	Form of Stock Purchase Right Grant Notice and Restricted Stock Purchase Agreement for Non-Employee Directors.	S-1	3/8/2021	10.2(c)	
10.2(d)#	Form of Restricted Stock Unit Award Grant Notice and RSU Award Agreement under the 2014 Equity Incentive Plan.	S-1	3/8/2021	10.2(d)	
10.3(a)#	2021 Incentive Award Plan.				Х
10.3(b)#	Form of Stock Option Grant Notice and Stock Option Agreement under the 2021 Incentive Award Plan.				Х
10.3(c)#	Form of Restricted Stock Award Grant Notice and Restricted Stock Award Agreement under the 2021 Incentive Award Plan.				Х
10.3(d)#	Form of Restricted Stock Unit Award Grant Notice and Restricted Stock Unit Award Agreement under the 2021 Incentive Award Plan.				Х
10.4#	Form of Indemnification Agreement for directors and officers.	S-1	3/8/2021	10.4	
10.5#	<u>Employment Agreement, dated as of March 15, 2016, by and between Frontier Airlines,</u> <u>Inc. and Barry L. Biffle.</u>	S-1	3/8/2021	10.5	
10.6#	Amended and Restated Employment Agreement, dated as of April 13, 2017, by and between Frontier Airlines, Inc. and James G. Dempsey.	S-1	3/8/2021	10.6	
10.7#	Employment Agreement, dated as of June 1, 2017, by and between Frontier Airlines, Inc. and Jake F. Filene.	S-1	3/8/2021	10.7	

		Incorporated By Reference			
Exhibit Number	Exhibit Description	Form	Date	Number	Filed Herewith
10.8#	Employment Letter, dated as of June 30, 2014, by and between Frontier Airlines, Inc. and Howard M. Diamond.	S-1	3/8/2021	10.8	
10.9#	Employment Letter, dated as of September 2, 2015, by and between Frontier Airlines, Inc. and Mark C. Mitchell.	S-1	3/8/2021	10.9	
10.10(a)#	Employment Agreement, dated as of June 25, 2012, by and between Frontier Airlines, Inc. and Daniel M. Shurz.	S-1	3/8/2021	10.10(a)	
10.10(b)#	Amendment to Employment Agreement, dated as of September 13, 2013, by and between Frontier Airlines, Inc. and Daniel M. Shurz.	S-1	3/8/2021	10.10(b)	
10.11#	Employment Agreement, dated as of February 13, 2019, by and between Frontier Airlines, Inc. and Trevor J. Stedke.	S-1	3/8/2021	10.11	
10.12#	Non-Employee Director Compensation Program.				Х
10.13(a)#	Amended and Restated Phantom Equity Investment Agreement, dated as of December 3, 2013, by and among Frontier Airlines, Inc., Falcon Acquisition Group, Inc. and FAPAInvest, LLC.	S-1	3/8/2021	10.13(a)	
10.13(b)#	Amendment to Amended and Restated Phantom Equity Investment Agreement, dated as of December 20, 2016, by and among Frontier Airlines, Inc., Frontier Group Holdings, Inc. and FAPAInvest, LLC.	S-1	3/8/2021	10.13(b)	
10.13(c)#	Second Amendment to Amended and Restated Phantom Equity Investment Agreement, dated as of December 27, 2019, by and among Frontier Airlines, Inc., Frontier Group Holdings, Inc. and FAPAInvest, LLC.	S-1	3/8/2021	10.13(c)	
10.14#	<u>Professional Services Agreement, dated December 3, 2013, by and among Indigo Partners</u> <u>LLC, Frontier Airlines Holdings, Inc. and Frontier Airlines, Inc.</u>	S-1	3/8/2021	10.14	
10.15#	Subscription Agreement, dated as of December 3, 2013, by and between Falcon Acquisition Group, Inc. and Indigo Frontier Holdings Company, LLC.	S-1	3/8/2021	10.15	
10.16(a)†	Airbus A320 Family Aircraft Purchase Agreement, dated as of September 30, 2011, by and between Airbus S.A.S. and Republic Airways Holdings Inc.	S-1	3/8/2021	10.16(a)	
10.16(b)†	Amended and Restated Letter Agreement No. 1, dated as of December 28, 2017, by and between Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.16(b)	
10.16(c)†	Second Amended and Restated Letter Agreement No. 2, dated as of October 9, 2019, by and between Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.16(c)	
10.16(d)†	Second Amended and Restated Letter Agreement No. 3, dated as of October 9, 2019, by and between Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.16(d)	
10.16(e)†	Amended and Restated Letter Agreement No. 4, dated as of December 28, 2017, by and between Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.16(e)	

Exhibit		Incorp	oorated By Ref	erence	Filed
Number	Exhibit Description	Form	Date	Number	Herewith
10.16(f)†	Letter Agreement No. 5, dated as of September 30, 2011, by and between Airbus S.A.S. and Republic Airways Holdings Inc.	S-1	3/8/2021	10.16(f)	
10.16(g)†	Letter Agreement No. 6A, dated as of September 30, 2011, by and between Airbus S.A.S. and Republic Airways Holdings Inc.	S-1	3/8/2021	10.16(g)	
10.16(h)†	Letter Agreement No. 6B, dated as of September 30, 2011, by and between Airbus S.A.S. and Republic Airways Holdings Inc.	S-1	3/8/2021	10.16(h)	
10.16(i)†	Letter Agreement No. 6D, dated as of October 9, 2019, by and between Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.16(i)	
10.16(j)†	Letter Agreement No. 6D-1, dated as of October 9, 2019, by and between and Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.16(j)	
10.16(k)†	Letter Agreement No. 6D-2, dated as of October 9, 2019, by and between and Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.16(k)	
10.16(l)†	Letter Agreement No. 6D-3, dated as of October 9, 2019, by and between and Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.16(l)	
10.16(m)†	Amended and Restated Letter Agreement No. 7, dated as of December 28, 2017, by and between Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.16(m)	
10.16(n)†	Letter Agreement No. 8, dated as of September 30, 2011, by and between Airbus S.A.S. and Republic Airways Holdings Inc.	S-1	3/8/2021	10.16(n)	
10.16(o)†	Letter Agreement No. 9, dated as of September 30, 2011, by and between Airbus S.A.S. and Republic Airways Holdings Inc.	S-1	3/8/2021	10.16(o)	
10.16(p)†	<u>Amended and Restated Letter Agreement No. 10, dated as of October 9, 2019, by and between Airbus S.A.S. and Frontier Airlines, Inc.</u>	S-1	3/8/2021	10.16(p)	
10.16(q)†	Amendment No. 1 to Airbus A320 Family Aircraft Purchase Agreement, dated as of January 10, 2013, by and between Airbus S.A.S. and Republic Airways Holdings Inc.	S-1	3/8/2021	10.16(q)	
10.16(r)†	<u>Amendment No. 2 to Airbus A320 Family Aircraft Purchase Agreement, dated as of</u> December <u>3, 2013, by and between Airbus S.A.S. and Frontier Airlines, Inc.</u>	S-1	3/8/2021	10.16(r)	
10.16(s)†	Amendment No. 3 to Airbus A320 Family Aircraft Purchase Agreement, dated as of October 31, 2014, by and between Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.16(s)	
10.16(t)†	Amendment No. 4 to Airbus A320 Family Aircraft Purchase Agreement, dated as of August 7, 2017, by and between Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.16(t)	
10.16(u)†	<u>Amendment No. 5 to Airbus A320 Family Aircraft Purchase Agreement, dated as of</u> <u>December 28, 2017, by and between Airbus S.A.S. and Frontier Airlines, Inc.</u>	S-1	3/8/2021	10.16(u)	
10.16(v)†	Amendment No. 6 to Airbus A320 Family Aircraft Purchase Agreement, dated as of July 1, 2019, by and between Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.16(v)	

j	Incorporated By Reference				
Exhibit Number	Exhibit Description	Form	Date	Number	Filed Herewith
10.16(w)†	Amendment No. 7 to Airbus A320 Family Aircraft Purchase Agreement, dated as of October 9, 2019, by and between Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.16(w)	
10.16(x)†	Amendment No. 8 to Airbus A320 Family Aircraft Purchase Agreement, dated as of March 16, 2020, by and between Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.16(x)	
10.16(y)†	Amendment No. 9 to Airbus A320 Family Aircraft Purchase Agreement, dated as of May 4, 2020, by and between Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.16(y)	
10.16(z)†	Amendment No. 10 to Airbus A320 Family Aircraft Purchase Agreement, dated as of December 2, 2020, by and between Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.16(z)	
10.16(aa)†	Letter Agreement, dated as of December 28, 2017, by and between Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.16(aa)	
10.17(a)†	<u>Airbus A321 Aircraft Purchase Agreement, dated as of October 31, 2014, by and between</u> <u>Airbus S.A.S. and Frontier Airlines, Inc.</u>	S-1	3/8/2021	10.17(a)	
10.17(b)†	Letter Agreement No. 5, dated as of October 31, 2014, by and between Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.17(b)	
10.17(c)†	Letter Agreement No. 6A, dated as of October 31, 2014, by and between Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.17(c)	
10.17(d)†	Letter Agreement No. 6B, dated as of October 31, 2014, by and between Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.17(d)	
10.17(e)†	Letter Agreement No. 8, dated as of October 31, 2014, by and between Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.17(e)	
10.17(f)†	Letter Agreement No. 9, dated as of October 31, 2014, by and between Airbus S.A.S. and Frontier Airlines, Inc.	S-1	3/8/2021	10.17(f)	
10.18†	Amended and Restated Frontier Airlines, Inc. Credit Card Affinity Agreement, dated as of September 15, 2020, by and between Frontier Airlines, Inc. and Barclays Bank Delaware, formerly known as Juniper Bank.	S-1/A	3/18/2021	10.18	
10.19(a)†	<u>General Terms Agreement No. 6-13616, dated as of June 30, 2000, by and between</u> <u>Frontier Airlines, Inc., CFM International, Inc. and Societe Nationale D'Etude et de</u> <u>Construction de Monteurs d'Aviation.</u>	S-1	3/8/2021	10.19(a)	
10.19(b)†	Letter Agreement No. 1, dated as of June 30, 2000, by and between Frontier Airlines, Inc. and CFM International, Inc.	S-1	3/8/2021	10.19(b)	
10.19(c)†	Letter Agreement No. 2, dated as of November 20, 2002, by and between Frontier Airlines, Inc. and CFM International, Inc.	S-1	3/8/2021	10.19(c)	
10.19(d)†	Letter Agreement No. 3, dated as of August 1, 2003, by and between Frontier Airlines, Inc. and CFM International, Inc.	S-1	3/8/2021	10.19(d)	
10.19(e)†	Letter Agreement No. 4, dated as of March 26, 2004, by and between Frontier Airlines, Inc. and CFM International, Inc.	S-1	3/8/2021	10.19(e)	

Exhibit		Incorp	orated By Ref	erence	Filed
Number	Exhibit Description	Form	Date	Number	Herewith
10.19(f)†	Letter Agreement No. 5, dated as of April 11, 2006, by and between Frontier Airlines, Inc. and CFM International, Inc.	S-1	3/8/2021	10.19(f)	
10.19(g)	Amendment No. 1 to GTA 6-13616, dated as of June 6, 2009, by and between Frontier Airlines, Inc. and CFM International, Inc.	S-1	3/8/2021	10.19(g)	
10.19(h)†	Letter Agreement No. 7, dated as of October 25, 2011, by and between Frontier Airlines, Inc. and CFM International, Inc.	S-1	3/8/2021	10.19(h)	
10.19(i)†	Letter Agreement No. 8, dated as of December 23, 2014, by and between Frontier Airlines, Inc. and CFM International, Inc.	S-1	3/8/2021	10.19(i)	
10.19(j)†	Letter Agreement No. 9, dated as of August 3, 2015, by and between Frontier Airlines, Inc. and CFM International, Inc.	S-1	3/8/2021	10.19(j)	
10.20(a)†	<u>General Terms Agreement No. CFM-1 1-2576101711, dated as of October 17, 2011, by and between Frontier Airlines, Inc. and CFM International, Inc.</u>	S-1	3/8/2021	10.20(a)	
10.20(b)†	Letter Agreement No. 1 to General Terms Agreement No. CFM-1 1-2576101711, dated as of October 26, 2011, by and between Frontier Airlines, Inc. and CFM International, Inc.	S-1	3/8/2021	10.20(b)	
10.20(c)†	Amendment No. 1 to Letter Agreement No. 1, dated as of December 23, 2014, by and between Frontier Airlines, Inc. and CFM International, Inc.	S-1	3/8/2021	10.20(c)	
10.21†	<u>Purchase Terms Agreement (Material-Single Event), dated as of November 5, 2014, by and between Frontier Airlines, Inc. and Lufthansa Technik AG.</u>	S-1	3/8/2021	10.21	
10.22(a)†	<u>Navitaire Hosted Services Agreement, dated as of June 20, 2014, by and between Frontier</u> <u>Airlines, Inc. and Navitaire LLC.</u>	S-1	3/8/2021	10.22(a)	
10.22(b)†	Amendment No. 1 to Navitaire Hosted Services Agreement, dated as of March 1, 2015, by and between Frontier Airlines, Inc. and Navitaire LLC.	S-1	3/8/2021	10.22(b)	
10.22(c)†	Amendment No. 2 to Navitaire Hosted Services Agreement, dated as of April 10, 2015, by and between Frontier Airlines, Inc. and Navitaire LLC.	S-1	3/8/2021	10.22(c)	
10.22(d)†	Amendment No. 3 to Navitaire Hosted Services Agreement, dated as of January 1, 2016, by and between Frontier Airlines, Inc. and Navitaire LLC.	S-1	3/8/2021	10.22(d)	
10.23†	Sixth Amended and Restated Credit Agreement, dated as of December 22, 2020, by and among Vertical Horizons, Ltd., Citibank, N.A., Citigroup Global Markets, Inc., Bank of Utah and each lender identified on Schedule I thereto.	S-1	3/8/2021	10.23	
10.24(a)	<u>Guarantee, dated as of December 23, 2014, by Frontier Airlines Holdings, Inc. for the benefit</u> of Airbus S.A.S.	S-1	3/8/2021	10.24(a)	
10.24(b)	<u>Guarantee, dated as of December 23, 2014, by Frontier Airlines, Inc. for the benefit of Airbus</u> <u>S.A.S.</u>	S-1	3/8/2021	10.24(b)	
10.25	Sixth Amended and Restated Guarantee, dated as of December 22, 2020, by Frontier Airlines, Inc. in favor of Bank of Utah.	S-1	3/8/2021	10.25	

Exhibit		Incorp	orated By Ref	erence	Filed
Number	Exhibit Description	Form	Date	Number	Herewith
10.26†	Sixth Amended and Restated Guarantee, dated as of December 22, 2020, by Frontier Airlines Holdings, Inc. in favor of Bank of Utah.	S-1	3/8/2021	10.26	
10.27(a)†	<u>Amended and Restated Step-In Agreement, dated as of March 19, 2020, by and among</u> <u>Vertical Horizons, Ltd., Bank of Utah and Airbus S.A.S.</u>	S-1	3/8/2021	10.27(a)	
10.27(b)†	Amendment to Amended and Restated Step-In Agreement, dated as of May 4, 2020, by and among Vertical Horizons, Ltd., Bank of Utah and Airbus S.A.S.	S-1	3/8/2021	10.27(b)	
10.27(c)†	Amendment to Amended and Restated Step-In Agreement, dated as of December 15, 2020, by and among Vertical Horizons, Ltd., Bank of Utah and Airbus S.A.S.	S-1	3/8/2021	10.27(c)	
10.28	<u>Subordinated Loan Agreement, dated as of December 23, 2014, by and between Frontier</u> <u>Airlines, Inc. and Vertical Horizons, Ltd.</u>	S-1	3/8/2021	10.28	
10.29†	Fifth Amended and Restated CFMI Engine Benefits Agreement, dated as of March 19, 2020, by and among Vertical Horizons, Ltd., CFM International, Inc., Bank of Utah and Frontier Airlines, Inc.	S-1	3/8/2021	10.29	
10.30(a)†	Amended and Restated Signatory Agreement (U.S. Visa and MasterCard Transactions), dated as of November 5, 2013, by and among Frontier Airlines Holdings Inc., Frontier Airlines, Inc. and U.S. Bank National Association.	S-1	3/8/2021	10.30(a)	
10.30(b)†	First Omnibus Amendment to Signatory Agreements, dated as of March 1, 2016, by and among Frontier Airlines Holdings, Inc., Frontier Airlines, Inc., U.S. Bank National Association, and Elavon Canada Company.	S-1	3/8/2021	10.30(b)	
10.30(c)†	Third Omnibus Amendment to Signatory Agreements, dated as May 1, 2018, by and among Frontier Airlines Holdings, Inc., Frontier Airlines, Inc., U.S. Bank National Association and Elavon Canada Company.	S-1	3/8/2021	10.30(c)	
10.30(d)†	Fourth Omnibus Amendment to Signatory Agreements, dated as of April 1, 2020, by and among Frontier Airlines Holdings, Inc., Frontier Airlines, Inc., U.S. Bank National Association and Elavon Canada Company.	S-1	3/8/2021	10.30(d)	
10.31(a)†	<u>Rate per Flight Hour Agreement, dated as of August 29, 2017, by and between CFM</u> International, Inc. and Frontier Airlines, Inc.	S-1	3/8/2021	10.31(a)	
10.31(b)†	Amendment No. 1 to Rate per Flight Hour Agreement, dated as of August 29, 2017, by and between CFM International, Inc. and Frontier Airlines, Inc.	S-1	3/8/2021	10.31(b)	
10.32†	<u>Codeshare Agreement, dated as of January 16, 2018, by and between Concesionaria Vuela</u> <u>Compania Aviacion S.A.P.I. de C.V. and Frontier Airlines, Inc.</u>	S-1	3/8/2021	10.32	
10.33	<u>Promissory Note, dated as of April 30, 2020, issued by Frontier Group Holdings, Inc. in the name of the United States Department of the Treasury and guaranteed by Frontier Airlines, Inc. and Frontier Airlines Holdings, Inc.</u>	S-1	3/8/2021	10.33	

it		Incorp	orated By Ref	erence	Filed
it er	Exhibit Description	Form	Date	Number	Herewith
	Payroll Support Program Agreement, dated as of April 30, 2020, between Frontier Airlines, Inc. and the United States Department of the Treasury.	S-1	3/8/2021	10.34	
	Warrant Agreement, dated as of April 30, 2020, between Frontier Group Holdings, Inc. and the United States Department of the Treasury.	S-1	3/8/2021	10.35	
	Form of Warrant (incorporated by reference to Annex B of Exhibit 10.35).	S-1	3/8/2021	10.36	
	Loan and Guarantee Agreement, dated as of September 28, 2020, by and among Frontier Airlines, Inc., Frontier Group Holdings, Inc., the United States Department of the Treasury and The Bank of New York Mellon, as administrative and collateral agent.	S-1	3/8/2021	10.37	
	Warrant Agreement, dated as of September 28, 2020, between Frontier Group Holdings, Inc. and the United States Department of the Treasury.	S-1	3/8/2021	10.38	
	Form of Warrant (incorporated by reference to Annex B of Exhibit 10.38).	S-1	3/8/2021	10.39	
	Letter Agreement, dated as of January 15, 2021, among Frontier Airlines, Inc., Frontier Group Holdings, Inc. and the United States Department of the Treasury and acknowledged by The Bank of New York Mellon, as administrative and collateral agent.	S-1	3/8/2021	10.40	
	<u>Payroll Support Program Extension Agreement, dated as of January 15, 2021, between Frontier</u> <u>Airlines, Inc. and the United States Department of the Treasury.</u>	S-1	3/8/2021	10.41	
	<u>Promissory Note, dated as of January 15, 2021, issued by Frontier Group Holdings, Inc. in the</u> name of the United States Department of the Treasury and guaranteed by Frontier Airlines, Inc. and Frontier Airlines Holdings, Inc.	S-1	3/8/2021	10.42	
	Warrant Agreement, dated as of January 15, 2021, between Frontier Group Holdings, Inc. and the United States Department of the Treasury.	S-1	3/8/2021	10.43	
	Form of Warrant (incorporated by reference to Annex B of Exhibit 10.43).	S-1	3/8/2021	10.44	
	Amended and Restated IAE Engine Benefits Agreement A321neo Aircraft (2022 and 2023 Deliveries), dated as of December 22, 2020, among Vertical Horizons, Ltd., International Aero Engines, LLC, Bank of Utah and Frontier Airlines, Inc.	S-1	3/8/2021	10.45	
ł	<u>PW1100G-JM Engine Purchase and Support Agreement, dated as of April 13, 2020, by and between International Aero Engines, LLC and Frontier Airlines, Inc.</u>	S-1	3/8/2021	10.46	
	Code of Ethics.				Х

		Incorporated By Reference			
Exhibit Number	Exhibit Description	Form	Date	Number	Filed Herewith
21.1	List of subsidiaries.	S-1	3/8/2021	21.1	
23.1	Consent of independent registered public accounting firm.	S-1/A	3/18/2021	23.1	
23.2*	Consent of Latham & Watkins LLP (included in Exhibit 5.1).				
24.1	Power of Attorney.	S-1	3/8/2021	24.1	

* To be filed by amendment.

Indicates management contract or compensatory plan.

+ Certain portions of this document that constitute confidential information have been redacted in accordance with Regulation S-K, Item 601(b)(10).

Item 17. Undertakings

The undersigned registrant hereby undertakes that:

- (1) for purposes of determining any liability under the Securities Act, the information omitted from the form of prospectus filed as part of this registration statement in reliance upon Rule 430A and contained in a form of prospectus filed by the registrant pursuant to Rule 424(b)(1) or (4) or 497(h) under the Securities Act shall be deemed to be part of this registration statement as of the time it was declared effective;
- (2) for the purpose of determining any liability under the Securities Act, each post-effective amendment that contains a form of prospectus shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof; and
- (3) the undersigned will provide to the underwriters at the closing specified in the underwriting agreement certificates in such denominations and registered in such names as required by the underwriters to permit prompt delivery to each purchaser.

Insofar as indemnification for liabilities arising under the Securities Act may be permitted to directors, officers and controlling persons of the registrant pursuant to the provisions described in Item 14, or otherwise, the registrant has been advised that in the opinion of the Securities and Exchange Commission such indemnification is against public policy as expressed in the Securities Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the registrant of expenses incurred or paid by a director, officer or controlling person of the registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Securities Act and will be governed by the final adjudication of such issue.

SIGNATURES

Pursuant to the requirements of the Securities Act of 1933, as amended, we have duly caused this Amendment No. 2 to the Registration Statement on Form S-1 to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Denver, State of Colorado, on the 19th day of March, 2021.

FRONTIER GROUP HOLDINGS, INC.

By: /s/ Barry L. Biffle

Barry L. Biffle President and Chief Executive Officer

Pursuant to the requirements of the Securities Act of 1933, as amended, this Amendment No. 1 to the Registration Statement has been signed by the following persons in the capacities indicated below on the dates indicated.

Signature	Title	Date
/s/ Barry L. Biffle Barry L. Biffle	President and Chief Executive Officer (principal executive officer)	March 19, 2021
/s/ James G. Dempsey James G. Dempsey	Executive Vice President and Chief Financial Officer (principal financial officer)	March 19, 2021
* Mark C. Mitchell	Chief Accounting Officer (principal accounting officer)	March 19, 2021
* William A. Franke	Director (Chairman of the Board)	March 19, 2021
* Andrew S. Broderick	Director	March 19, 2021
* Josh T. Connor	Director	March 19, 2021
* Brian H. Franke	Director	March 19, 2021
* Robert J. Genise	Director	March 19, 2021
* Bernard L. Han	Director	March 19, 2021
* Michael R. MacDonald	Director	March 19, 2021

	Signature	Title	Date
	* Patricia Salas Pineda	Director	March 19, 2021
	* Alejandro D. Wolff	Director	March 19, 2021
*By:	/s/ Barry L. Biffle Barry L. Biffle Attorney-in-Fact		

AMENDED AND RESTATED

CERTIFICATE OF INCORPORATION

OF

FRONTIER GROUP HOLDINGS, INC.

Frontier Group Holdings, Inc. (the "Corporation"), a corporation organized and existing under the laws of the General Corporation Law of the State of Delaware (as the same exists or may hereafter be amended, the "DGCL"), does hereby certify as follows:

1. The name of the Corporation is Frontier Group Holdings, Inc. The original Certificate of Incorporation of the Corporation was filed with the Secretary of State of the State of Delaware on September 11, 2013 under the original name, Falcon Acquisition Group, Inc.

3. This Amended and Restated Certificate of Incorporation of the Corporation has been duly adopted by the Board of Directors and stockholders of the Corporation in accordance with Sections 242 and 245 of the Delaware General Corporation Law and by the written consent of its stockholders in accordance with Section 228 of the Delaware General Corporation Law.

4. The Amended and Restated Certificate of Incorporation of the Corporation, as amended, is hereby amended and restated in its entirety to read as follows:

ARTICLE I NAME

The name of the corporation is Frontier Group Holdings, Inc. (the "Corporation").

ARTICLE II REGISTERED OFFICE AND AGENT

The address of the Corporation's registered office in the State of Delaware is 1209 Orange Street in the City of Wilmington, County of New Castle, 19801. The name of its registered agent at such address is The Corporation Trust Company.

ARTICLE III PURPOSE AND DURATION

The purpose of the Corporation is to engage in any lawful activity for which corporations may be organized under the General Corporation Law of the State of Delaware, as amended (the "<u>DGCL</u>"). The Corporation is to have a perpetual existence.

ARTICLE IV CAPITAL STOCK

Section 1. <u>Authorized Shares; Forward Stock Split</u>. Effective upon the filing of this Amended and Restated Certificate of Incorporation (the "<u>Certificate of Incorporation</u>") with the Secretary of State of the State of Delaware (the "<u>Effective Time</u>"), each share of Common Stock (as defined below) issued and outstanding shall be reclassified as [_____] shares of Common Stock (the "<u>Forward Stock Split</u>").

Each stock certificate representing shares of any class or series of Common Stock immediately prior to the Effective Time shall, from and after the Effective Time, represent that number of shares of the class or series of Common Stock into which such shares shall have been reclassified pursuant to the Forward Stock Split; provided, however, that each holder of any stock certificate(s) that represented shares of Common Stock immediately prior to the Effective Time shall be entitled to receive, upon surrender of such certificate(s), one or more certificates (or book entry shares) evidencing and representing the number of shares of Common Stock into which the shares represented by such certificate(s) shall have been reclassified pursuant to the Forward Stock Split.

No fractional shares shall be issued for shares of Common Stock pursuant to the Forward Stock Split. If the Forward Stock Split would result in the issuance of any fractional share of any class or series of Common Stock, the Corporation shall, in lieu of issuing any such fractional share, pay cash in an amount equal to the fair value of such fractional share (as determined in good faith by the Board of Directors). All share, per share and dollar references in this Restated Certificate have been adjusted for the Forward Stock Split only as explicitly provided herein.

The total number of shares of stock which the Corporation is authorized to issue is [_____] shares, of which [_____] shares shall be shares of Common Stock, par value \$0.001 per share (the "<u>Voting Common Stock</u>"), [_____] shares shall be shares of Non-Voting Common Stock, par value \$0.001 per share (the "<u>Non-Voting Common Stock</u>", and together with the Voting Common Stock, the "<u>Common Stock</u>"), and 1,000,000 shares shall be shares of Preferred Stock, par value \$0.001 per share (the "<u>Preferred Stock</u>").

Section 2. <u>Preferred Stock.</u> Shares of Preferred Stock may be issued from time to time in one or more series, each of such series to have such terms as stated in the resolution or resolutions providing for the establishment of such series adopted by the Board of Directors of the Corporation (the "<u>Board of Directors</u>") as hereinafter provided. Authority is hereby expressly granted to the Board of Directors to issue, from time to time, shares of Preferred Stock in one or more series, and, in connection with the establishment of any such series by resolution or resolutions, to determine and fix such voting powers, full or limited, or no voting powers, and such other powers, designations, preferences and relative, participating, optional, and other special rights, and the qualifications, limitations, and restrictions thereof, if any, including, without limitation, dividend rights, conversion rights, redemption privileges and liquidation preferences, as shall be stated in such resolution or resolutions, all to the fullest extent permitted by the DGCL. Without limiting the generality of the foregoing, the resolution or resolutions providing for the establishment of any series of Preferred Stock may, to the extent permitted by law, provide that such series shall be superior to, rank equally with or be junior to the Preferred

Stock of any other series. The powers, preferences and relative, participating, optional and other special rights of each series of Preferred Stock, and the qualifications, limitations or restrictions thereof, if any, may be different from those of any and all other series at any time outstanding. Except as otherwise expressly provided in the resolution or resolutions providing for the establishment of any series of Preferred Stock, no vote of the holders of shares of Preferred Stock or Common Stock shall be a prerequisite to the issuance of any shares of any series of the Preferred Stock authorized by and complying with the conditions of this Certificate of Incorporation.

Section 3. <u>Common Stock</u>. The holders of shares of Common Stock shall have such rights as are set forth in the DGCL and, to the extent consistent therewith, such rights as are set forth below:

(a) Mandatory Conversion. Each share of Non-Voting Common Stock shall be mandatorily convertible, at the sole option of the Corporation at any time and from time to time, into one fully paid and non-assessable share of Voting Common Stock. The Corporation shall exercise such right by delivering written notice (the "Mandatory Conversion Notice") of such conversion to each holder of Non-Voting Common Stock to be so converted stating (i) the Corporation's intention to convert the shares of Non-Voting Common Stock held by such holder and (ii) the date or event upon which such conversion shall be effective (the "Mandatory Conversion Date"), which such Mandatory Conversion Date shall be no less than one calendar day after the date upon which the Corporation delivers the Mandatory Conversion Notice and no more than 180 days thereafter; provided that such conversion may be designated as the occurrence of a given event (for example, completion of a public offering or sale transaction) provided that such event does not occur less than one day or more than 180 calendar days after the related Mandatory Conversion Notice. Following the Corporation's delivery of the Mandatory Conversion Notice, each share of Non-Voting Common Stock held by such holder shall be converted automatically and without further action on the part of such holder into an equal number of shares of Voting Common Stock on the Mandatory Conversion Date. Immediately upon conversion of shares of Non-Voting Common Stock hereunder, the rights of the holders of shares of Non-Voting Common Stock as such shall cease, and such holders shall be treated for all purposes as having become the record holder or holders of such shares of Voting Common Stock. The issuance of certificates, if any, for shares of Voting Common Stock upon conversion of shares of Non-Voting Common Stock hereunder shall be made without charge to the holders of such shares for any stamp or other similar tax in respect of such issuance; provided, however, that if any such certificate is to be issued in a name other than that of the holder of the share or shares of Non-Voting Common Stock converted, then the individual, entity or other person requesting the issuance thereof shall pay to the Corporation the amount of any tax that may be payable in respect of any transfer involved in such issuance or shall establish to the satisfaction of the Corporation that such tax has been paid or is not payable. Except as otherwise provided hereinabove in this Section 3(a), the Non-Voting Common Stock shall not be convertible into any other securities of the Company or under any other circumstances.

(b) <u>Voting</u>. Except as otherwise provided by applicable law, the holders of Voting Common Stock shall be entitled to one vote per share on all matters to be voted on by the stockholders of the Corporation, and the holders of Non-Voting Common Stock shall have no right to vote on any matters to be voted on by the stockholders of the Corporation.

(c) <u>Dividends</u>. Except as may be provided in a resolution or resolutions of the Board of Directors providing for any series of Preferred Stock outstanding at any time, the holders of Voting Common Stock and the holders of Non-Voting Common Stock shall be entitled to share equally, on a per share basis, in such dividends and other distributions of cash, property or shares of stock of the Corporation as may be declared by the Board of Directors from time to time with respect to the Common Stock out of assets or funds of the Corporation legally available therefor; <u>provided</u>, <u>however</u>, that (i) if dividends are declared or paid in shares of Common Stock, the dividends payable to holders of Voting Common Stock shall be payable in Voting Common Stock and the dividends payable to the holders of Non-Voting Common Stock shall be payable in Non-Voting Common Stock and (ii) if the dividends consist of other voting securities of the Corporation, the Corporation shall make available to each holder of Non-Voting Common Stock dividends consisting of non-voting securities (except as otherwise required by law) of the Corporation which are otherwise identical to the voting securities.

(d) <u>Liquidation</u>, <u>Dissolution</u>, <u>etc</u>. Except as may be provided in a resolution or resolutions of the Board of Directors providing for any series of Preferred Stock outstanding at any time, in the event of a voluntary or involuntary liquidation, dissolution, distribution of assets or winding up of the Corporation, the holders of Voting Common Stock and the holders of Non-Voting Common Stock shall be entitled to share equally, on a per share basis, in all assets of the Corporation of whatever kind available for distribution to the holders of Common Stock.

(e) <u>Subdivision or Combination</u>. If the Corporation in any manner subdivides or combines the outstanding shares of one class of Common Stock, the outstanding shares of the other class of Common Stock will be subdivided or combined in the same manner.

(f) Equal Status. Except as expressly provided in this Article IV, shares of Voting Common Stock and Non-Voting Common Stock shall have the same rights and privileges and rank equally, share ratably and be identical in all respect as to all matters. In any merger, consolidation, reorganization or other business combination, the consideration received per share by the holders of the Voting Common Stock and the holders of the Non-Voting Common Stock in such merger, consolidation, reorganization or other business combination shall be identical; provided, however, that if such consideration consists, in whole or in part, of shares of capital stock of, or other equity interests in, the Corporation or any other corporation, partnership, limited liability company or other entity, then the powers, designations, preferences and relative, common, participating, optional or other special rights and qualifications, limitations and restrictions of such shares of capital stock or other equity interests may differ to the extent that the powers, designations, preferences and relative, common, participating optional or other special rights and qualifications, limitations and restrictions of the Voting Common Stock and Non-Voting Common Stock differ as provided herein (including, without limitation, with respect to the voting rights and conversion provisions hereof); and provided further, that, if the holders of the Voting Common Stock or the holders of the Non-Voting Common Stock are granted the right to elect to receive one of two or more alternative forms of

consideration, the foregoing provision shall be deemed satisfied if holders of the other class are granted identical election rights. Any consideration to be paid to or received by holders of Voting Common Stock or holders of Non-Voting Common Stock pursuant to any employment, consulting, severance, non-competition or other similar arrangement approved by the Board of Directors, or any duly authorized committee thereof, shall not be considered to be "consideration received per share" for purposes of the foregoing provision, regardless of whether such consideration is paid in connection with, or conditioned upon the completion of, such merger, consolidation, reorganization or other business combination.

(g) <u>No Preemptive or Subscription Rights</u>. No holder of shares of Common Stock shall be entitled to preemptive or subscription rights.

Section 4. <u>Power to Sell and Purchase Shares</u>. Subject to the requirements of applicable law, the Corporation shall have the power to issue and sell all or any part of any shares of any class of stock herein or hereafter authorized to such persons, and for such consideration, as the Board of Directors shall from time to time, in its discretion, determine, whether or not greater consideration could be received upon the issue or sale of the same number of shares of another class, and as otherwise permitted by law. Subject to the requirements of applicable law, the Corporation shall have the power to purchase any shares of any class of stock herein or hereafter authorized from such persons, and for such consideration, as the Board of Directors shall from time to time, in its discretion, determine, whether or not less consideration could be paid upon the purchase of the same number of shares of another class, and as otherwise permitted by law.

ARTICLE V RESTRICTIONS ON OWNERSHIP

Section 1. Limitations of Ownership by Non-Citizens. At no time shall either (a) the percentage of voting interest of the Corporation or (b) the percentage of the Corporation's capital stock (including Voting Common Stock and Non-Voting Common Stock), owned or controlled by persons who are not "citizens of the United States" (as such term is defined in Title 49, United States Code, Section 40102 and administrative interpretations thereof issued by the Department of Transportation or its successor, or as the same may be from time to time amended, supplemented or succeeded, "<u>Applicable Laws</u>") ("<u>Non-Citizens</u>") exceed the limitations provided under Applicable Laws (which, as of the date hereof and for informational purposes only, is 25.0% and 49.0%, respectively). In the event that Non-Citizens shall own (beneficially or of record) or have voting control over any shares of capital stock of the Corporation, the voting rights of such persons shall be subject to automatic suspension to the extent required to ensure that the Corporation is in compliance with applicable provisions of law and regulations relating to ownership or control of a U.S. air carrier. The Bylaws of the Corporation shall contain provisions restricting or removing voting rights as to shares of voting stock owned or controlled by Non-Citizens. Any determination as to ownership, control or citizenship made by the Board of Directors shall be conclusive and binding as between the Corporation and any stockholder for purposes of this Article V.

Section 2. Legend. Each certificate or other representative document for capital stock of the Corporation with voting rights (including each such certificate or representative document for such capital stock issued upon any permitted transfer of capital stock) shall contain a legend in substantially the following form:

THE SECURITIES OF FRONTIER GROUP HOLDINGS, INC. REPRESENTED BY THIS CERTIFICATE OR DOCUMENT ARE SUBJECT TO VOTING RESTRICTIONS WITH RESPECT TO CERTAIN SECURITIES HELD BY PERSONS OR ENTITIES THAT FAIL TO QUALIFY AS "CITIZENS OF THE UNITED STATES" AS THE TERM IS DEFINED IN SECTION 40102(a)(15) OF SUBTITLE VII OF TITLE 49 OF THE UNITED STATES CODE, AS AMENDED, IN ANY SIMILAR LEGISLATION OF THE UNITED STATES CODE, AS AMENDED, IN ANY SIMILAR LEGISLATION OF THE UNITED STATES ENACTED IN SUBSTITUTION OR REPLACEMENT THEREFOR, AND AS INTERPRETED BY THE DEPARTMENT OF TRANSPORTATION, ITS PREDECESSORS AND SUCCESSORS, FROM TIME TO TIME. SUCH VOTING RESTRICTIONS ARE CONTAINED IN THE AMENDED AND RESTATED CERTIFICATE OF INCORPORATION AND THE BYLAWS OF FRONTIER GROUP HOLDINGS, INC., AS THE SAME MAY BE AMENDED OR RESTATED FROM TIME TO TIME. A COMPLETE AND CORRECT COPY OF SUCH AMENDED AND RESTATED CERTIFICATE OF INCORPORATION AND THE BYLAWS SHALL BE FURNISHED FREE OF CHARGE TO THE HOLDER OF THE SECURITIES REPRESENTED HEREBY UPON WRITTEN REQUEST TO THE SECRETARY OF FRONTIER GROUP HOLDINGS, INC.

ARTICLE VI BOARD OF DIRECTORS

Section 1. <u>Powers of the Board</u>. The business and affairs of the Corporation shall be managed by or under the direction of the Board of Directors. In addition to the powers and authority expressly conferred upon them by applicable law or by the Certificate of Incorporation or the Bylaws of the Corporation, the directors are hereby empowered to exercise all such powers and do all such acts and things as may be exercised or done by the Corporation.

Section 2. <u>Non-Citizens</u>. Notwithstanding anything to the contrary in this Certificate of Incorporation or the Corporation's Bylaws, the number of Non-Citizens who can serve as members of the Board of Directors shall at no time exceed the limitations provided under Applicable Laws (which, as of the date hereof and for informational purposes only, is one-third (1/3) of the total number of members then serving on the Board of Directors).

Section 3. Bylaws. The Board of Directors is expressly authorized to make, alter or repeal Bylaws of the Corporation.

Section 4. Elections of Directors. Elections of directors need not be by ballot unless the Bylaws of the Corporation shall so provide.

ARTICLE VII STOCKHOLDERS

Section 1. <u>Meeting Location</u>. Meetings of stockholders may be held within or without the State of Delaware, as the Bylaws of the Corporation may provide. The books of the Corporation may be kept outside the State of Delaware at such place or places as may be designated from time to time by the Board of Directors or in the Bylaws of the Corporation.

ARTICLE VIII LIABILITY AND INDEMNIFICATION

Section 1. <u>Director Liability</u>. To the maximum extent permitted by the DGCL, a director of the Corporation shall not be personally liable to the Corporation or its stockholders for monetary damages for breach of fiduciary duty as a director. If the DGCL is amended after approval by the stockholders of this Article VIII to authorize corporate action further eliminating or limiting the personal liability of directors, then the liability of a director of the Corporation shall be eliminated or limited to the fullest extent permitted by the DGCL as so amended.

Section 2. <u>Indemnification</u>. The Corporation shall indemnify its directors and officers to the fullest extent authorized or permitted by the DGCL, and such right to indemnification shall continue as to a person who has ceased to be director or officer of the Corporation and shall inure to the benefit of his or her heirs, executors and administrators; <u>provided</u>, <u>however</u>, that, except for proceedings to enforce rights to indemnification, the Corporation shall not be obligated to indemnify any director or officer (or his or her heirs, executors or administrators) in connection with a proceeding (or part thereof) initiated by such person unless such proceeding (or part thereof) was authorized by the Board of Directors. The right to indemnification conferred in this paragraph shall be a contract right and shall include the right to be paid by the Corporation the expenses incurred in defending or otherwise participating in any proceeding in advance of its final disposition.

Section 3. <u>Right to Indemnify</u>. To the fullest extent authorized or permitted by the DGCL, the Corporation shall have the express authority to enter into such agreements as the Board of Directors deems appropriate for the indemnification of directors and officers of the Corporation. Such agreements may contain provisions relating to, among other things, the advancement of expenses, a person's right to bring suit against the Corporation to enforce his or her right to indemnification, the establishment of a trust to assure the availability of funds to satisfy the Corporation's indemnification obligations to such person and other matters as the Board of Directors deems appropriate or advisable.

Section 4. <u>Non-Exclusive</u>. The rights to indemnification and to the advancement of expenses conferred in this Article VIII shall not be exclusive of any other right which any person may have or hereafter acquire under this Certificate of Incorporation, the Bylaws of the Corporation, any statute, agreement, vote of stockholders or disinterested directors or otherwise.

Section 5. <u>Insurance</u>. The Corporation may maintain insurance, at its expense, to protect itself and any director, officer, employee or agent of the Corporation or another corporation, partnership, joint venture, trust, employee benefit plan or other enterprise against any expense, liability or loss, whether or not the Corporation would have the power to indemnify such person against such expense, liability or loss under the DGCL.

Section 6. <u>Amendment or Repeal</u>. Neither any amendment nor repeal of this Article VIII, nor the adoption of any provision of this Certificate of Incorporation inconsistent with this Article VIII, shall eliminate or reduce the effect of this Article VIII in respect of any matter occurring, or any action or proceeding accruing or arising or that, but for this Article VIII, would accrue or arise, prior to such amendment, repeal or adoption of an inconsistent provision.

ARTICLE IX EXCLUSIVE FORUM

Unless the Corporation consents in writing to the selection of an alternative forum, the Court of Chancery of the State of Delaware (or, if such court does not have subject matter jurisdiction thereof, the federal district court of the State of Delaware) shall, to the fullest extent permitted by law, be the sole and exclusive forum for: (i) any derivative action or proceeding brought on behalf of the Corporation, (ii) any action asserting a claim for or based on a breach of a fiduciary duty owed by any current or former director, officer, other employee, agent or stockholder of the Corporation to the Corporation or any current or former director, officer, employee, agent or stockholder of the Corporation arising pursuant to any provision of the General Corporation Law or the Corporation's Certificate of Incorporation or Bylaws or as to which the General Corporation that is governed by the internal affairs doctrine.

Unless the Corporation consents in writing to the selection of an alternative forum, the federal district courts of the United States of America shall, to the fullest extent permitted by law, be the sole and exclusive forum for the resolution of any complaint asserting a cause or causes of action arising under the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder, including all causes of action asserted against any defendant to such complaint. For the avoidance of doubt, this provision is intended to benefit and may be enforced by the Corporation, its officers and directors, the underwriters to any offering giving rise to such complaint, and any other professional or entity whose profession gives authority to a statement made by that person or entity and who has prepared or certified any part of the documents underlying the offering.

Any person or entity purchasing or otherwise acquiring or holding any interest in shares of capital stock of the Corporation shall be deemed to have notice of and consented to the provisions of this Article.

Failure to enforce the foregoing provisions would cause the Corporation irreparable harm, and the Corporation shall be entitled to equitable relief, including injunctive relief and specific performance, to enforce the foregoing provisions.

This exclusive forum provision will not apply to suits brought to enforce any liability or duty created by the Exchange Act of 1934, as amended, or any other claim for which the federal courts have exclusive jurisdiction.

IN WITNESS WHEREOF, the undersigned have executed this Amended and Restated Certificate of Incorporation on this [___] day of March, 2021.

FRONTIER GROUP HOLDINGS, INC.

By:

Name: Barry L. Biffle Title: President and Chief Executive Officer

AMENDED AND RESTATED

CERTIFICATE OF INCORPORATION

OF

FRONTIER GROUP HOLDINGS, INC.

Frontier Group Holdings, Inc. (the "<u>Corporation</u>"), a corporation organized and existing under the laws of the General Corporation Law of the State of Delaware (as the same exists or may hereafter be amended, the "<u>DGCL</u>"), does hereby certify as follows:

1. The name of the Corporation is Frontier Group Holdings, Inc. The original Certificate of Incorporation of the Corporation was filed with the Secretary of State of the State of Delaware on September 11, 2013 under the original name Falcon Acquisition Group, Inc.

2. This Amended and Restated Certificate of Incorporation of the Corporation has been duly adopted by the Corporation's Board of Directors and stockholders in accordance with Sections 242 and 245 of the DGCL and by the written consent of its stockholders in accordance with Section 228 of the DGCL.

3. The Amended and Restated Certificate of Incorporation of the Corporation is hereby amended and restated in its entirety to read as follows:

ARTICLE I <u>NAME</u>

The name of the corporation is Frontier Group Holdings, Inc. (the "Corporation").

ARTICLE II REGISTERED OFFICE AND AGENT

The address of the Corporation's registered office in the State of Delaware is 1209 Orange Street in the City of Wilmington, County of New Castle, Delaware 19801. The name of its registered agent at such address is The Corporation Trust Company.

ARTICLE III PURPOSE AND DURATION

The purpose of the Corporation is to engage in any lawful activity for which corporations may be organized under the General Corporation Law of the State of Delaware (as the same exists or may hereafter be amended, the "<u>DGCL</u>"). The Corporation is to have a perpetual existence.

ARTICLE IV CAPITAL STOCK

Section 1. <u>Authorized Shares. The total number of shares of stock which the Corporation is authorized to issue is 910,000,000 shares, of which</u> <u>750,000,000 shares shall be shares of voting common stock, par value \$0.001 per share (the "Voting Common Stock"), 150,000,000 shares shall be</u> <u>shares of non-voting common stock, par value \$0.001 per share (the "Non-Voting Common Stock," and together with the Voting Common Stock, the</u> <u>"Common Stock"), and 10,000,000 shares shall be shares of preferred stock, par value \$0.001 per share (the "Preferred Stock").</u>

Section 2. <u>Preferred Stock</u>. Shares of Preferred Stock may be issued from time to time in one or more series, each of such series to have such terms as stated in the resolution or resolutions providing for the creation and issuance of such series adopted by the Board of Directors of the Corporation as hereinafter provided. Authority is hereby expressly granted to the Board of Directors of the Corporation to issue, from time to time, shares of Preferred Stock in one or more series, and, in connection with the creation of any such series, by adopting a resolution or resolutions providing for the issuance of the shares thereof and by filing a certificate of designation relating thereto in accordance with the DGCL (a "<u>Certificate of Designation</u>"), to determine and fix the number of shares of such series and such voting powers, full or limited, or no voting powers, and such designations, preferences and relative participating, optional or other special rights, and qualifications, limitations or restrictions thereof, including without limitation thereof, dividend rights, conversion rights, redemption privileges and liquidation preferences as shall be stated and expressed in such resolutions, and to increase (but not above the total number of authorized shares of such series) or decrease (but not below the number of shares of such series, all to the fullest extent now or hereafter permitted by the DGCL. Without limiting the generality of the foregoing, the resolution or resolutions providing for the creation and issuance of any series of Preferred Stock may provide that such series shall be superior or rank equally or be junior to any other series of Preferred Stock to the extent permitted by law and this Certificate of Incorporation (including any Certificate of Designation) (this "<u>Certificate of Incorporation</u>"). Except as otherwise required by law, holders of any series of Preferred Stock shall be entitled only to such voting rights, if any, as shall expressly be granted thereto by this Certificate of Incorpora

Subject to the rights and preferences, if any, of any holders of any outstanding series of Preferred Stock, the number of authorized shares of Preferred Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the stock of the Corporation entitled to vote thereon, irrespective of the provisions of Section 242(b)(2) of the DGCL.

Section 3. <u>Common Stock</u>. Except as stated in this Certificate of Incorporation, the holders of shares of Common Stock shall have such rights as are set forth in the DGCL:

(a) Conversion. Each share of Non-Voting Common Stock shall be convertible, at the option of the holder thereof at any time and from time to time, into one fully paid and non-assessable share of Voting Common Stock, subject to any restrictions on ownership and control of voting stock by Non-Citizens as set forth in this Certificate of Incorporation and the Bylaws. Such right shall be exercised by the surrender to the Corporation of the certificate or certificates, if any, representing the shares of Non-Voting Common Stock to be converted at any time during normal business hours at the office of the Corporation's transfer agent (the "Transfer Agent"), accompanied by a written notice from the holder of such shares stating that such holder desires to convert such shares, or a stated number of the shares represented by such certificate or certificates, if any, into an equal number of shares of Voting Common Stock, and (if so required by the Transfer Agent) by instruments of transfer, in form satisfactory to the Transfer Agent, duly executed by such holder or such holder's duly authorized attorney-in-fact, and transfer tax stamps or funds therefor if required pursuant to this Section 3(a) of Article IV. To the extent permitted by law, such conversion shall be deemed to have been effected at the close of business on the date of such surrender. Subject to the last sentence of Section 3(c) of this Article IV, immediately upon conversion of shares of Non-Voting Common Stock, the rights of the holders of shares of Non-Voting Common Stock as such shall cease, and such holders shall be treated for all purposes as having become the record holder or holders of such shares of Voting Common Stock. The issuance of certificates, if any, for shares of Voting Common Stock upon conversion of shares of Non-Voting Common Stock shall be made without charge to the holders of such shares for any stamp or other similar tax in respect of such issuance; provided, however, that if any such certificate is to be issued in a name other than that of the holder of the share or shares of Non-Voting Common Stock converted, then the individual, entity or other person requesting the issuance thereof shall pay to the Corporation the amount of any tax that may be payable in respect of any transfer involved in such issuance or shall establish to the satisfaction of the Corporation that such tax has been paid or is not payable.

(b) <u>Voting</u>. Except as otherwise provided herein or expressly required by law, the holders of Voting Common Stock, as such, shall be entitled to one vote per share on all matters to be voted on by the stockholders of the Corporation, and the holders of Non-Voting Common Stock, as such, shall have no right to vote on any matters to be voted on by the stockholders of the Corporation. Except as otherwise required by law, holders of Common Stock, as such, shall not be entitled to vote on any amendment to this Certificate of Incorporation that relates solely to the rights, powers, preferences (or the qualifications, limitations or restrictions thereof) or other terms of one or more outstanding series of Preferred Stock if the holders of such affected series are entitled, either separately or together with the holders of one or more other such series, to vote thereon pursuant to this Certificate of Incorporation or pursuant to the DGCL.

Subject to the rights and preferences, if any, of any holders of any outstanding series of Preferred Stock, the number of authorized shares of Common Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the stock of the Corporation entitled to vote thereon, irrespective of the provisions of Section 242(b)(2) of the DGCL.

(c) Dividends. Subject to applicable law and the rights and preferences of any holders of any outstanding series of Preferred Stock, the holders of Voting Common Stock and the holders of Non-Voting Common Stock shall be entitled to share equally, on a per-share basis, in such dividends of cash, property or shares of stock of the Corporation as may be declared by the Board of Directors from time to time with respect to the Common Stock out of assets or funds of the Corporation legally available therefor; provided, however, that (i) if dividends are declared and paid in shares of Common Stock, the dividends payable to holders of Voting Common Stock shall be payable in Voting Common Stock, and the dividends payable to the holders of Non-Voting Common Stock shall be payable in Non-Voting Common Stock, and (ii) if the dividends are declared and paid in other voting securities of the Corporation, the Corporation shall pay to each holder of Voting Common Stock dividends consisting of voting securities and shall pay to each holder of Non-Voting Common Stock dividends consisting of non-voting securities (except as otherwise required by law) of the Corporation which are otherwise identical to the voting securities of the Corporation so declared and paid on the Voting Common Stock. Notwithstanding the foregoing, if the date on which any share of Non-Voting Common Stock is converted into Voting Common Stock pursuant to Section 3(a) of this Article IV is after the record date for the determination of the holders of Non-Voting Common Stock entitled to receive any dividend and prior to the date on which such dividend is to be paid to such holders, the holder of the Voting Common Stock issued upon the conversion of such converted share of Non-Voting Common Stock will be entitled to receive such dividend on such payment date; provided, however, that to the extent that such dividend is payable in shares of Non-Voting Common Stock, no such shares of Non-Voting Common Stock shall be issued in payment thereof, and such dividend shall instead be paid by the issuance of such number of shares of Voting Common Stock into which such shares of Non-Voting Common Stock, if issued, would have been convertible on such payment date.

(d) <u>Liquidation</u>, <u>Dissolution</u>, <u>etc</u>. Subject to the rights and preferences, if any, of any holders of any outstanding series of Preferred Stock, in the event of a voluntary or involuntary liquidation, dissolution, or winding up of the Corporation, the holders of Voting Common Stock and the holders of Non-Voting Common Stock shall be entitled to share equally, on a per-share basis, in all assets of the Corporation of whatever kind available for distribution to the holders of Common Stock.

(e) <u>Subdivision or Combination</u>. If the Corporation in any manner subdivides or combines the outstanding shares of one class of Common Stock, the outstanding shares of the other class of Common Stock will be subdivided or combined in the same manner.

(f) Equal Status. Except as expressly provided in this Article IV, shares of Voting Common Stock and Non-Voting Common Stock shall have the same rights and privileges and rank equally, share ratably and be identical in all respect as to all matters. In any merger, consolidation, reorganization or other business combination, the consideration received per share by the holders of the Voting Common Stock and the holders of the Non-Voting Common Stock in such merger, consolidation, reorganization or other business combination shall be identical; provided, however, that if such consideration consists, in whole or in part, of shares of capital stock of, or other equity interests in, the Corporation or any other corporation, partnership, limited liability company or other entity, then the powers, designations, preferences and relative, common, participating, optional or other special rights and qualifications, limitations and restrictions of such shares of capital stock or other equity interests may differ to the extent that the powers, designations, preferences and relative,

common, participating, optional or other special rights and qualifications, limitations and restrictions of the Voting Common Stock and Non-Voting Common Stock differ as provided herein (including, without limitation, with respect to the voting rights and conversion provisions hereof); and <u>provided further</u>, that, if the holders of the Voting Common Stock or the holders of the Non-Voting Common Stock are granted the right to elect to receive one of two or more alternative forms of consideration, the foregoing provision shall be deemed satisfied if holders of the other class are granted identical election rights. Any consideration to be paid to or received by holders of Voting Common Stock or holders of Non-Voting Common Stock pursuant to any employment, consulting, severance, non-competition or other similar arrangement approved by the Board of Directors, or any duly authorized committee thereof, shall not be considered to be "consideration received per share" for purposes of the foregoing provision, regardless of whether such consideration is paid in connection with, or conditioned upon the completion of, such merger, consolidation, reorganization or other business combination.

(g) No Preemptive or Subscription Rights. No holder of shares of Common Stock shall be entitled to preemptive or subscription rights.

ARTICLE V RESTRICTIONS ON OWNERSHIP AND CONTROL

Section 1. Limitations of Ownership by Non-Citizens. All shares of (i) Voting Common Stock, (ii) Non-Voting Common Stock, and (iii) Preferred Stock (collectively, "Equity Securities") shall be subject to the following limitations:

(a) <u>Non-Citizen Voting and Ownership Limitations</u>. In no event shall persons or entities who fail to qualify as a "citizen of the United States," as the term is defined in Section 40102(a)(15) of Subtile VII of Title 49 of the United States Code, as amended, in any similar legislation of the United States enacted in substitution or replacement therefor, and as interpreted by the United States Department of Transportation, its predecessors and successors, from time to time (the "<u>Applicable Law</u>"), including any agent, trustee or representative of such persons or entities (collectively, the "<u>Non-Citizens</u>"), be entitled to own (beneficially or of record) and/or control Equity Securities representing in excess of (i) the percentage provided for under Applicable Law of the aggregate votes of all outstanding Equity Securities of the Corporation (the "<u>Voting Limitation</u>") and (ii) the percentage provided for under Applicable Law of all outstanding Equity Securities of the Corporation (the "<u>Outstanding Limitation</u>"), in each case as more specifically set forth in the Bylaws of the Corporation. As of the effectiveness of the filing of this Certificate of Incorporation in connection with the Corporation's initial underwritten public offering under the Securities Act of 1933, as amended (the "<u>Effective Time</u>"), and for informational purposes only, under Applicable Law, the Voting Limitation is 25.0%, and the Outstanding Limitation is 49.0% (the "<u>Cap Amounts</u>").

(b) Enforcement of Cap Amounts. Except as otherwise set forth in the Bylaws, the restrictions imposed by the Cap Amounts shall be applied to each Non-Citizen in reverse chronological order based upon the date of registration (or attempted registration in the case of the Outstanding Limitation) on the separate stock record maintained by the Corporation or any transfer agent for the registration of Equity Securities held by Non-Citizens (the "Foreign Stock Record") or the stock transfer records of the Corporation. At no time shall the Equity Securities held by Non-Citizens be voted, unless such shares are registered on the Foreign Stock Record. In the event that Non-Citizens shall own (beneficially or of record) or have voting control over any Equity Securities, the voting rights of such persons shall be subject to automatic suspension to the extent required to ensure that the Corporation is in compliance with Applicable Law. In the event that any transfer of Equity Securities to a Non-Citizen would result in Non-Citizens owning (beneficially or of record) or controlling more than the Outstanding Limitation, such transfer shall be void and of no effect and shall not be recorded in the books and records of the Corporation. The Bylaws shall contain provisions to implement this Article V, including, without limitation, provisions restricting or prohibiting the transfer of Equity Securities to Non-Citizens if such transfer will result in Non-Citizens owning and/or controlling Equity Securities in excess of the Voting Limitation or the Outstanding Limitation and provisions restricting or removing voting rights as to Equity Securities owned or controlled by Non-Citizens. Any determination as to ownership, control or citizenship made by the Board of Directors shall be conclusive and binding as between the Corporation and any stockholder.

Section 2. <u>Legend</u>. Each certificate or other representative document for capital stock of the Corporation (including each such certificate or representative document for such capital stock issued upon any permitted transfer of capital stock) shall contain a legend related to the applicable limitation on ownership and/or control by Non-Citizens in substantially the form specified in the Bylaws.

ARTICLE VI BOARD OF DIRECTORS

Section 1. <u>Powers of the Board</u>. Except as otherwise provided by this Certificate of Incorporation or the DGCL, the business and affairs of the Corporation shall be managed by or under the direction of the Board of Directors. In addition to the powers and authority expressly conferred upon them by applicable law or by this Certificate of Incorporation or the Bylaws of the Corporation, the directors are hereby empowered to exercise all such powers and do all such acts and things as may be exercised or done by the Corporation, except for such powers, acts and things that are by the DGCL, this Certificate of Incorporation of the Bylaws of the Corporation required to be exercised or done by the stockholders.

Section 2. <u>Classification of the Board</u>. Subject to the rights and preferences, if any, of any holders of any outstanding series of Preferred Stock to elect directors, the directors of the Corporation shall be classified with respect to the time for which they severally hold office into three classes, designated as Class I, Class II and Class III. The initial Class I directors shall serve for a term expiring at the first annual meeting of the stockholders following the time at which the initial classification of the Board becomes effective; the initial Class II directors shall serve for a term expiring at the initial Class III directors shall serve for a term expiring at the second annual meeting of the stockholders following the time at which the initial classification of the Board becomes effective; and the initial Class III directors shall serve for a term

expiring at the third annual meeting following the time at which the initial classification of the Board becomes effective. At each annual meeting of stockholders of the Corporation beginning with the first annual meeting of stockholders following the time at which the initial classification of the Board becomes effective, subject to the rights and preferences, if any, of any holders of any outstanding series of Preferred Stock to elect directors, the successors of the class of directors whose term expires at that meeting shall be elected to hold office for a term expiring at the annual meeting of stockholders held in the third year following the year of their election. Each director shall hold office until his or her successor is duly elected and qualified or until his or her earlier death, resignation, disqualification or removal. No decrease in the number of directors shall shorten the term of any incumbent director. The Board of Directors is authorized to assign members of the Board of Directors already in office to Class I, Class II and Class III.

Section 3. <u>Number of Directors</u>. Subject to the rights and preferences, if any, of any holders of any outstanding series of Preferred Stock to elect directors, the number of directors which shall constitute the whole Board of Directors shall be no less than one and not more than fifteen members and be fixed exclusively by one or more resolutions adopted from time to time by the affirmative vote of a majority of the directors then in office, even though less than a quorum of the Board of Directors. Notwithstanding anything to the contrary in this Certificate of Incorporation or the Bylaws of the Corporation, at least two-thirds of the members of the Board of Directors shall be "citizens of the United States" as provided under Applicable Law (as defined in Article V) and the Chairman of the Board shall be a "citizen of the United States" as provided under Applicable Law for so long as required by Applicable Law. If the number of Non-Citizens (as defined in Article V) serving on the Board of Directors at any time exceeds the limitations provided under Applicable Law, one or more directors who are Non-Citizens shall, in reverse chronological order based on their tenure of service on the Board, cease to be qualified as directors and shall automatically cease to be directors.

Section 4. <u>Removal of Directors</u>. Subject to the rights and preferences, if any, of any holders of any outstanding series of Preferred Stock to elect directors, (a) until such time as Indigo Frontier Holdings Company, LLC or any successor thereto by merger, consolidation, acquisition of all or substantially all assets or otherwise by operation of law or any of its Affiliates (as such term is defined in Rule 12b-2 promulgated under the Securities Exchange Act of 1934, as amended) ("<u>Indigo</u>") ceases to beneficially own shares of Common Stock representing at least a majority of the voting power ("<u>Indigo Control</u>") of all the then outstanding shares of capital stock of the Corporation entitled to vote at an election of directors (the "<u>Voting Stock</u>"), the Board of Directors or any individual director may be removed from office at any time with or without cause by the affirmative vote of the holders of a majority of the voting power of the Voting Stock and (b) from and after such time as the Corporation is no longer under Indigo Control, the Board of Directors or any individual director may be removed from office at any time only for cause by the affirmative vote of a majority of the voting power of the Voting Stock and (b) from office at any time only for cause by the affirmative vote of a majority of the voting power of the Voting Stock.

Section 5. <u>Vacancies and Newly Created Directorships</u>. Subject to the rights and preferences, if any, of any holders of any outstanding series of Preferred Stock to elect directors, any vacancies on the Board of Directors resulting from death, resignation, disqualification, removal or other causes and any newly created directorships resulting from any increase in the number of directors shall, unless the Board determines by resolution that any such vacancies or newly created directorships shall be filled by stockholders, be filled only by the affirmative vote of a majority of the directors then in office, even though less than a quorum of the Board of Directors, and not by the stockholders. Any director elected in accordance with the preceding sentence shall hold office for the remainder of the full term of the class for which the vacancy or newly created directorship was created or occurred and until such director's successor shall have been elected and qualified, or until such director's earlier death, resignation, disqualification or removal.

Section 6. <u>Preferred Directors</u>. Whenever the holders of any one or more series of Preferred Stock issued by the Corporation shall have the right, voting separately as a series or separately as a class with one or more such other series, to elect directors pursuant to this Certificate of Incorporation, then upon commencement and for the duration of the period during which such right continues: (i) the then otherwise total number of authorized directors of the Corporation shall automatically be increased by such specified number of directors, and the holders of such Preferred Stock shall be entitled to elect the additional directors so provided for or fixed pursuant to this Certificate of Incorporation, and (ii) each such additional director shall serve until such director's successor shall have been duly elected and qualified, or until such director's right to hold such office terminates pursuant to said provisions, whichever occurs earlier, subject to his earlier death, disqualification, resignation or removal. Except as otherwise provided in the Certificate of Designation(s) in respect of one or more series of Preferred Stock, whenever the holders of any series of Preferred Stock having such right to elect additional directors are divested of such right pursuant to the provisions of such Certificate of Designation(s), the terms of office of all such additional directors elected by the holders of such series of Preferred Stock, or elected to fill any vacancies resulting from the death, resignation, disqualification or removal of such additional directors, shall forthwith terminate, such additional directors shall automatically cease to be qualified and the total authorized number of directors of the Corporation shall automatically be reduced accordingly.

Section 7. <u>Bylaws</u>. In furtherance and not in limitation of the powers conferred by statute, the Board of Directors is expressly authorized to adopt, amend or repeal Bylaws of the Corporation. In addition to any vote of the holders of any class or series of stock of the Corporation required by applicable law or by this Certificate of Incorporation or the Bylaws of the Corporation, the adoption, amendment or repeal of the Bylaws of the Corporation by the stockholders of the Corporation shall require the affirmative vote of the holders of (a) a majority of the voting power of the Voting Stock while the Corporation is under Indigo Control and (b) from and after the time that the Corporation is no longer under Indigo Control, at least sixty-six and two-thirds percent (662/3%) of the voting power of the Voting Stock.

Section 8. Elections of Directors. Elections of directors need not be by written ballot unless the Bylaws of the Corporation shall so provide.

Section 9. <u>Officers</u>. Except as otherwise expressly delegated by resolution of the Board of Directors, the Board of Directors shall have the exclusive power and authority to appoint and remove officers of the Corporation. Notwithstanding anything to the contrary in this Certificate of Incorporation or the Corporation's Bylaws, (a) the Chief Executive Officer, if any, and the President, and at least two-thirds of the other managing officers of the Corporation shall be "citizens of the United States," as defined under Applicable Law (as defined above) for so long as such restrictions are required by Applicable Law (as defined in Article X below).

ARTICLE VII STOCKHOLDERS

Section 1. <u>Actions by Consent</u>. Except as may be provided in a resolution or resolutions of the Board of Directors providing for any series of Preferred Stock, any action required or permitted to be taken by the stockholders of the Corporation may be effected by an action by consent in lieu of a meeting with the approval of the holders of outstanding capital stock having not less than the minimum voting power that would be necessary to authorize or take such action at a meeting at which all shares of capital stock entitled to vote thereon were present and voted; <u>provided</u>, that from and after the time that the Corporation is no longer under Indigo Control, any action required or permitted to be taken by the stockholders of the Corporation must be effected only at a duly called annual or special meeting of such stockholders and may not be effected by any consent in lieu of a meeting by such stockholders.

Section 2. <u>Special Meetings of Stockholders</u>. Subject to the rights and preferences, if any, of any holders of any outstanding series of Preferred Stock, special meetings of stockholders of the Corporation may be called at any time (a) by the Chair of the Board of Directors or by the Secretary of the Corporation upon direction of the Board of Directors pursuant to a resolution adopted by a majority of the entire Board of Directors or by the holders of a majority of the voting power of the Voting Stock while the Corporation is under Indigo Control and (b) from and after the time that the Corporation is no longer under Indigo Control, only by the Chair of the Board of Directors or by the Secretary of the Corporation upon direction of the Board of Directors or by the entire Board of Directors, but such special meetings may not be called by any other person or persons.

Section 3. <u>Meeting Location</u>. Meetings of stockholders may be held within or without the State of Delaware, as the Bylaws of the Corporation may provide. The books of the Corporation may be kept outside the State of Delaware at such place or places as may be designated from time to time by the Board of Directors, or in the Bylaws of the Corporation.

Section 4. <u>Advance Notice</u>. Advance notice of stockholder nominations for the election of directors and of business to be brought by stockholders before any meeting of the stockholders of the Corporation shall be given in the manner provided in the Bylaws of the Corporation.

ARTICLE VIII LIABILITY AND INDEMNIFICATION

Section 1. <u>Director Limitation of Liability</u>. No director of the Corporation shall have any personal liability to the Corporation or its stockholders for monetary damages for any breach of fiduciary duty as a director, except to the extent such exemption from liability or limitation thereof is not permitted under the DGCL as the same exists or hereafter may be amended. Any amendment, repeal or modification of this Article VIII, or the adoption of any provision of this Certificate of Incorporation inconsistent with this Article VIII, shall not adversely affect any right or protection of a director of the Corporation with respect to any act or omission occurring prior to such amendment, repeal, modification or adoption. If the DGCL is amended after approval by the stockholders of this Article VIII to authorize corporate action further eliminating or limiting the personal liability of directors, then the liability of a director of the Corporation shall be eliminated or limited to the fullest extent permitted by the DGCL as so amended.

Section 2. Right to Indemnification.

(a) <u>Directors and Officers.</u> The Corporation shall indemnify and hold harmless, to the fullest extent permitted by the DGCL as it presently exists or may hereafter be amended, any director or officer of the Corporation who was or is made or is threatened to be made a party or is otherwise involved in any action, suit or proceeding, whether civil, criminal, administrative or investigative (a "<u>Proceeding</u>") by reason of the fact that he or she, or a person for whom he or she is the legal representative, is or was a director or officer of the Corporation or, while serving as a director or officer of the Corporation, is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership (a "covered person"), joint venture, trust, enterprise or non-profit entity, including service with respect to employee benefit plans, against all liability and loss suffered and expenses (including attorneys' fees, judgments, fines ERISA excise taxes or penalties and amounts paid in settlement) reasonably incurred by such person in connection with any such Proceeding. Notwithstanding the foregoing, except as otherwise provided in Section 3 of this Article VIII, the Corporation shall be required to indemnify a person in connection with a Proceeding initiated by such person only if the Proceeding was authorized in the specific case by the Board.

(b) <u>Others</u>. The Corporation shall have the power to indemnify and hold harmless, to the fullest extent permitted by applicable law as it presently exists or may hereafter be amended, any employee or agent of the Corporation who was or is made or is threatened to be made a party or is otherwise involved in any Proceeding by reason of the fact that he or she, or a person for whom he or she is the legal representative, is or was an employee or agent of the Corporation or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust, enterprise or non-profit entity, including service with respect to employee benefit plans, against all liability and loss suffered and expenses reasonably incurred by such person in connection with any such Proceeding.

Section 3. <u>Prepayment of Expenses</u>. The Corporation shall to the fullest extent not prohibited by applicable law pay the reasonable expenses (including attorneys' fees) incurred by any covered person, and may pay the expenses incurred by any employee or agent of the Corporation, in defending any Proceeding in advance of its final disposition; *provided, however*, that such payment of expenses in advance of the final disposition of the Proceeding shall be made only upon receipt of an undertaking by the person to repay all amounts advanced if it should be ultimately determined that the person is not entitled to be indemnified under this Article VIII or otherwise.

Section 4. <u>Non-Exclusivity of Rights</u>. The rights conferred on any person by this Article VIII shall not be exclusive of any other rights which such person may have or hereafter acquire under any statute, provision of the Certificate of Incorporation, the Bylaws of the Corporation, agreement, vote of stockholders or disinterested directors or otherwise.

Section 5. <u>Insurance</u>. The Corporation may purchase and maintain insurance on behalf of any person who is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust enterprise or non-profit entity against any liability asserted against him or her and incurred by him or her in any such capacity, or arising out of his or her status as such, whether or not the Corporation would have the power to indemnify him or her against such liability under the provisions of the DGCL.

Section 6. <u>Other Indemnification</u>. The Corporation's obligation, if any, to indemnify or advance expenses to any person who was or is serving at its request as a director, officer, employee or agent of another corporation, partnership, joint venture, trust, enterprise or non-profit entity shall be reduced by any amount such person may collect as indemnification or advancement of expenses from such other corporation, partnership, joint venture, trust, enterprise or non-profit enterprise.

Section 7. <u>Continuation of Indemnification</u>. The Corporation's obligation, if any, to indemnify or advance expenses to any person who was or is serving at its request as a director, officer, employee or agent of another corporation, partnership, joint venture, trust, enterprise or non-profit entity shall be reduced by any amount such person may collect as indemnification or advancement of expenses from such other corporation, partnership, joint venture, trust, enterprise or non-profit enterprise.

Section 8. Amendment or Repeal; Interpretation.

(a) The provisions of this Article VIII shall constitute a contract between the Corporation, on the one hand, and, on the other hand, each individual who serves or has served as a director or officer of the Corporation (whether before or after the Effective Time), in consideration of such person's performance of such services, and pursuant to this Article VIII the Corporation intends to be legally bound to each such current or former director or officer of the Corporation. With respect to current and former directors and officers of the Corporation, the rights conferred under this Article VIII are present contractual rights and such rights are fully vested, and shall be deemed to have vested fully, immediately upon the Effective Time. With respect to any directors or officers of the Corporation who commence service following the Effective Time, the rights conferred under this provision shall be present contractual rights and such rights shall fully vest, and be deemed to have vested fully, immediately upon such director or officer commencing service as a director or officer of the Corporation. Any repeal or modification of the foregoing provisions of this Article VIII shall not adversely affect any right or protection (i) hereunder of any person in respect of any act or omission occurring prior to the time of such repeal or modification or divancement of expenses to an officer or director of the Corporation in effect prior to the time of such repeal or modification.

(b) Any reference to an officer of the Corporation in this Article IX shall be deemed to refer exclusively to the Chief Executive Officer, President, and Secretary, or other officer of the Corporation appointed by (x) the Board pursuant to the Bylaws of the Corporation ws or (y) an officer to whom the Board has delegated the power to appoint officers pursuant to the Bylaws of the Corporation, and any reference to an officer of any other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise shall be deemed to refer exclusively to an officer appointed by the board of directors (or equivalent governing body) of such other entity pursuant to the certificate of incorporation and bylaws (or equivalent organizational documents) of such other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise. The fact that any person who is or was an employee of the Corporation or an employee of any other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise has been given or has used the title of "Vice President" or any other title that could be construed to suggest or imply that such person is or may be an officer of the Corporation or of such other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise shall not result in such person being constituted as, or being deemed to be, an officer of the Corporation or of such other corporation or of such other Korporation or of such other enterprise for purposes of this Article VIII.

ARTICLE IX DGCL SECTION 203

The Corporation shall not be governed by Section 203 of the DGCL (or any successor provision thereto), and the restrictions contained in Section 203 of the DGCL shall not apply to the Corporation, until immediately following the time at which both of the following conditions exist (if ever): (a) Section 203 of the DGCL by its terms would, but for the provisions of this Article IX, apply to the Corporation; and (b) Indigo does not beneficially own shares of Voting Stock representing at least fifteen percent (15%) of the voting power of all the then outstanding shares of Voting Stock, and the Corporation shall thereafter be governed by Section 203 of the DGCL if and for so long as Section 203 of the DGCL by its terms shall apply to the Corporation.

ARTICLE X EXCLUSIVE FORUM

Unless the Corporation consents in writing to the selection of an alternative forum, the Court of Chancery of the State of Delaware (or, if such court does not have subject matter jurisdiction thereof, the federal district court of the State of Delaware) shall, to the fullest extent permitted by law, be the sole and exclusive forum for: (i) any derivative action or proceeding brought on behalf of the Corporation, (ii) any action asserting a claim for or based on a breach of a fiduciary duty owed by any current or former director, officer, other employee, agent or stockholder of the Corporation to the Corporation or the Corporation's stockholders, including without limitation a claim alleging the

aiding and abetting of such a breach of fiduciary duty, (iii) any action asserting a claim against the Corporation or any current or former director, officer, employee, agent or stockholder of the Corporation arising pursuant to any provision of the General Corporation Law or the Corporation's Certificate of Incorporation or Bylaws or as to which the General Corporation Law confers jurisdiction on the Court of Chancery of the State of Delaware, or (iv) any action asserting a claim related to or involving the Corporation that is governed by the internal affairs doctrine.

Unless the Corporation consents in writing to the selection of an alternative forum, the federal district courts of the United States of America shall, to the fullest extent permitted by law, be the sole and exclusive forum for the resolution of any complaint asserting a cause or causes of action arising under the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder, including all causes of action asserted against any defendant to such complaint. For the avoidance of doubt, this provision is intended to benefit and may be enforced by the Corporation, its officers and directors, the underwriters to any offering giving rise to such complaint, and any other professional or entity whose profession gives authority to a statement made by that person or entity and who has prepared or certified any part of the documents underlying the offering.

Any person or entity purchasing or otherwise acquiring or holding any interest in shares of capital stock of the Corporation shall be deemed to have notice of and consented to the provisions of this Article.

Failure to enforce the foregoing provisions would cause the Corporation irreparable harm, and the Corporation shall be entitled to equitable relief, including injunctive relief and specific performance, to enforce the foregoing provisions.

This exclusive forum provision will not apply to suits brought to enforce any liability or duty created by the Exchange Act of 1934, as amended, or any other claim for which the federal courts have exclusive jurisdiction.

ARTICLE XI AMENDMENT

The Corporation reserves the right to amend, alter, change or repeal any provision contained in this Certificate of Incorporation, in the manner now or hereafter prescribed by statute, and all rights conferred upon stockholders herein are granted subject to this reservation. Notwithstanding any other provisions of this Certificate of Incorporation or any provision of law which might otherwise permit a lesser vote or no vote, but in addition to any affirmative vote of the holders of any particular class or series of the Voting Stock required by law or by the Bylaws of the Corporation or by this Certificate of Incorporation (or by any certificate of designation hereto), any alteration, amendment or repeal of Articles VI, VII, VIII, IX, XI or XII of this Certificate of Incorporation shall require the affirmative vote of (a) a majority of the voting power of the Voting Stock while the Corporation is under Indigo Control and (b) from and after the time that the Corporation is no longer under Indigo Control, at least sixty-six and two-thirds percent (66²/₃%) of the voting power of the Voting Stock.

If any provision or provisions of this Certificate of Incorporation shall be held to be invalid, illegal or unenforceable as applied to any circumstance for any reason whatsoever: (i) the validity, legality and enforceability of such provisions in any other circumstance and of the remaining provisions of this Certificate of Incorporation (including, without limitation, each portion of any paragraph of this Certificate of Incorporation containing any such provision held to be invalid, illegal or unenforceable that is not itself held to be invalid, illegal or unenforceable) shall not, to the fullest extent permitted by applicable law, in any way be affected or impaired thereby and (ii) to the fullest extent permitted by applicable law, the provisions of this Certificate of Incorporation (including, without limitation, each such portion of any paragraph of this Certificate of Incorporation containing any such provision held to be invalid, illegal or unenforceable) shall not, to the fullest extent permitted by applicable law, in any way be affected or impaired thereby and (ii) to the fullest extent permitted by applicable law, the provisions of this Certificate of Incorporation (including, without limitation, each such portion of any paragraph of this Certificate of Incorporation containing any such provision held to be invalid, illegal or unenforceable) shall be construed so as to permit the Corporation to protect its directors, officers, employees and agents from personal liability in respect of their good faith service to or for the benefit of the Corporation to the fullest extent permitted by law.

ARTICLE XII CORPORATE OPPORTUNITIES

To the fullest extent permitted by applicable law, each of Indigo (other than the Corporation and its subsidiaries), any of the Corporation's non-employee directors who are employees, affiliates or consultants of Indigo or its affiliates (other than the Corporation or its subsidiaries) or any of their respective affiliates (each person described in this sentence, a "Specified Party"), have participated (directly or indirectly) in and may, and shall have no duty not to, continue to (A) participate (directly or indirectly) in venture capital, private equity or similar investments in other entities conducting business of any kind, nature or description ("Other Investments") and (B) have interests in, participate with, aid and maintain seats on the boards of directors or similar governing bodies of Other Investments, in each case that may, are or will be competitive with the business of the Corporation and its subsidiaries or in the same or similar lines of business as the Corporation and its subsidiaries, or that could be suitable for the Corporation or its subsidiaries. To the fullest extent permitted by applicable law, the Corporation, on behalf of itself and its subsidiaries, renounces pursuant to Section 122(17) of the DGCL any interest or expectancy of the Corporation and its subsidiaries in, or in being offered an opportunity to participate in, any such Other Investment or any business opportunities for such Other Investments that are from time to time presented to any Specified Party or are business opportunities in which a Specified Party participates or desires to participate, even if the Other Investment or business opportunity is one that the Corporation or its subsidiaries might reasonably be deemed to have pursued or had the ability or desire to pursue if granted the opportunity to do so, and each such Specified Party shall have no duty to communicate or offer any such Other Investment or business opportunity to the Corporation and, to the fullest extent permitted by applicable law, shall not be liable to the Corporation or any of its subsidiaries or any stockholder, including for breach of any fiduciary or other duty, as a director or officer or controlling stockholder or otherwise, and the Corporation shall indemnify each Specified Party against any claim that such Specified Party is liable to the Corporation or its stockholders for breach of any fiduciary duty, by reason of the fact that such Specified Party (a) participates in any such Other Investment or pursues or acquires any such business opportunity, (y) directs any such business

opportunity to another person or (z) fails to present any such Other Investment or business opportunity, or information regarding any such Other Investment or business opportunity, to the Corporation or its subsidiaries, unless, in the case of a Specified Party who is a director or officer of the Corporation, such business opportunity is expressly offered to such Specified Party in writing solely in his or her capacity as a director or officer of the Corporation.

Neither the amendment nor repeal of this Article XII, nor the adoption of any provision of this Certificate of Incorporation or the bylaws of the Corporation, nor, to the fullest extent permitted by applicable law, any modification of law, shall eliminate, reduce or otherwise adversely affect any right or protection of any person granted pursuant hereto existing at, or arising out of or related to any event, act or omission that occurred prior to, the time of such amendment, repeal, adoption or modification (regardless of when any proceeding (or part thereof) relating to such event, act or omission arises or is first threatened, commenced or completed).

If any provision or provisions of this Article XII shall be held to be invalid, illegal or unenforceable as applied to any circumstance for any reason whatsoever, (a) the validity, legality and enforceability of such provisions in any other circumstance and of the remaining provisions of this Article XII (including, without limitation, each portion of any paragraph of this Article XII containing any such provision held to be invalid, illegal or unenforceable) shall not in any way be affected or impaired thereby and (b) to the fullest extent possible, the provisions of this Article XII (including, without limitation, each such portion of any paragraph of the comportion of any paragraph of this Article XII containing any such provision held to be invalid, illegal or unenforceable) shall not in any way be affected or impaired thereby and (b) to the fullest extent possible, the provisions of this Article XII (including, without limitation, each such portion of any paragraph of this Article XII containing any such provision held to be invalid, illegal or unenforceable) shall be construed so as to permit the Corporation to protect its directors, officers, employees and agents from personal liability in respect of their good faith service to or for the benefit of the Corporation to the fullest extent permitted by applicable law.

This Article XII shall not limit any protections or defenses available to, or indemnification or advancement rights of, any director or officer of the Corporation under this Certificate of Incorporation, the Bylaws, applicable law or as may be set forth in individual indemnification agreements with such director or officer.

* *

IN WITNESS WHEREOF, the undersigned have executed this Amended and Restated Certificate of Incorporation on this [_____]^t___], 2021.

[

___]th day of

FRONTIER GROUP HOLDINGS, INC.

By:

Barry L. Biffle President and Chief Executive Officer

[Signature Page to Frontier Group Holdings, Inc. Amended and Restated Certificate of Incorporation]

AMENDED AND RESTATED BYLAWS OF

FRONTIER GROUP HOLDINGS, INC.

(a Delaware corporation)

ARTICLE	I - CORPORATE OFFICES	<u>Page</u> 1
1.1	REGISTERED OFFICE	1
1.2		1
	IL MEETINGS OF STOCKHOLDERS	1
ARTICLE	II - MEETINGS OF STOCKHOLDERS	1
2.1	PLACE OF MEETINGS	1
2.2	ANNUAL MEETING	1
2.3	SPECIAL MEETING	1
2.4	ADVANCE NOTICE PROCEDURES FOR BUSINESS BROUGHT BEFORE A MEETING	2
2.5	ADVANCE NOTICE PROCEDURES FOR NOMINATIONS OF DIRECTORS	6
2.6	ADDITIONAL REQUIREMENTS FOR VALID NOMINATION OF CANDIDATES TO SERVE AS DIRECTOR AND, IF	0
2.7	ELECTED, TO BE SEATED AS DIRECTORS	8
2.7 2.8	NOTICE OF STOCKHOLDERS' MEETINGS QUORUM	10 10
2.0 2.9	ADJOURNED MEETING; NOTICE	10
	CONDUCT OF BUSINESS	10
	VOTING; ELECTION OF DIRECTORS	10
	STOCKHOLDER ACTION BY WRITTEN CONSENT WITHOUT A MEETING	11
	RECORD DATE FOR STOCKHOLDER NOTICE; VOTING; GIVING CONSENTS	11
	PROXIES	12
	LIST OF STOCKHOLDERS ENTITLED TO VOTE	12
	INSPECTORS OF ELECTION	13
	DELIVERY TO THE CORPORATION	13
ARTICLE	III - DIRECTORS	14
3.1	POWERS	14
	NUMBER OF DIRECTORS	14
3.3	ELECTION, QUALIFICATION, CITIZENSHIP, AND TERM OF OFFICE OF DIRECTORS	14
3.4	RESIGNATION AND VACANCIES	14
3.5	PLACE OF MEETINGS; MEETINGS BY TELEPHONE	15
3.6	REGULAR MEETINGS	15
3.7	SPECIAL MEETINGS; NOTICE	15
3.8	QUORUM	16
3.9	BOARD ACTION BY WRITTEN CONSENT WITHOUT A MEETING	16
	FEES AND COMPENSATION OF DIRECTORS	16
3.11	REMOVAL OF DIRECTORS	16
ARTICLE	IV - COMMITTEES	16
4.1	COMMITTEES OF DIRECTORS	16
4.2	COMMITTEE MINUTES	17
4.3	MEETINGS AND ACTION OF COMMITTEES	17
4.4	SUBCOMMITTEES	17

i

4.4 SUBCOMMITTEES

TABLE OF CONTENTS (continued)

	Page
ARTICLE V - OFFICERS	18
 5.1 OFFICERS 5.2 APPOINTMENT OF OFFICERS 5.3 SUBORDINATE OFFICERS 5.4 REMOVAL AND RESIGNATION OF OFFICERS 5.5 VACANCIES IN OFFICES 5.6 REPRESENTATION OF SHARES OF OTHER CORPORATIONS 5.7 AUTHORITY AND DUTIES OF OFFICERS 5.8 LIMITATIONS ON NON-CITIZENS AS OFFICERS 	18 18 18 18 18 18 18 19 19 19 19
ARTICLE VI - RECORDS AND REPORTS	19
6.1 MAINTENANCE AND INSPECTION OF RECORDS6.2 INSPECTION BY DIRECTORS	19 19
ARTICLE VII - GENERAL MATTERS	19
 7.1 EXECUTION OF CORPORATE CONTRACTS AND INSTRUMI 7.2 STOCK CERTIFICATES; PARTLY PAID SHARES 7.3 SPECIAL DESIGNATION ON CERTIFICATES 7.4 LOST CERTIFICATES 7.5 CONSTRUCTION; DEFINITIONS 7.6 DIVIDENDS 7.7 FISCAL YEAR 7.8 SEAL 7.9 TRANSFER OF STOCK 7.10 STOCK TRANSFER AGREEMENTS 7.11 REGISTERED STOCKHOLDERS 7.12 WAIVER OF NOTICE 	ENTS 19 20 20 20 20 21 21 21 21 21 21 21 21 21 21 22 22
ARTICLE VIII - NOTICE BY ELECTRONIC TRANSMISSION	22
 ARTICLE IX - INDEMNIFICATION 9.1 INDEMNIFICATION OF DIRECTORS AND OFFICERS 9.2 INDEMNIFICATION OF OTHERS 9.3 PREPAYMENT OF EXPENSES 9.4 NON-EXCLUSIVITY OF RIGHTS 9.5 INSURANCE 9.6 OTHER INDEMNIFICATION 9.7 CONTINUATION OF INDEMNIFICATION 9.8 AMENDMENT OR REPEAL; INTERPRETATION 	23 23 23 23 24 24 24 24 24 24
ARTICLE X - LIMITATIONS OF OWNERSHIP BY NON-CITIZENS	25
10.1 EQUITY SECURITIES10.2 NON-U.S. CITIZEN VOTING AND OWNERSHIP LIMITATION	25 S 25

ii

TABLE OF CONTENTS (continued)

	Page	
10.3 FOREIGN STOCK RECORD	26	
10.4 REGISTRATION OF SHARES	26	
10.5 CERTIFICATION OF SHARES	27	
10.6 LEGEND	27	
ARTICLE XI—AMENDMENTS; WAIVERS	28	
ARTICLE XII—FORUM SELECTION		

iii

AMENDED AND RESTATED BYLAWS OF FRONTIER GROUP HOLDINGS, INC.

FROM HER GROUP HOLDINGS, INC.

ADOPTED AS OF _____, 2021

ARTICLE I—CORPORATE OFFICES

1.1 REGISTERED OFFICE.

The registered office of Frontier Group Holdings, Inc. (the "<u>Corporation</u>") shall be fixed in the Corporation's certificate of incorporation, as the same may be amended from time to time (the "<u>Certificate of Incorporation</u>").

1.2 OTHER OFFICES.

The Corporation's board of directors (the "<u>Board</u>") may at any time establish other offices at any place or places where the Corporation is qualified to do business.

ARTICLE II—MEETINGS OF STOCKHOLDERS

2.1 PLACE OF MEETINGS.

Meetings of stockholders shall be held at any place, within or outside the State of Delaware, designated by the Board. The Board may, in its sole discretion, determine that a meeting of stockholders shall not be held at any place, but may instead be held solely by means of remote communication as authorized by Section 211(a)(2) of the Delaware General Corporation Law (the "<u>DGCL</u>"). In the absence of any such designation or determination, stockholders' meetings shall be held at the Corporation's principal executive office.

2.2 ANNUAL MEETING.

The Board shall designate the date and time of the annual meeting. At the annual meeting, directors shall be elected and other business properly brought before the meeting in accordance with Section 2.4 of this Article II may be transacted. The Corporation may postpone, reschedule or cancel any annual meeting of stockholders previously scheduled by the Board.

2.3 SPECIAL MEETING.

A special meeting of the stockholders may be called at any time by the Chair of the Board or by the Secretary of the Corporation upon direction of the Board pursuant to a resolution adopted by a majority of the entire Board, but such special meetings may not be called by any other person or persons; <u>provided</u>, <u>however</u>, that notwithstanding the foregoing, until such time as Indigo Frontier Holdings Company, LLC, or any successor thereto by merger, consolidation, acquisition of all or substantially all assets or otherwise by operation of law or any of its Affiliates (as such term is defined in Rule 12b-2 promulgated under the Securities Exchange Act of 1934, as amended) ("<u>Indigo</u>") ceases to beneficially own shares of Common Stock representing at least a majority of the voting power ("<u>Indigo Control</u>") of all the then outstanding shares of capital stock of the Corporation entitled to vote at an election of directors (the "<u>Voting Stock</u>"), special meetings of stockholders of the Corporation may be called at any time by the holders of a majority of the voting power of the Voting Stock.

No business may be transacted at such special meeting other than the business specified in such notice to stockholders. Nothing contained in this paragraph of this Section 2.3 shall be construed as limiting, fixing, or affecting the time when a meeting of stockholders called by action of the Board may be held. The Corporation may postpone, reschedule or cancel any special meeting of stockholders previously scheduled by the Chair of the Board or by the Secretary of the Corporation upon direction of the Board.

2.4 ADVANCE NOTICE PROCEDURES FOR BUSINESS BROUGHT BEFORE A MEETING.

(i) At an annual meeting of the stockholders, only such business shall be conducted as shall have been properly brought before the meeting. To be properly brought before an annual meeting, business must be (a) specified in a notice of meeting given by or at the direction of the Board, (b) if not specified in a notice of meeting, otherwise brought before the meeting by or at the direction of the Board or the Chair of the Board, (c) be brought before the meeting by or at the direction of Indigo while the Corporation is under Indigo Control or (d) otherwise properly brought before the meeting by a stockholder present in person who (A)(1) was a record owner of shares of the Corporation both at the time of giving the notice provided for in this Section 2.4 and at the time of the meeting, (2) is entitled to vote at the meeting and (3) has complied with this Section 2.4 in all applicable respects, or (B) properly made such proposal in accordance with Rule 14a-8 under the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder (as so amended and inclusive of such rules and regulations, the "Exchange Act"). The foregoing clause (d) shall be the exclusive means for a stockholder (other than Indigo while the Corporation is under Indigo Control) to propose business to be brought before an annual meeting of the stockholders. The only matters that may be brought before a special meeting are the matters specified in the notice of meeting given by or at the direction of the person calling the meeting pursuant to Section 2.3 of these Bylaws, and stockholders shall not be permitted to propose business to be brought before a special meeting of the stockholders. For purposes of this Section 2.4, "present in person" shall mean that the stockholder proposing that the business be brought before the annual meeting of the Corporation, or, if the proposing stockholder is not an individual, a qualified representative of such proposing stockholder, appear at such annual meeting. A "qualified representative" of such proposing stockholder shall be a duly authorized officer, manager or partner of such stockholder or any other person authorized by a writing executed by such stockholder or an electronic transmission delivered by such stockholder to act for such stockholder as proxy at the meeting of stockholders and such person must produce such writing or electronic transmission, or a reliable reproduction of the writing or electronic transmission, at the meeting of stockholders. Stockholders seeking to nominate persons for election to the Board must comply with Section 2.5 and Section 2.6 of these Bylaws, and this Section 2.4 shall not be applicable to nominations except as expressly provided in Section 2.5 and Section 2.6 of these Bylaws.

(ii) For business to be properly brought before an annual meeting by a stockholder pursuant to Section 2.4(a)(iv), the stockholder must (a) provide Timely Notice (as defined below) thereof in writing and in proper form to the Secretary of the Corporation and (b) provide any updates or supplements to such notice at the times and in the forms required by this Section 2.4. To be timely, a stockholder's notice must be delivered to, or mailed and received at, the principal executive offices of the Corporation not less than ninety (90) days nor more than one hundred twenty (120) days prior to the one-year anniversary of the preceding year's annual meeting which, in the case of the first annual meeting of stockholders following the

closing of the Corporation's initial underwritten public offering of common stock, the date of the preceding year's annual meeting shall be deemed to be June 1; <u>provided</u>, <u>however</u>, that if the date of the annual meeting is more than thirty (30) days before or more than sixty (60) days after such anniversary date, notice by the stockholder to be timely must be so delivered, or mailed and received, not later than the ninetieth (90th) day prior to such annual meeting or, if later, the tenth (10th) day following the day on which public disclosure of the date of such annual meeting was first made by the Corporation (such notice within such time periods, "<u>Timely Notice</u>"). In no event shall any adjournment or postponement of an annual meeting or the announcement thereof commence a new time period for the giving of Timely Notice as described above.

(iii) To be in proper form for purposes of this Section 2.4, a stockholder's notice to the Secretary shall set forth:

(a) As to each Proposing Person (as defined below), (A) the name and address of such Proposing Person (including, if applicable, the name and address that appear on the Corporation's books and records); and (B) the class or series and number of shares of the Corporation that are, directly or indirectly, owned of record or beneficially owned (within the meaning of Rule 13d-3 under the Exchange Act) by such Proposing Person, except that such Proposing Person shall in all events be deemed to beneficially own any shares of any class or series of the Corporation as to which such Proposing Person has a right to acquire beneficial ownership at any time in the future (the disclosures to be made pursuant to the foregoing clauses (A) and (B) are referred to as "Stockholder Information");

(b) As to each Proposing Person, (A) the full notional amount of any securities that, directly or indirectly, underlie any "derivative security" (as such term is defined in Rule 16a-1(c) under the Exchange Act) that constitutes a "call equivalent position" (as such term is defined in Rule 16a-1(b) under the Exchange Act) ("Synthetic Equity Position") and that is, directly or indirectly, held or maintained by such Proposing Person with respect to any shares of any class or series of shares of the Corporation; provided that, for the purposes of the definition of "Synthetic Equity Position," the term "derivative security" shall also include any security or instrument that would not otherwise constitute a "derivative security" as a result of any feature that would make any conversion, exercise or similar right or privilege of such security or instrument becoming determinable only at some future date or upon the happening of a future occurrence, in which case the determination of the amount of securities into which such security or instrument would be convertible or exercisable shall be made assuming that such security or instrument is immediately convertible or exercisable at the time of such determination; and, provided, further, that any Proposing Person satisfying the requirements of Rule 13d-1(b)(1) under the Exchange Act (other than a Proposing Person that so satisfies Rule 13d-1(b)(1) under the Exchange Act solely by reason of Rule 13d-1(b)(1)(ii)(E)) shall not be deemed to hold or maintain the notional amount of any securities that underlie a Synthetic Equity Position held by such Proposing Person as a hedge with respect to a bona fide derivatives trade or position of such Proposing Person arising in the ordinary course of such Proposing Person's business as a derivatives dealer, (B) any rights to dividends on the shares of any class or series of shares of the Corporation owned beneficially by such Proposing Person that are separated or separable from the underlying shares of the Corporation, (C)(x) if such Proposing Person is (i) a general or limited partnership, syndicate or other group, the identity of each general partner and each person who functions as a general partner of the general or limited partnership, each member of the syndicate or group and each person controlling the general partner or member, (ii) a corporation or a limited liability company, the identity of each officer and each person who functions as an officer of

the corporation or limited liability company, each person controlling the corporation or limited liability company and each officer, director, general partner and person who functions as an officer, director or general partner of any entity ultimately in control of the corporation or limited liability company or (iii) a trust, any trustee of such trust (each such person or persons set forth in the preceding clauses (i), (ii) and (iii), a "Responsible Person"), any fiduciary duties owed by such Responsible Person to the equity holders or other beneficiaries of such Proposing Person and any material interests or relationships of such Responsible Person that are not shared generally by other record or beneficial holders of the shares of any class or series of the Corporation and that reasonably could have influenced the decision of such Proposing Person to propose such business to be brought before the meeting, and (v) if such Proposing Person is a natural person, any material interests or relationships of such natural person that are not shared generally by other record or beneficial holders of the shares of any class or series of the Corporation and that reasonably could have influenced the decision of such Proposing Person to propose such business to be brought before the meeting, (D) any material shares or any Synthetic Equity Position in any principal competitor of the Corporation in any principal industry of the Corporation held by such Proposing Persons, (E) a summary of any material discussions regarding the business proposed to be brought before the meeting (x) between or among any of the Proposing Persons or (y) between or among any Proposing Person and any other record or beneficial holder of the shares of any class or series of the Corporation (including their names), (F) any material pending or threatened legal proceeding in which such Proposing Person is a party or material participant involving the Corporation or any of its officers or directors, or any affiliate of the Corporation, (G) any other material relationship between such Proposing Person, on the one hand, and the Corporation, any affiliate of the Corporation or any principal competitor of the Corporation, on the other hand, (H) any direct or indirect material interest in any material contract or agreement of such Proposing Person with the Corporation, any affiliate of the Corporation or any principal competitor of the Corporation (including, in any such case, any employment agreement, collective bargaining agreement or consulting agreement), (I) a representation that such Proposing Person intends or is part of a group which intends to deliver a proxy statement or form of proxy to holders of at least the percentage of the Corporation's outstanding capital stock required to approve or adopt the proposal or otherwise solicit proxies from stockholders in support of such proposal and (J) any other information relating to such Proposing Person that would be required to be disclosed in a proxy statement or other filing required to be made in connection with solicitations of proxies or consents by such Proposing Person in support of the business proposed to be brought before the meeting pursuant to Section 14(a) of the Exchange Act (the disclosures to be made pursuant to the foregoing clauses (A) through (J) are referred to as "Disclosable Interests"); provided, however, that Disclosable Interests shall not include any such disclosures with respect to the ordinary course business activities of any broker, dealer, commercial bank, trust company or other nominee who is a Proposing Person solely as a result of being the stockholder directed to prepare and submit the notice required by these Bylaws on behalf of a beneficial owner; and

(c) As to each item of business that the stockholder proposes to bring before the annual meeting, (A) a brief description of the business desired to be brought before the annual meeting, the reasons for conducting such business at the annual meeting and any material interest in such business of each Proposing Person, (B) the text of the proposal or business (including the text of any resolutions proposed for consideration and in the event that such business includes a proposal to amend the Bylaws of the Corporation, the language

of the proposed amendment), (C) a reasonably detailed description of all agreements, arrangements and understandings between or among any of the Proposing Persons or between or among any Proposing Person and any other person or entity (including their names) in connection with the proposal of such business by such stockholder and (D) any other information relating to such item of business that would be required to be disclosed in a proxy statement or other filing required to be made in connection with solicitations of proxies in support of the business proposed to be brought before the meeting pursuant to Section 14(a) of the Exchange Act; <u>provided</u>, <u>however</u>, that the disclosures required by this Section 2.4(iii) shall not include any disclosures with respect to any broker, dealer, commercial bank, trust company or other nominee who is a Proposing Person solely as a result of being the stockholder directed to prepare and submit the notice required by these Bylaws on behalf of a beneficial owner.

(iv) For purposes of this Section 2.4, the term "Proposing Person" shall mean (a) the stockholder providing the notice of business proposed to be brought before an annual meeting, (b) the beneficial owner or beneficial owners, if different, on whose behalf the notice of the business proposed to be brought before the annual meeting is made, (c) any participant (as defined in paragraphs (a)(ii)-(vi) of Instruction 3 to Item 4 of Schedule 14A) with such stockholder in such solicitation or associate (within the meaning of Rule 12b-2 under the Exchange Act for the purposes of these Bylaws) of such stockholder or beneficial owner and (d) any other person with whom such stockholder or such beneficial owner (or any of their respective associates or other participants in such solicitation) is Acting in Concert. A person shall be deemed to be "Acting in Concert" with another person for purposes of these Bylaws if such person knowingly acts (whether or not pursuant to an express agreement, arrangement or understanding) in concert or in parallel with, or towards a common goal with such other person, relating to changing or influencing the control of the Corporation or in connection with or as a participant in any transaction having that purpose or effect, where (A) each person is conscious of the other person's conduct, and this awareness is an element in their decision-making processes, and (B) at least one additional factor suggests that such persons intend to act in concert or in parallel, which such additional factors may include, without limitation, exchanging information (whether publicly or privately), attending meetings, conducting discussions or making or soliciting invitations to act in concert or in parallel; provided, that a person shall not be deemed to be Acting in Concert with any other person solely as a result of the solicitation or receipt of (1) revocable proxies or consents from such other person in response to a solicitation made pursuant to, and in accordance with, Section 14(a) of the Exchange Act by way of a proxy or consent solicitation statement filed on Schedule 14A or (2) tenders of securities from such other person in a public tender or exchange offer made pursuant to, and in accordance with, Section 14(d) of the Exchange Act by means of a tender offer statement filed on Schedule TO. A person Acting in Concert with another person shall be deemed to be Acting in Concert with any third party who is also Acting in Concert with such other person.

(v) A Proposing Person shall update and supplement its notice to the Corporation of its intent to propose business at an annual meeting, if necessary, so that the information provided or required to be provided in such notice pursuant to this Section 2.4 shall be true and correct as of the record date for stockholders entitled to vote at the meeting and as of the date that is ten (10) business days prior to the meeting or any adjournment or postponement thereof, and such update and supplement shall be delivered to, or mailed and received by, the Secretary at the principal executive offices of the Corporation not later than five (5) business days after the record date for stockholders entitled to vote at the meeting (in the case of the update and supplement required to be made as of such record date), and not later than eight (8) business days prior to the date for the meeting or, if practicable, any adjournment or postponement thereof (and, if not practicable, on the first practicable date prior to the date to which the meeting has been adjourned or postponed) (in the case of the update and supplement required to be made as of ten (10) business days prior to

the meeting or any adjournment or postponement thereof). For the avoidance of doubt, the obligation to update and supplement as set forth in this paragraph or any other Section of these Bylaws shall not limit the Corporation's rights with respect to any deficiencies in any notice provided by a stockholder, extend any applicable deadlines hereunder or enable or be deemed to permit a stockholder who has previously submitted notice hereunder to amend or update any proposal or to submit any new proposal, including by changing or adding matters, business or resolutions proposed to be brought before a meeting of the stockholders.

(vi) Notwithstanding anything in these Bylaws to the contrary, no business shall be conducted at an annual meeting that is not properly brought before the meeting in accordance with this Section 2.4. The presiding officer of the meeting shall, if the facts warrant, determine that the business was not properly brought before the meeting in accordance with this Section 2.4, and if he or she should so determine, he or she shall so declare to the meeting and any such business not properly brought before the meeting shall not be transacted.

(vii) This Section 2.4 is expressly intended to apply to any business proposed to be brought before an annual meeting of stockholders, other than any proposal made in accordance with Rule 14a-8 under the Exchange Act and included in the Corporation's proxy statement. In addition to the requirements of this Section 2.4 with respect to any business proposed to be brought before an annual meeting, each Proposing Person shall comply with all applicable requirements of the Exchange Act with respect to any such business. Nothing in this Section 2.4 shall be deemed to affect the rights of stockholders to request inclusion of proposals in the Corporation's proxy statement pursuant to Rule 14a-8 under the Exchange Act.

(viii) For purposes of these Bylaws, "<u>public disclosure</u>" shall mean disclosure in a press release reported by a national news service or in a document publicly filed by the Corporation with the Securities and Exchange Commission pursuant to Sections 13, 14 or 15(d) of the Exchange Act.

2.5 ADVANCE NOTICE PROCEDURES FOR NOMINATIONS OF DIRECTORS.

(i) Nominations of any person for election to the Board at an annual meeting or at a special meeting (but only if the election of directors is a matter specified in the notice of meeting given by or at the direction of the person calling such special meeting) may be made at such meeting only (a) by or at the direction of the Board, including by any committee or persons authorized to do so by the Board or these Bylaws, (b) by or at the direction of Indigo while the Corporation is under Indigo Control or (c) by a stockholder present in person who (A) was a beneficial owner of shares of the Corporation both at the time of giving the notice provided for in this Section 2.5 and at the time of the meeting, (B) is entitled to vote at the meeting and (C) has complied with this Section 2.5 and Section 2.6 as to such notice and nomination. The foregoing clause (b) shall be the exclusive means for a stockholder (other than Indigo while the Corporation is under Indigo Control) to make any nomination of a person or persons for election to the Board at an annual meeting or special meeting. For purposes of this Section 2.5, "present in person" shall mean that the stockholder proposing that the business be brought before the meeting of the Corporation, or, if the proposing stockholder is not an individual, a qualified representative of such stockholder, appear at such meeting. A "qualified representative" of such proposing stockholder or an electronic transmission delivered by such stockholder to act for such stockholder as proxy at the meeting of stockholders and such person must produce such writing or electronic transmission, or a reliable reproduction of the writing or electronic transmission, at the meeting of stockholders. The foregoing clause (i) shall be the exclusive means for a stockholder to make any nomination of a person or persons for election to the Board at an annual meeting or special meeting.

(ii) For a stockholder to make any nomination of a person or persons for election to the Board at an annual meeting, the stockholder must (a) provide Timely Notice (as defined in Section 2.4(ii) of these Bylaws) thereof in writing and in proper form to the Secretary of the Corporation, (b) provide the information with respect to such stockholder and its proposed nominee as required by this Section 2.5, and (c) provide any updates or supplements to such notice at the times and in the forms required by this Section 2.5. If the election of directors is a matter specified in the notice of meeting given by or at the direction of the person calling such special meeting, then for a stockholder to make any nomination of a person or persons for election to the Board at a special meeting, the stockholder must (a) provide timely notice thereof in writing and in proper form to the Secretary of the Corporation at the principal executive offices of the Corporation, (b) provide the information with respect to such stockholder and its proposed nominee as required by this Section 2.5, and (c) provide any updates or supplements to such notice at the times and in the forms required by this Section 2.5. To be timely, a stockholder's notice for nominations to be made at a special meeting must be delivered to, or mailed and received at, the principal executive offices of the Corporation not earlier than the one hundred twentieth (120th) day prior to such special meeting and not later than the ninetieth (90th) day prior to such special meeting or, if later, the tenth (10th) day following the day on which public disclosure (as defined in Section 2.4(ix) of these Bylaws) of the date of such special meeting was first made. In no event shall any adjournment or postponement of an annual meeting or special meeting or the announcement thereof commence a new time period for the giving of a stockholder's notice as described above. In no event may a Nominating Person provide Timely Notice with respect to a greater number of director candidates than are subject to election by shareholders at the applicable meeting. If the Corporation shall, subsequent to such notice, increase the number of directors subject to election at the meeting, such notice as to any additional nominees shall be due on the later of (i) the conclusion of the time period for Timely Notice, (ii) the date set forth in Section 2.4(b)(ii) or (iii) the tenth day following the date of public disclosure (as defined in Section 2.4) of such increase.

(iii) To be in proper form for purposes of this Section 2.5, a stockholder's notice to the Secretary shall set forth:

(a) As to each Nominating Person (as defined below), the Stockholder Information (as defined herein, except that for purposes of this Section 2.5, the term "Nominating Person" shall be substituted for the term "Proposing Person" in all places it appears in Section 2.4(iii)(a));

(b) As to each Nominating Person, any Disclosable Interests (as defined herein, except that for purposes of this Section 2.5 the term "Nominating Person" shall be substituted for the term "Proposing Person" in all places it appears in Section 2.4(iii)(b) and the disclosure with respect to the business to be brought before the meeting in Section 2.4(iii)(b) shall be made with respect to the election of directors at the meeting);

(c) As to each person whom a Nominating Person proposes to nominate for election as a director, (A) all information with respect to such proposed nominee that would be required to be set forth in a stockholder's notice pursuant to this Section 2.5 if such proposed nominee were a Nominating Person, (B) all information relating to such proposed nominee that is required to be disclosed in a proxy statement or other filings required to be made in connection with solicitations of proxies for election of directors in a contested election pursuant to Section 14(a) under the Exchange Act (including such proposed nominee's written consent to being named in the proxy statement as a nominee and to serving as a director if elected), (C) a description of any direct or indirect material interest in any material contract or agreement between or among any Nominating Person, on the one hand, and each

proposed nominee or his or her respective associates or any other participants in such solicitation, on the other hand, including, without limitation, all information that would be required to be disclosed pursuant to Item 404 under Regulation S-K if such Nominating Person were the "registrant" for purposes of such rule and the proposed nominee were a director or executive officer of such registrant (the disclosures to be made pursuant to the foregoing clauses (A) through (C) are referred to as "<u>Nominee Information</u>"), and (D) a completed and signed questionnaire, representation and agreement as provided in Section 2.6; and

(d) The Corporation may require any proposed nominee to furnish such other information (A) as may reasonably be required by the Corporation to determine the eligibility of such proposed nominee to serve as an independent director of the Corporation in accordance with the Corporation's Corporate Governance Guidelines or (B) that could be material to a reasonable stockholder's understanding of the independence or lack of independence of such proposed nominee.

(iv) For purposes of this Section 2.5, the term "<u>Nominating Person</u>" shall mean (a) the stockholder providing the notice of the nomination proposed to be made at the meeting, (b) the beneficial owner or beneficial owners, if different, on whose behalf the notice of the nomination proposed to be made at the meeting is made and (c) any associate of such stockholder or beneficial owner or any other participant in such solicitation.

(v) A stockholder providing notice of any nomination proposed to be made at a meeting shall further update and supplement such notice, if necessary, so that the information provided or required to be provided in such notice pursuant to this Section 2.5 shall be true and correct as of the record date for stockholders entitled to vote at the meeting and as of the date that is ten (10) business days prior to the meeting or any adjournment or postponement thereof, and such update and supplement shall be delivered to, or mailed and received by, the Secretary at the principal executive offices of the Corporation not later than five (5) business days after the record date for stockholders entitled to vote at the meeting (in the case of the update and supplement required to be made as of such record date), and not later than eight (8) business days prior to the date for the meeting or, if practicable, any adjournment or postponement thereof (and, if not practicable, on the first practicable date prior to the date to which the meeting has been adjourned or postponement thereof). For the avoidance of doubt, the obligation to update and supplement as set forth in this paragraph or any other Section of these Bylaws shall not limit the Corporation's rights with respect to any deficiencies in any notice provided by a stockholder, extend any applicable deadlines hereunder or enable or be deemed to permit a stockholder who has previously submitted notice hereunder to amend or update any nomination or to submit any new nomination.

(vi) In addition to the requirements of this Section 2.5 with respect to any nomination proposed to be made at a meeting, each Nominating Person shall comply with all applicable requirements of the Exchange Act with respect to any such nominations.

2.6 ADDITIONAL REQUIREMENTS FOR VALID NOMINATION OF CANDIDATES TO SERVE AS DIRECTOR AND, IF ELECTED, TO BE SEATED AS DIRECTORS.

(i) To be eligible to be a nominee for election as a director of the Corporation at an annual or special meeting, the proposed nominee must be nominated in the manner prescribed in Section 2.5 and must deliver (in accordance with the time period prescribed for delivery in a notice to such proposed

nominee given by or on behalf of the Board), to the Secretary at the principal executive offices of the Corporation, (a) a completed written questionnaire (in a form provided by the Corporation) with respect to the background, qualifications, citizenship, stock ownership and independence of such proposed nominee and (b) a written representation and agreement (in form provided by the Corporation) that such proposed nominee (A) is not and, if elected as a director during his or her term of office, will not become a party to (1) any agreement, arrangement or understanding with, and has not given and will not give any commitment or assurance to, any person or entity as to how such proposed nominee, if elected as a director of the Corporation, will act or vote on any issue or question (a "<u>Voting Commitment</u>") or (2) any Voting Commitment that could limit or interfere with such proposed nominee's ability to comply, if elected as a director of the Corporation, with such proposed nominee's fiduciary duties under applicable law, (B) is not, and will not become a party to, any agreement, arrangement or understanding with any person or entity other than the Corporation, will comply with all applicable corporate governance, conflict of interest, confidentiality, stock ownership and trading and other policies and guidelines of the Corporation applicable to directors and in effect during such person's term in office as a director (and, if requested by any proposed nominee, the Secretary of the Corporation shall provide to such proposed nominee all such policies and guidelines then in effect), (D) if elected as director of the Corporation, intends to serve the entire term until the next meeting at which such proposed nominee would face re-election and (E) consents to being named as a nominee in the Corporation's proxy statement pursuant to Rule 14a-4(d) under the Exchange Act and any associated proxy card of the Corporation and agrees to serve if elected as a director.

(ii) In addition to the requirements of this Section 2.5 with respect to any nomination proposed to be made at a meeting, each Proposing Person shall comply with all applicable requirements of the Exchange Act with respect to any such nominations.

(iii) A proposed nominee for nomination as a director shall further update and supplement the materials delivered pursuant to this Section 2.6, if necessary, so that the information provided or required to be provided pursuant to this Section 2.6 shall be true and correct as of the record date for stockholders entitled to vote at the meeting and as of the date that is ten (10) business days prior to the meeting or any adjournment or postponement thereof, and such update and supplement shall be delivered to, or mailed and received by, the Secretary at the principal executive offices of the Corporation (or any other office specified by the Corporation in any public announcement) not later than five (5) business days after the record date for stockholders entitled to vote at the meeting (in the case of the update and supplement required to be made as of such record date), and not later than eight (8) business days prior to the date for the meeting or, if practicable, any adjournment or postponement thereof (and, if not practicable, on the first practicable date prior to the date to which the meeting has been adjourned or postponed) (in the case of the update and supplement required to be made as of ten (10) business days prior to the meeting or any adjournment or postponement thereof). For the avoidance of doubt, the obligation to update and supplement as set forth in this paragraph or any other Section of these Bylaws shall not limit the Corporation's rights with respect to any deficiencies in any notice provided by a stockholder, extend any applicable deadlines hereunder or enable or be deemed to permit a stockholder who has previously submitted notice hereunder to amend or update any proposal or to submit any new proposal, including by changing or adding nominees, matters, business or resolutions proposed to be brought before a meeting of the stockholders.

(iv) No proposed nominee shall be eligible for nomination as a director of the Corporation unless such proposed nominee and the Nominating Person seeking to place such proposed nominee's name in nomination have complied with Section 2.5 and this Section 2.6, as applicable. The presiding officer at the meeting shall, if the facts warrant, determine that a nomination was not properly made

in accordance with this with Section 2.5 and this Section 2.6, and if he or she should so determine, he or she shall so declare such determination to the meeting, the defective nomination shall be disregarded and any ballots cast for the proposed nominee in question (but in the case of any form of ballot listing other qualified nominees, only the ballots cast for the nominee in question) shall be null and void *ab initio*, and of no force or effect.

2.7 NOTICE OF STOCKHOLDERS' MEETINGS.

Unless otherwise provided by law, the Certificate of Incorporation or these Bylaws, the notice of any meeting of stockholders shall be sent or otherwise given in accordance with either Section 2.7 or Section 8.1 of these Bylaws not less than ten (10) nor more than sixty (60) days before the date of the meeting to each stockholder entitled to vote at such meeting. The notice shall specify the place, if any, date and hour of the meeting, the means of remote communication, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such meeting, and, in the case of a special meeting, the purpose or purposes for which the meeting is called.

2.8 QUORUM.

Unless otherwise provided by law, the Certificate of Incorporation or these Bylaws, the holders of a majority in voting power of the stock issued and outstanding and entitled to vote, present in person, or by remote communication, if applicable, or represented by proxy, shall constitute a quorum for the transaction of business at all meetings of the stockholders. A quorum, once established at a meeting, shall not be broken by the withdrawal of enough votes to leave less than a quorum. If, however, a quorum is not present or represented at any meeting of the stockholders, then either (i) the chair of the meeting or (ii) a majority in voting power of the stockholders entitled to vote at the meeting, present in person, or by remote communication, if applicable, or represented by proxy, shall have power to recess the meeting or adjourn the meeting from time to time in the manner provided in Section 2.9 of these Bylaws until a quorum is present or represented. At any recessed or adjourned meeting at which a quorum is present or represented, any business may be transacted that might have been transacted at the meeting as originally noticed.

2.9 ADJOURNED MEETING; NOTICE.

When a meeting is adjourned to another time or place, unless these Bylaws otherwise require, notice need not be given of the adjourned meeting if the time, place, if any, thereof, and the means of remote communications, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such adjourned meeting are announced at the meeting at which the adjournment is taken. At the adjourned meeting, the Corporation may transact any business which might have been transacted at the original meeting. If the adjournment is for more than thirty (30) days, or if after the adjournment a new record date is fixed for the adjourned meeting, a notice of the adjourned meeting shall be given to each stockholder of record entitled to vote at the meeting.

2.10 CONDUCT OF BUSINESS.

The date and time of the opening and the closing of the polls for each matter upon which the stockholders will vote at a meeting shall be announced at the meeting by the chair of the meeting. The Board may adopt by resolution such rules and regulations for the conduct of the meeting of stockholders as it shall deem appropriate. Except to the extent inconsistent with such rules and regulations as adopted by the Board, the chair of any meeting of stockholders shall have the right and authority to convene and (for any or no reason) to recess and/or adjourn the meeting, to prescribe such rules, regulations and procedures (which need

not be in writing) and to do all such acts as, in the judgment of such chair of the meeting, are appropriate for the proper conduct of the meeting. Such rules, regulations or procedures, whether adopted by the Board or prescribed by the chair of the meeting, may include, without limitation, the following: (i) the establishment of an agenda or order of business for the meeting; (ii) rules and procedures for maintaining order at the meeting and the safety of those present (including, without limitation, rules and procedures for removal of disruptive persons from the meeting); (iii) limitations on attendance at or participation in the meeting shall determine; (iv) restrictions on entry to the meeting after the time fixed for the commencement thereof; and (v) limitations on the time allotted to questions or comments by participants. The chair of any meeting of stockholders, in addition to making any other determinations that may be appropriate to the conduct of the meeting (including, without limitation, determinations with respect to the administration and/or interpretation of any of the rules, regulations or procedures of the meeting that a matter or business was not properly brought before the meeting and declare to the meeting shall so declare to the meeting and any such matter or business not properly brought before the meeting shall not be transacted or considered. Unless and to the extent determined by the Board or the chair of the meeting, meetings of stockholders shall not be required to be held in accordance with the rules of parliamentary procedure.

2.11 VOTING; ELECTION OF DIRECTORS.

The stockholders entitled to vote at any meeting of stockholders shall be determined in accordance with the provisions of Section 2.13 of these Bylaws, subject to Section 217 (relating to voting rights of fiduciaries, pledgors and joint owners of stock) and Section 218 (relating to voting trusts and other voting agreements) of the DGCL.

Except as may be otherwise provided in the Certificate of Incorporation or these Bylaws, each stockholder entitled to vote at a meeting of stockholders shall be entitled to one (1) vote for each share of capital stock held by such stockholder which has voting power upon the matter in question.

At all duly called or convened meetings of stockholders, at which a quorum is present, for the election of directors, a plurality of the votes cast shall be sufficient to elect a director. Except as otherwise provided by the Certificate of Incorporation, these Bylaws, the rules or regulations of any stock exchange applicable to the Corporation, or applicable law or pursuant to any regulation applicable to the Corporation or its securities, all other elections and questions presented to the stockholders at a duly called or convened meeting, at which a quorum is present, shall be decided by the majority of the votes present and in person or by proxy and entitled to vote thereon and shall be valid and binding upon the Corporation.

2.12 STOCKHOLDER ACTION BY WRITTEN CONSENT WITHOUT A MEETING.

Except as may be provided in a resolution or resolutions of the Board providing for any series of Preferred Stock, any action required or permitted to be taken by the stockholders of the Corporation may be effected by an action by consent in lieu of a meeting with the approval of the holders of outstanding capital stock having not less than the minimum voting power that would be necessary to authorize or take such action at a meeting at which all shares of capital stock entitled to vote thereon were present and voted; <u>provided</u>, that from and after such time that the Corporation is no longer under Indigo Control, any action required or permitted to be taken by the stockholders of the Corporation must be effected only at a duly called annual or special meeting of such stockholders of the Corporation and may not be effected by any consent in lieu of a meeting by such stockholders. For the avoidance of doubt, at any time when action by consent in lieu of a meeting is permissible pursuant to this Section 2.12, the notice requirements of Sections 2.4 and 2.5 of these Bylaws shall not apply.

2.13 RECORD DATE FOR STOCKHOLDER NOTICE; VOTING; GIVING CONSENTS.

In order that the Corporation may determine the stockholders entitled to notice of or to vote at any meeting of stockholders or any adjournment thereof, or entitled to receive payment of any dividend or other distribution or allotment of any rights, or entitled to exercise any rights in respect of any change, conversion or exchange of stock or for the purpose of any other lawful action, the Board may fix, in advance, a record date, which record date shall not precede the date on which the resolution fixing the record date is adopted and which shall not be more than sixty (60) nor less than ten (10) days before the date of such meeting, nor more than sixty (60) days prior to any other such action.

If the Board does not so fix a record date:

(i) The record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be at the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the day next preceding the day on which the meeting is held.

(ii) The record date for determining stockholders for any other purpose shall be at the close of business on the day on which the Board adopts the resolution relating thereto.

A determination of stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting; <u>provided</u>, <u>however</u>, that the Board may fix a new record date for the adjourned meeting.

2.14 PROXIES.

Each stockholder entitled to vote at a meeting of stockholders may authorize another person or persons to act for such stockholder by proxy authorized by an instrument in writing or by a transmission permitted by law filed in accordance with the procedure established for the meeting, but no such proxy shall be voted or acted upon after three (3) years from its date, unless the proxy provides for a longer period. The revocability of a proxy that states on its face that it is irrevocable shall be governed by the provisions of Section 212 of the DGCL. A proxy may be in the form of a telegram, cablegram or other means of electronic transmission which sets forth or is submitted with information from which it can be determined that the telegram, cablegram or other means of electronic transmission was authorized by the stockholder.

2.15 LIST OF STOCKHOLDERS ENTITLED TO VOTE.

The Corporation shall prepare, at least ten (10) days before every meeting of stockholders, a complete list of the stockholders entitled to vote at the meeting, arranged in alphabetical order, and showing the address of each stockholder and the number of shares registered in the name of each stockholder. The Corporation shall not be required to include electronic mail addresses or other electronic contact information on such list. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting for a period of at least ten (10) days prior to the meeting: (i) on a reasonably accessible electronic network, provided that the information required to gain access to such list is provided with the notice of the meeting, or (ii) during ordinary business hours, at the Corporation's principal executive office. In the event that the Corporation determines to make the list available on an electronic network, the Corporation may take

reasonable steps to ensure that such information is available only to stockholders of the Corporation. If the meeting is to be held at a place, then the list shall be produced and kept at the time and place of the meeting during the whole time thereof, and may be inspected by any stockholder who is present. If the meeting is to be held solely by means of remote communication, then the list shall also be open to the examination of any stockholder during the whole time of the meeting on a reasonably accessible electronic network, and the information required to access such list shall be provided with the notice of the meeting. Such list shall presumptively determine the identity of the stockholders entitled to vote at the meeting and the number of shares held by each of them.

2.16 INSPECTORS OF ELECTION.

Before any meeting of stockholders, the Board shall appoint an inspector or inspectors of election to act at the meeting or its adjournment and make a written report thereof. The number of inspectors shall be either one (1) or three (3). If any person appointed as inspector fails to appear or fails or refuses to act, then the chair of the meeting may, and upon the request of any stockholder or a stockholder's proxy shall, appoint a person to fill that vacancy.

Such inspectors shall:

- (i) ascertain the number of shares of capital stock of the corporation outstanding and the voting power of each such share;
- (ii) determine the shares of capital stock of the corporation represented at the meeting and the validity of proxies and ballots;
- (iii) count all votes and ballots;

and

(iv) determine and retain for a reasonable period a record of the disposition of any challenges made to any determination by the inspectors;

(v) certify their determination of the number of shares of capital stock of the corporation represented at the meeting and such inspectors' count of all votes and ballots.

The inspectors of election shall perform their duties impartially, in good faith, to the best of their ability and as expeditiously as is practical. If there are three (3) inspectors of election, the decision, act or certificate of a majority is effective in all respects as the decision, act or certificate of all. Any report or certificate made by the inspectors of election is prima facie evidence of the facts stated therein.

2.17 DELIVERY TO THE CORPORATION.

Whenever this Article II requires one or more persons (including a record or beneficial owner of stock) to deliver a document or information to the Corporation or any officer, employee or agent thereof (including any notice, request, questionnaire, revocation, representation or other document or agreement), such document or information shall be in writing exclusively (and not in an electronic transmission) and shall be delivered exclusively by hand (including, without limitation, overnight courier service) or by certified or registered mail, return receipt requested, and the Corporation shall not be required to accept delivery of any document not in such written form or so delivered. For the avoidance of doubt, the Corporation expressly opts out of Section 116 of the DGCL with respect to the delivery of information and documents to the Corporation required by this Article II.

ARTICLE III—DIRECTORS

3.1 POWERS.

Subject to the provisions of the DGCL and any limitations in the Certificate of Incorporation, the business and affairs of the Corporation shall be managed and all corporate powers shall be exercised by or under the direction of the Board.

3.2 NUMBER OF DIRECTORS.

Subject to the rights and preferences, if any, of any holders of any outstanding series of Preferred Stock to elect directors, the number of directors which shall constitute the whole Board of Directors shall be no less than one and not more than fifteen members and be fixed exclusively by one or more resolutions adopted from time to time by the affirmative vote of a majority of the directors then in office, even though less than a quorum of the Board of Directors.

3.3 ELECTION, QUALIFICATION, CITIZENSHIP, AND TERM OF OFFICE OF DIRECTORS.

Except as provided in Section 3.4 of these Bylaws, and subject to the Certificate of Incorporation, each director, including a director elected to fill a vacancy or newly created directorship, shall hold office until the expiration of the term for which elected and until such director's successor is elected and qualified or until such director's earlier death, resignation, disqualification or removal. Directors need not be stockholders unless so required by the Certificate of Incorporation or these Bylaws. The Certificate of Incorporation or these Bylaws may prescribe other qualifications for directors. Notwithstanding anything to the contrary in these Bylaws, at least two-thirds of the members of the Board of Directors shall be "citizens of the United States" as provided under Applicable Law (as defined in Article X below) and the Chairman of the Board shall be a "citizen of the United States" as provided under Applicable Law for so long as required by Applicable Law. If the number of Non-Citizens (as defined in Article X below) serving on the Board of Directors at any time exceeds the limitations provided under Applicable Law, one or more directors who are Non-Citizens shall, in reverse chronological order based on their tenure of service on the Board, cease to be qualified as directors and shall automatically cease to be directors.

If so provided in the Certificate of Incorporation, the directors of the Corporation shall be divided into three (3) classes.

3.4 RESIGNATION AND VACANCIES.

Any director may resign at any time upon notice given in writing or by electronic transmission to the Corporation. When one or more directors so resigns and the resignation is effective at a future date, a majority of the directors then in office, including those who have so resigned, shall have power to fill such vacancy or vacancies, the vote thereon to take effect when such resignation or resignations shall become effective, and each director so chosen shall hold office as provided in Section 3.3.

Unless otherwise provided in the Certificate of Incorporation or these Bylaws, vacancies resulting from the death, resignation, disqualification or removal of any director, and newly created directorships resulting from any increase in the authorized number of directors shall, unless the Board determines by resolution that any such vacancies or newly created directorships shall be filled by stockholders, be filled only by a majority of the directors then in office, although less than a quorum, or by a sole remaining director.

3.5 PLACE OF MEETINGS; MEETINGS BY TELEPHONE.

The Board may hold meetings, both regular and special, either within or outside the State of Delaware.

Unless otherwise restricted by the Certificate of Incorporation or these Bylaws, members of the Board, or any committee designated by the Board, may participate in a meeting of the Board, or any committee, by means of conference telephone or other communications equipment by means of which all persons participating in the meeting can hear each other, and such participation in a meeting pursuant to these Bylaws shall constitute presence in person at the meeting.

3.6 REGULAR MEETINGS.

Regular meetings of the Board may be held within or outside the State of Delaware and at such time and at such place as which has been designated by the Board and publicized among all directors, either orally or in writing, by telephone, including a voice-messaging system or other system designed to record and communicate messages, facsimile, telegraph or telex, or by electronic mail or other means of electronic transmission. No further notice shall be required for regular meetings of the Board.

3.7 SPECIAL MEETINGS; NOTICE.

Special meetings of the Board for any purpose or purposes may be called at any time by the Chair of the Board, the Chief Executive Officer, the President, the Secretary or a majority of the authorized number of directors.

Notice of the time and place of special meetings shall be:

- (i) delivered personally by hand, by courier or by telephone;
- (ii) sent by United States first-class mail, postage prepaid;
- (iii) sent by facsimile; or
- (iv) sent by electronic mail,

directed to each director at that director's address, telephone number, facsimile number or electronic mail address, as the case may be, as shown on the Corporation's records.

If the notice is (i) delivered personally by hand, by courier or by telephone, (ii) sent by facsimile or (iii) sent by electronic mail, it shall be delivered or sent at least twenty-four (24) hours before the time of the holding of the meeting. If the notice is sent by United States mail, it shall be deposited in the United States mail at least four (4) days before the time of the holding of the meeting. The notice need not specify the place of the meeting (if the meeting is to be held at the Corporation's principal executive office) nor the purpose of the meeting.

3.8 QUORUM.

At all meetings of the Board, unless otherwise provided by the Certificate of Incorporation, a majority of the authorized number of directors shall constitute a quorum for the transaction of business. The vote of a majority of the directors present at any meeting at which a quorum is present shall be the act of the Board, except as may be otherwise specifically provided by statute, the Certificate of Incorporation or these Bylaws. If a quorum is not present at any meeting of the Board, then the directors present thereat may adjourn the meeting from time to time, without notice other than announcement at the meeting, until a quorum is present.

3.9 BOARD ACTION BY WRITTEN CONSENT WITHOUT A MEETING.

Unless otherwise restricted by the Certificate of Incorporation or these Bylaws, any action required or permitted to be taken at any meeting of the Board, or of any committee thereof, may be taken without a meeting if all members of the Board or committee, as the case may be, consent thereto in writing or by electronic transmission. After an action is taken, the consent or consents relating thereto shall be filed with the minutes of the proceedings of the Board, or the committee thereof, in the same paper or electronic form as the minutes are maintained. Such action by written consent or consent by electronic transmission shall have the same force and effect as a unanimous vote of the Board.

3.10 FEES AND COMPENSATION OF DIRECTORS.

To the greatest extent permitted by applicable law, the Board shall have the authority to fix the compensation, including fees and reimbursement of expenses, of directors for services to the Corporation in any capacity.

3.11 REMOVAL OF DIRECTORS.

Except as otherwise provided by the DGCL, the Board of Directors or any individual director may be removed from office at any time (a) with or without cause by the affirmative vote of the holders of a majority of the voting power of the Voting Stock while the Corporation is under Indigo Control and (b) from and after the time the Corporation is no longer under Indigo Control, only for cause by the affirmative vote of the holders of a majority of the voting power of the Voting Stock while the Corporation is an any time (a) with or the voting power of the Voting Stock.

No reduction of the authorized number of directors shall have the effect of removing any director prior to the expiration of such director's term of office.

ARTICLE IV—COMMITTEES

4.1 COMMITTEES OF DIRECTORS .

The Board may designate one (1) or more committees, each committee to consist of one (1) or more of the directors of the Corporation. The Board may designate one (1) or more directors as alternate members of any committee, who may replace any absent or disqualified member at any meeting of the committee. In the absence or disqualification of a member of a committee, the member or members thereof present at any meeting and not disqualified from voting, whether or not such member or members constitute a quorum, may unanimously appoint another member of the Board to act at the meeting in the place of any such absent or disqualified member. Any such committee, to the extent provided in the resolution of the Board or in these Bylaws, shall have and may exercise all the powers and authority of the Board in the management of the business and affairs of the Corporation, and may authorize the seal of the Corporation to be affixed to all papers that may require it; but no such committee shall have the power or authority to (i) approve or adopt, or recommend to the stockholders, any action or matter expressly required by the DGCL to be submitted to stockholders for approval, or (ii) adopt, amend or repeal any bylaw of the Corporation. At least two-thirds

(2/3) of the members of each committee of the Board shall be comprised of individuals who meet the definition of "a citizen of the United States," as defined under Applicable Law (as defined in Article X below); <u>provided</u>, <u>however</u>, that if a committee of the Board has one (1) member, such member shall be a "a citizen of the United States," as defined immediately above.

4.2 COMMITTEE MINUTES.

Each committee shall keep regular minutes of its meetings and report the same to the Board when required.

4.3 MEETINGS AND ACTION OF COMMITTEES .

Meetings and actions of committees shall be governed by, and held and taken in accordance with, the provisions of:

(i) Section 3.5 (place of meetings and meetings by telephone);

(ii) Section 3.6 (regular meetings);

(iii) Section 3.7 (special meetings and notice);

(iv) Section 3.8 (quorum);

(v) Section 7.12 (waiver of notice); and

(vi) Section 3.9 (action without a meeting),

with such changes in the context of those Bylaws as are necessary to substitute the committee and its members for the Board and its members. However:

(i) the time of regular meetings of committees may be determined either by resolution of the Board or by resolution of the committee;

(ii) special meetings of committees may also be called by resolution of the Board; and

(iii) notice of special meetings of committees shall also be given to all alternate members, if any, who shall have the right to attend all meetings of the committee. The Board may adopt rules for the governance of any committee not inconsistent with the provisions of these Bylaws.

4.4 SUBCOMMITTEES.

Unless otherwise provided in the Certificate of Incorporation, these Bylaws or the resolutions of the Board designating the committee, a committee may create one (1) or more subcommittees, each subcommittee to consist of one (1) or more members of the committee, and delegate to a subcommittee any or all of the powers and authority of the committee. At least two-thirds (2/3) of the members of each subcommittee of the Board shall be comprised of individuals who meet the definition of "a citizen of the United States," as defined under Applicable Law (as defined in Article X below); <u>provided</u>, <u>however</u>, that if a subcommittee of the Board has one (1) member, such member shall be a "a citizen of the United States," as defined immediately above.

ARTICLE V - OFFICERS

5.1 OFFICERS

The officers of the Corporation shall be a President and a Secretary. The Corporation may also have, at the discretion of the Board, a chair of the Board, a vice chair of the Board, a chief executive officer, a chief financial officer, a treasurer, one or more vice presidents, one or more assistant treasurers, one or more assistant secretaries, and any such other officers as may be appointed in accordance with the provisions of these Bylaws. Any number of offices may be held by the same person.

5.2 APPOINTMENT OF OFFICERS

The Board shall appoint the officers of the Corporation, except such officers as may be appointed in accordance with the provisions of Section 5.3 of these Bylaws, subject to the rights, if any, of an officer under any contract of employment.

5.3 SUBORDINATE OFFICERS

The Board may appoint, or empower the Chief Executive Officer or, in the absence of a chief executive officer, the President, to appoint, such other officers and agents as the business of the Corporation may require. Each of such officers and agents shall hold office for such period, have such authority, and perform such duties as are provided in these Bylaws or as the Board may from time to time determine.

5.4 REMOVAL AND RESIGNATION OF OFFICERS

Any officer may be removed, either with or without cause, by an affirmative vote of the majority of the Board or by any officer upon whom such power of removal may be conferred by the Board.

Any officer may resign at any time by giving written notice to the Corporation. Any resignation shall take effect at the date of the receipt of that notice or at any later time specified in that notice. Unless otherwise specified in the notice of resignation, the acceptance of the resignation shall not be necessary to make it effective. Any resignation is without prejudice to the rights, if any, of the Corporation under any contract to which the officer is a party.

Any removal or resignation of an officer pursuant to this Section 5.4 shall be without prejudice to any rights of the Corporation or such officer pursuant to any contract of employment of such officer.

5.5 VACANCIES IN OFFICES

Any vacancy occurring in any office of the Corporation shall be filled by the Board or as provided in Section 5.2.

5.6 REPRESENTATION OF SHARES OF OTHER CORPORATIONS

The Chair of the Board, the President, any Vice President, the Treasurer, the Secretary or Assistant Secretary of this Corporation, or any other person authorized by the Board, is authorized to vote, represent and exercise on behalf of this Corporation all rights incident to any and all shares of any other corporation or corporations held in the name of this Corporation. The authority granted herein may be exercised either by such person directly or by any other person authorized to do so by proxy or power of attorney duly executed by such person having the authority.

5.7 AUTHORITY AND DUTIES OF OFFICERS

All officers of the Corporation shall respectively have such authority and perform such duties in the management of the business of the Corporation as may be designated from time to time by the Board or the stockholders and, to the extent not so provided, as generally pertain to their respective offices, subject to the control of the Board.

5.8 LIMITATIONS ON NON-CITIZENS AS OFFICERS

Notwithstanding anything to the contrary in these Bylaws, (a) the Chief Executive Officer, if any, and the President and at least two-thirds of the other managing officers of the Corporation shall be "citizens of the United States," as defined under Applicable Law (as defined in Article X below) for so long as such restrictions are required by Applicable Law (as defined in Article X below).

ARTICLE VI - RECORDS AND REPORTS

6.1 MAINTENANCE AND INSPECTION OF RECORDS

The Corporation shall, either at its principal executive office or at such place or places as designated by the Board, keep a record of its stockholders listing their names and addresses and the number and class of shares held by each stockholder, a copy of these Bylaws as amended to date, accounting books and other records. Notwithstanding anything to the contrary in these Bylaws, any stockholder of record shall be entitled to all rights to which such stockholder is entitled pursuant to Section 220 of the DGCL.

6.2 INSPECTION BY DIRECTORS

Any director shall have the right to examine the Corporation's stock ledger, a list of its stockholders, and its other books and records for a purpose reasonably related to his or her position as a director to the extent such director is entitled to do so under Section 220 of the DGCL.

ARTICLE VII - GENERAL MATTERS

7.1 EXECUTION OF CORPORATE CONTRACTS AND INSTRUMENTS

The Board, except as otherwise provided in these Bylaws, may authorize any officer or officers, or agent or agents, to enter into any contract or execute any instrument in the name of and on behalf of the Corporation; such authority may be general or confined to specific instances.

7.2 STOCK CERTIFICATES; PARTLY PAID SHARES

The shares of the Corporation shall be represented by certificates, provided that the Board by resolution may provide that some or all of the shares of any class or series of stock of the Corporation shall be uncertificated. Certificates for the shares of stock, if any, shall be in such form as is consistent with the Certificate of Incorporation and applicable law. Every holder of stock represented by a certificate shall be entitled to have a certificate signed by, or in the name of the Corporation by, any two officers authorized to sign stock certificates representing the number of shares registered in certificate form. The Chair or Vice Chair of the Board, Chief Executive Officer, the President, Vice President, the Treasurer, any Assistant Treasurer, the Secretary or any Assistant Secretary of the Corporation shall be specifically authorized to sign stock certificates. Any or all of the signatures on the certificate may be a facsimile. In case any officer, transfer agent or registrar who has signed or whose facsimile signature has been placed upon a certificate has ceased to be such officer, transfer agent or registrar before such certificate is issued, it may be issued by the Corporation with the same effect as if he or she were such officer, transfer agent or registrar at the date of issue.

The Corporation may issue the whole or any part of its shares as partly paid and subject to call for the remainder of the consideration to be paid therefor. Upon the face or back of each stock certificate issued to represent any such partly paid shares, or upon the books and records of the Corporation in the case of uncertificated partly paid shares, the total amount of the consideration to be paid therefor and the amount paid thereon shall be stated. Upon the declaration of any dividend on fully paid shares, the Corporation shall declare a dividend upon partly paid shares of the same class, but only upon the basis of the percentage of the consideration actually paid thereon.

7.3 SPECIAL DESIGNATION ON CERTIFICATES

If the Corporation is authorized to issue more than one class of stock or more than one series of any class, then the powers, the designations, the preferences and the relative, participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights shall be set forth in full or summarized on the face or on the back of the certificate that the Corporation shall issue to represent such class or series of stock (or, in the case of uncertificated shares, set forth in a notice provided pursuant to Section 151 of the DGCL); provided, however, that, except as otherwise provided in Section 202 of the DGCL, in lieu of the foregoing requirements, there may be set forth on the face or back of the certificate that the Corporation shall issue to represent such class or series of any uncertificated shares, included in the aforementioned notice) a statement that the Corporation will furnish without charge to each stockholder who so requests the powers, the designations, the preferences and the relative, participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations, limitations or restrictions of such preferences and/or rights.

7.4 LOST CERTIFICATES

Except as provided in this Section 7.4, no new certificates for shares shall be issued to replace a previously issued certificate unless the latter is surrendered to the Corporation and cancelled at the same time. The Corporation may issue a new certificate of stock or uncertificated shares in the place of any certificate theretofore issued by it, alleged to have been lost, stolen or destroyed, and the Corporation may require the owner of the lost, stolen or destroyed certificate, or such owner's legal representative, to give the Corporation a bond sufficient to indemnify it against any claim that may be made against it on account of the alleged loss, theft or destruction of any such certificate or the issuance of such new certificate or uncertificated shares.

7.5 CONSTRUCTION; DEFINITIONS

Unless the context requires otherwise, the general provisions, rules of construction and definitions in the DGCL shall govern the construction of these Bylaws. Without limiting the generality of this provision, the singular number includes the plural, the plural number includes the singular, and the term "person" includes both a corporation and a natural person.

7.6 DIVIDENDS

The Board, subject to any restrictions contained in either (i) the DGCL or (ii) the Certificate of Incorporation, may declare and pay dividends upon the shares of its capital stock. Dividends may be paid in cash, in property or in shares of the Corporation's capital stock. The Board may set apart out of any of the funds of the Corporation available for dividends a reserve or reserves for any proper purpose and may abolish any such reserve. Such purposes shall include but not be limited to equalizing dividends, repairing or maintaining any property of the Corporation, and meeting contingencies.

7.7 FISCAL YEAR

The fiscal year of the Corporation shall be fixed by resolution of the Board and may be changed by the Board.

7.8 SEAL

The Corporation may adopt a corporate seal, which shall be adopted and which may be altered by the Board. The Corporation may use the corporate seal by causing it or a facsimile thereof to be impressed or affixed or in any other manner reproduced.

7.9 TRANSFER OF STOCK

Shares of the Corporation shall be transferable in the manner prescribed by law and in these Bylaws. Shares of stock of the Corporation shall be transferred on the books of the Corporation only by the holder of record thereof or by such holder's attorney duly authorized in writing, upon surrender to the Corporation of the certificate or certificates representing such shares endorsed by the appropriate person or persons (or by delivery of duly executed instructions with respect to uncertificated shares), with such evidence of the authenticity of such endorsement or execution, transfer, authorization and other matters as the Corporation may reasonably require, and accompanied by all necessary stock transfer stamps. No transfer of stock shall be valid as against the Corporation for any purpose until it shall have been entered in the stock records of the Corporation by an entry showing the names of the persons from and to whom it was transferred. Notwithstanding the foregoing, the transfer of shares shall be restricted in accordance with Article X of these Bylaws.

7.10 STOCK TRANSFER AGREEMENTS

The Corporation shall have power to enter into and perform any agreement with any number of stockholders of any one or more classes of stock of the Corporation to restrict the transfer of shares of stock of the Corporation of any one or more classes owned by such stockholders in any manner not prohibited by the DGCL.

7.11 REGISTERED STOCKHOLDERS

The Corporation:

(i) shall be entitled to recognize the exclusive right of a person registered on its books as the owner of shares to receive dividends and to vote as such owner; and

(ii) shall not be bound to recognize any equitable or other claim to or interest in such share or shares on the part of another person, whether or not it shall have express or other notice thereof, except as otherwise provided by the laws of Delaware.

7.12 WAIVER OF NOTICE

Whenever notice is required to be given under any provision of the DGCL, the Certificate of Incorporation or these Bylaws, a written waiver, signed by the person entitled to notice, or a waiver by electronic transmission by the person entitled to notice, whether before or after the time of the event for which notice is to be given, shall be deemed equivalent to notice. Attendance of a person at a meeting shall constitute a waiver of notice of such meeting, except when the person attends a meeting for the express purpose of objecting at the beginning of the meeting, to the transaction of any business because the meeting is not lawfully called or convened. Neither the business to be transacted at, nor the purpose of, any regular or special meeting of the stockholders need be specified in any written waiver of notice or any waiver by electronic transmission unless so required by the Certificate of Incorporation or these Bylaws.

ARTICLE VIII - NOTICE BY ELECTRONIC TRANSMISSION

Without limiting the manner by which notice otherwise may be given effectively to stockholders, any notice to stockholders given by the Corporation under any provisions of the DGCL, the Certificate of Incorporation, or these Bylaws may be given in writing directed to the stockholder's mailing address (or by electronic transmission directed to the stockholder's electronic mail address, as applicable) as it appears on the records of the Corporation and shall be given (1) if mailed, when the notice is deposited in the U.S. mail, postage prepaid, (2) if delivered by courier service, the earlier of when the notice is received or left at such stockholder's address or (3) if given by electronic mail, when directed to such stockholder's electronic mail address unless the stockholder has notified the Corporation in writing or by electronic transmission of an objection to receiving notice by electronic mail. A notice by electronic mail must include a prominent legend that the communication is an important notice regarding the Corporation.

Without limiting the manner by which notice otherwise may be given effectively to stockholders, any notice to stockholders given by the Corporation under any provision of the DGCL, the Certificate of Incorporation or these Bylaws shall be effective if given by a form of electronic transmission consented to by the stockholder to whom the notice is given. Any such consent shall be revocable by the stockholder by written notice or electronic transmission to the Corporation. Notwithstanding the provisions of this paragraph, the Corporation may give a notice by electronic mail in accordance with the first paragraph of this section without obtaining the consent required by this paragraph.

Any notice given pursuant to the preceding paragraph shall be deemed given:

- (i) if by facsimile telecommunication, when directed to a number at which the stockholder has consented to receive notice;
- (ii) if by a posting on an electronic network together with separate notice to the stockholder of such specific posting, upon the later of (A) such posting and (B) the giving of such separate notice; and

(iii) if by any other form of electronic transmission, when directed to the stockholder.

Notwithstanding the foregoing, a notice may not be given by an electronic transmission from and after the time that (1) the Corporation is unable to deliver by such electronic transmission two (2) consecutive notices given by the Corporation and (2) such inability becomes known to the Secretary or an Assistant Secretary of the Corporation or to the transfer agent, or other person responsible for the giving of notice, provided, however, the inadvertent failure to discover such inability shall not invalidate any meeting or other action.

An affidavit of the Secretary or an Assistant Secretary or of the transfer agent or other agent of the Corporation that the notice has been given shall, in the absence of fraud, be prima facie evidence of the facts stated therein.

ARTICLE IX - INDEMNIFICATION

9.1 INDEMNIFICATION OF DIRECTORS AND OFFICERS.

The Corporation shall indemnify and hold harmless, to the fullest extent permitted by the DGCL as it presently exists or may hereafter be amended, any director or officer of the Corporation who was or is made or is threatened to be made a party or is otherwise involved in any action, suit or proceeding, whether civil, criminal, administrative or investigative (a "<u>Proceeding</u>") by reason of the fact that he or she, or a person for whom he or she is the legal representative, is or was a director or officer of the Corporation or, while serving as a director or officer of the Corporation, is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership (a "covered person"), joint venture, trust, enterprise or non-profit entity, including service with respect to employee benefit plans, against all liability and loss suffered and expenses (including attorneys' fees, judgments, fines ERISA excise taxes or penalties and amounts paid in settlement) reasonably incurred by such person in connection with any such Proceeding. Notwithstanding the foregoing, except as otherwise provided in Section 9.4, the Corporation shall be required to indemnify a person in connection with a Proceeding initiated by such person only if the Proceeding was authorized in the specific case by the Board.

9.2 INDEMNIFICATION OF OTHERS.

The Corporation shall have the power to indemnify and hold harmless, to the fullest extent permitted by applicable law as it presently exists or may hereafter be amended, any employee or agent of the Corporation who was or is made or is threatened to be made a party or is otherwise involved in any Proceeding by reason of the fact that he or she, or a person for whom he or she is the legal representative, is or was an employee or agent of the Corporation or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust, enterprise or non-profit entity, including service with respect to employee benefit plans, against all liability and loss suffered and expenses reasonably incurred by such person in connection with any such Proceeding.

9.3 PREPAYMENT OF EXPENSES.

The Corporation shall to the fullest extent not prohibited by applicable law pay the reasonable expenses (including attorneys' fees) incurred by any covered person, and may pay the expenses incurred by any employee or agent of the Corporation, in defending any Proceeding in advance of its final disposition; *provided, however*, that such payment of expenses in advance of the final disposition of the Proceeding shall be made only upon receipt of an undertaking by the person to repay all amounts advanced if it should be ultimately determined that the person is not entitled to be indemnified under this Article IX or otherwise.

9.4 NON-EXCLUSIVITY OF RIGHTS.

The rights conferred on any person by this Article IX shall not be exclusive of any other rights which such person may have or hereafter acquire under any statute, provision of the Certificate of Incorporation, these Bylaws, agreement, vote of stockholders or disinterested directors or otherwise.

9.5 INSURANCE.

The Corporation may purchase and maintain insurance on behalf of any person who is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust enterprise or non-profit entity against any liability asserted against him or her and incurred by him or her in any such capacity, or arising out of his or her status as such, whether or not the Corporation would have the power to indemnify him or her against such liability under the provisions of the DGCL.

9.6 OTHER INDEMNIFICATION.

The Corporation's obligation, if any, to indemnify or advance expenses to any person who was or is serving at its request as a director, officer, employee or agent of another corporation, partnership, joint venture, trust, enterprise or non-profit entity shall be reduced by any amount such person may collect as indemnification or advancement of expenses from such other corporation, partnership, joint venture, trust, enterprise or non-profit enterprise.

9.7 CONTINUATION OF INDEMNIFICATION.

The Corporation's obligation, if any, to indemnify or advance expenses to any person who was or is serving at its request as a director, officer, employee or agent of another corporation, partnership, joint venture, trust, enterprise or non-profit entity shall be reduced by any amount such person may collect as indemnification or advancement of expenses from such other corporation, partnership, joint venture, trust, enterprise or non-profit entity shall be reduced by any amount such person may collect as indemnification or advancement of expenses from such other corporation, partnership, joint venture, trust, enterprise or non-profit enterprise.

9.8 AMENDMENT OR REPEAL; INTERPRETATION.

The provisions of this Article IX shall constitute a contract between the Corporation, on the one hand, and, on the other hand, each individual who serves or has served as a director or officer of the Corporation (whether before or after the adoption of these Bylaws), in consideration of such person's performance of such services, and pursuant to this Article IX the Corporation intends to be legally bound to each such current or former director or officer of the Corporation. With respect to current and former directors and officers of the Corporation, the rights conferred under this Article IX are present contractual rights and such rights are fully vested, and shall be deemed to have vested fully, immediately upon adoption of theses bylaws. With respect to any directors or officers of the Corporation who commence service following adoption of these Bylaws, the rights conferred under this provision shall be present contractual rights and such rights shall fully vest, and be deemed to have vested fully, immediately upon such director or officer commencing service as a director or officer of the Corporation. Any repeal or modification of the foregoing provisions of this Article IX shall not adversely affect any right or protection (i) hereunder of any person in respect of any act or omission occurring prior to the time of such repeal or modification or (ii) under any agreement providing for indemnification or advancement of expenses to an officer or director of the Corporation in effect prior to the time of such repeal or modification.

Any reference to an officer of the Corporation in this Article IX shall be deemed to refer exclusively to the Chief Executive Officer, President, and Secretary, or other officer of the Corporation appointed by (x) the Board pursuant to Article V of these Bylaws or (y) an officer to whom the Board has delegated the power to appoint officers pursuant to Article V of these Bylaws, and any reference to an officer of any other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise shall be deemed to refer exclusively to an officer appointed by the board of directors (or equivalent governing body) of such other entity pursuant to the certificate of incorporation and bylaws (or equivalent organizational documents) of such other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise. The fact that any person who is or was an employee of the Corporation or an employee of any other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise has been given or has used the title of "Vice President" or any other title that could be construed to suggest or imply that such person is or may be an officer of the Corporation or of such other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise shall not result in such person being constituted as, or being deemed to be, an officer of the Corporation or of such other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise shall not result in such person being constituted as, or being deemed to be, an officer of the Corporation or of such other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise shall not result in such person being constituted as, or being deemed to be, an officer of the Corporation or of such other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise for purposes of this Article IX.

ARTICLE X - LIMITATIONS OF OWNERSHIP BY NON-CITIZENS

10.1 EQUITY SECURITIES.

All capital stock of the Corporation (collectively, "Equity Securities") shall be subject to the limitations set forth in this Article X.

10.2 NON-U.S. CITIZEN VOTING AND OWNERSHIP LIMITATIONS.

In no event shall persons or entities, including any agent, trustee or representative of such persons or entities, who fail to qualify as a "citizen of the United States" ("<u>Non-Citizens</u>"), as defined in Section 40102(a)(15) of Subtitle VII of Title 49 of the United States Code, as the same may be from time to time amended (the "<u>Aviation Act</u>"), in any similar legislation of the United States enacted in substitution or replacement thereof, and as interpreted by the Department of Transportation, its predecessors and successors, from time to time (the "<u>Applicable Law</u>"), own (beneficially or of record) or control Equity Securities representing in excess of (i) the percentage provided for under Applicable Law of the aggregate votes of all outstanding Equity Securities of the Corporation (the "<u>Voting Limitation</u>") and (ii) the percentage provided for under Applicable Law of all outstanding Equity Securities of the Corporation (the "<u>Outstanding Limitation</u>"). As of the Effective Time (as defined below) and for informational purposes only, under Applicable Law, the Voting Limitation is 25.0%, and the Outstanding Limitation is 49.0%. If Non-Citizens nonetheless at any time own and/or control more than the Voting Limitation, the voting rights of the Equity Securities in excess of the Voting Limitation shall be automatically suspended in accordance with Section 10.3 below. Further, if at any time a transfer or issuance of Equity Securities to a Non-Citizen would result in Non-Citizens owning more than the Outstanding Limitation, such transfer or issuance shall be null and void *ab initio*, and of no force or effect, in accordance with Section 10.3 below.

10.3 FOREIGN STOCK RECORD.

(i) The Corporation or any transfer agent shall maintain a separate stock record, designated the "Foreign Stock Record," for the registration of Equity Securities held by Non-Citizens. It is the duty of each stockholder who is a Non-Citizen to register his, her or its Equity Securities on the Foreign Stock Record. The beneficial ownership or control of Equity Securities by Non-Citizens shall be determined in conformity with regulations prescribed by the Board of Directors. Only Equity Securities that have been issued and are outstanding may be registered in the Foreign Stock Record. The Foreign Stock Record shall include (a) the name and nationality of each Non-Citizen owning (beneficially or of record) or controlling Equity Securities, (b) the number of Equity Securities owned (beneficially or of record) or controlled by each such Non-Citizen and (c) the date of registration of such Equity Securities in the Foreign Stock Record.

(ii) At no time shall Equity Securities held by Non-Citizens be voted, unless such shares are registered on the Foreign Stock Record. In no event shall Equity Securities owned (beneficially or of record) or controlled by Non-Citizens representing more than the Voting Limitation be voted. In the event that Non-Citizens shall own (beneficially or of record) or have voting control over any Equity Securities, the voting rights of such persons shall be subject to automatic suspension to the extent required to ensure that the Corporation is in compliance with Applicable Law. Voting rights of Equity Securities owned (beneficially or of record) or controlled by Non-Citizens shall be suspended in reverse chronological order based upon the date of registration in the Foreign Stock Record.

(iii) In the event that any transfer or issuance of Equity Securities to a Non-Citizen would result in Non-Citizens owning (beneficially or of record) or controlling more than the Outstanding Limitation, such transfer or issuance shall be null and void *ab initio*, and of no force or effect, and shall not be recorded in the Foreign Stock Record or the stock records of the Corporation. In the event that the Corporation shall determine that the Equity Securities registered on the Foreign Stock Record or otherwise registered on the stock records of the Corporation and owned (beneficially or of record) or controlled by Non-Citizens, taken together (without duplication), exceed the Outstanding Limitation, such number of shares shall be removed from the Foreign Stock Record and the stock records of the Corporation, as applicable, in reverse chronological order based on the date of registration (or attempted registration) in the Foreign Stock Record and the stock records of the Corporation, as applicable, and any transfer or issuance that resulted in such event shall be deemed null and void *ab initio*, and of no force or effect, such that the Foreign Stock Record and the stock records of the Corporation, as applicable, reflect the ownership of shares without giving effect to any transfer or issuance that caused the Corporation to exceed the Outstanding Limitation until the aggregate number of shares registered in the Foreign Stock Record or otherwise registered to Non-Citizens is equal to the Outstanding Limitation.

10.4 REGISTRATION OF SHARES.

Registry of the ownership of Equity Securities by Non-Citizens shall be effected by written notice to, and in the form specified from time to time by, the Secretary of the Corporation. Subject to any limitations or exceptions set forth in this Article X, the order in which such shares shall be registered on the Foreign Stock Record shall be chronological, based on the date the Corporation received notice to so register such shares; provided, that any Non-Citizen who purchases or otherwise acquires shares that are registered on the Foreign Stock Record and who registers such shares in its own name within 30 days of such acquisition will assume the position of the seller of such shares in the chronological order of shares registered on the Foreign Stock Record.

10.5 CERTIFICATION OF SHARES.

(i) The Corporation may by notice in writing (which may be included in the form of proxy or ballot distributed to stockholders in connection with the annual meeting or any special meeting of the stockholders of the Corporation, or otherwise) require a person that is a holder of record of shares or that the Corporation knows to have, or has reasonable cause to believe has beneficial ownership or control of shares to certify in such manner as the Corporation shall deem appropriate (including by way of execution of any form of proxy or ballot of such person) that, to the knowledge of such person:

(a) all shares as to which such person has record ownership or beneficial ownership are owned and controlled only by "citizens of the United States" (as defined above); or

(b) the number of shares of record or beneficially owned by such person that are owned and/or controlled by Non-Citizens is as set forth in such certificate.

(ii) With respect to any shares identified in response to clause (i)(b) above, the Corporation may require such person to provide such further information as the Corporation may reasonably require in order to implement the provisions of this Article X.

(iii) For purposes of applying the provisions of this Article X with respect to any shares, in the event of the failure of any person to provide the certificate or other information to which the Corporation is entitled pursuant to this Section 10.5, the Corporation may presume that the shares in question are owned and/or controlled by Non-Citizens.

10.6 LEGEND.

Each certificate or other representative document for capital stock of the Corporation (including each such certificate or representative document for such capital stock issued upon any permitted transfer of capital stock) shall contain a legend in substantially the following form:

THE SECURITIES OF FRONTIER GROUP HOLDINGS, INC. REPRESENTED BY THIS CERTIFICATE OR DOCUMENT ARE SUBJECT TO OWNERSHIP, CONTROL, AND VOTING RESTRICTIONS WITH RESPECT TO CERTAIN SECURITIES HELD BY PERSONS OR ENTITIES THAT FAIL TO QUALIFY AS "CITIZENS OF THE UNITED STATES" AS THE TERM IS DEFINED IN SECTION 40102(a)(15) OF SUBTITLE VII OF TITLE 49 OF THE UNITED STATES CODE, AS AMENDED, IN ANY SIMILAR LEGISLATION OF THE UNITED STATES ENACTED IN SUBSTITUTION OR REPLACEMENT THEREFOR, AND AS INTERPRETED BY THE UNITED STATES DEPARTMENT OF TRANSPORTATION, ITS PREDECESSORS AND SUCCESSORS, FROM TIME TO TIME. SUCH OWNERSHIP AND VOTING RESTRICTIONS ARE CONTAINED IN THE AMENDED AND RESTATED CERTIFICATE OF INCORPORATION AND THE BYLAWS OF FRONTIER GROUP HOLDINGS, INC., AS THE SAME MAY BE AMENDED OR RESTATED FROM TIME TO TIME. A COMPLETE AND CORRECT COPY OF SUCH AMENDED AND RESTATED CERTIFICATE OF INCORPORATION AND THE BYLAWS SHALL BE FURNISHED FREE OF CHARGE TO THE HOLDER OF THE SECURITIES REPRESENTED HEREBY UPON WRITTEN REQUEST TO THE SECRETARY OF FRONTIER GROUP HOLDINGS, INC.

ARTICLE XI - AMENDMENTS; WAIVERS

Subject to the limitations set forth in Section 9.9 of these Bylaws or the provisions of the Certificate of Incorporation, the Board is expressly empowered to adopt, amend or repeal the Bylaws of the Corporation or waive the observance of any bylaw (either generally or in a particular instance, and either retroactively or prospectively). Any adoption, amendment or repeal of the Bylaws of the Corporation or waiver of the observance of any bylaw by the Board shall require the approval of a majority of the authorized number of directors. The stockholders also shall have power to adopt, amend or repeal the Bylaws of the Corporation; <u>provided</u>, <u>however</u>, that, in addition to any vote of the holders of any class or series of stock of the Corporation required by law or by the Corporation's Certificate of Incorporation, such action by stockholders shall require the affirmative vote of (a) a majority of the voting power of the Voting Stock while the Corporation is under Indigo Control and (b) from and after the time that the Corporation no longer under Indigo Control, at least sixty-six and two-thirds percent (662/3%) of the voting power of the Voting Stock. In addition, any notice period required by the Certificate of Incorporation or these Bylaws may, to the extent permitted by law, be waived with the approval of a majority of the authorized number of directors.

ARTICLE XII - FORUM SELECTION

Unless the Corporation consents in writing to the selection of an alternative forum, the Court of Chancery of the State of Delaware (or, if such court does not have subject matter jurisdiction thereof, the federal district court of the State of Delaware) shall, to the fullest extent permitted by law, be the sole and exclusive forum for: (i) any derivative action or proceeding brought on behalf of the Corporation, (ii) any action asserting a claim for or based on a breach of a fiduciary duty owed by any current or former director, officer, other employee, agent or stockholder of the Corporation to the Corporation or the Corporation or any current or former director, officer, employee, agent or stockholder of the Corporation arising pursuant to any provision of the DGCL or the Certificate of Incorporation or bylaws or as to which the DGCL confers jurisdiction on the Court of Chancery of the State of Delaware, or (iv) any action asserting a claim related to or involving the Corporation that is governed by the internal affairs doctrine.

Unless the Corporation consents in writing to the selection of an alternative forum, the federal district courts of the United States of America shall, to the fullest extent permitted by law, be the sole and exclusive forum for the resolution of any complaint asserting a cause or causes of action arising under the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder, including all causes of action asserted against any defendant to such complaint. For the avoidance of doubt, this provision is intended to benefit and may be enforced by the Corporation, its officers and directors, the underwriters to any offering giving rise to such complaint, and any other professional or entity whose profession gives authority to a statement made by that person or entity and who has prepared or certified any part of the documents underlying the offering.

Any person or entity purchasing or otherwise acquiring or holding any interest in shares of capital stock of the Corporation shall be deemed to have notice of and consented to the provisions of this Article XI.

Failure to enforce the foregoing provisions would cause the Corporation irreparable harm, and the Corporation shall be entitled to equitable relief, including injunctive relief and specific performance, to enforce the foregoing provisions.

This exclusive forum provision will not apply to suits brought to enforce any liability or duty created by the Exchange Act of 1934, as amended, or any other claim for which the federal courts have exclusive jurisdiction.

* * * *

FRONTIER GROUP HOLDINGS, INC.

CERTIFICATE OF AMENDMENT AND RESTATEMENT OF BYLAWS

The undersigned hereby certifies that he or she is the duly elected, qualified, and acting Secretary of Frontier Group Holdings, Inc., a Delaware corporation, and that the foregoing Bylaws, comprising 29 pages, were amended and restated on , 2021 by the Corporation's board of directors.

IN WITNESS WHEREOF, the undersigned has hereunto set his or her hand this day of , 2021.

Howard M. Diamond Secretary

REGISTRATION RIGHTS AGREEMENT

BY AND BETWEEN

FRONTIER GROUP HOLDINGS, INC.

AND

INDIGO FRONTIER HOLDINGS COMPANY, LLC

_____, 2021

TABLE OF CONTENTS

SECTION I.	DEMAND REGISTRATIONS
1.1	Requests for Registration
1.2	Demand Notice
1.3	Demand Registration Expenses
1.4	Short-Form Registrations
1.5	Priority on Demand Registrations
1.6	Selection of Underwriters
1.7	Other Registration Rights
SECTION II.	PIGGYBACK REGISTRATIONS.
2.1	Right to Piggyback
2.2	Piggyback Expenses
2.3	Priority on Primary Registrations
2.4	Priority on Secondary Registrations
SECTION III.	REGISTRATION AND COORDINATION GENERALLY.
3.1	Registration Procedures
3.2	Registration Expenses
3.3	Participation in Underwritten Offerings
3.4	Company Holdback
3.5	Current Public Information
3.6	Shelf Take-Downs
SECTION IV.	INDEMNIFICATION
4.1	Indemnification by the Company
4.2	Indemnification by the Stockholders
4.3	Procedure
4.4	Entry of Judgment; Settlement
4.5	Contribution
4.6	Other Rights
SECTION V.	TERMINATION
SECTION VI.	DEFINITIONS
SECTION VII.	MISCELLANEOUS
7.1	No Inconsistent Agreements
7.2	Adjustments Affecting Registrable Securities
7.3	Amendment and Waiver
7.4	Successors and Assigns; Transferees
7.5	Severability
7.6	Counterparts
7.7	Descriptive Headings
7.8	Notices
7.9	Governing Law
7.10	Exercise of Rights and Remedies

REGISTRATION RIGHTS AGREEMENT

This Registration Rights Agreement (this "<u>Agreement</u>") is made as of _______, 2021, by and among Frontier Group Holdings, Inc., a Delaware corporation (the "<u>Company</u>"), Indigo Frontier Holdings Company, LLC, a Delaware limited liability company (the "<u>Sponsor</u>"), and such other persons, if any, that from time to time become parties hereto pursuant to the terms hereof or who join this Agreement pursuant to a Joinder Agreement substantially in the form of <u>Exhibit A</u> (together, with the Sponsor, the "<u>Stockholders</u>"). This agreement shall become effective immediately prior to the consummation of the Initial Public Offering (such date, the "<u>Effective Date</u>"). Unless otherwise noted herein, capitalized terms used herein shall have the meanings set forth in <u>Section 5</u>.

RECITALS

A. Pursuant to that certain Subscription Agreement, dated as of December 3, 2013, by and between the Company, formerly known as Falcon Acquisition Group, Inc., a Delaware corporation, and the Sponsor, the Sponsor previously acquired 5,200,000 shares of Common Stock, par value \$0.001 per share (as adjusted for any stock splits, stock dividends, reclassifications, reorganizations or similar transactions since December 3, 2013, the "<u>Shares</u>"), of the Company (the "<u>Subscription Agreement</u>").

B. As partial consideration for the investment by the Sponsor, pursuant to Section 5.1 of the Subscription Agreement, the Company agreed upon the request of the Sponsor to enter into a registration rights agreement with the Sponsor containing certain specified terms and conditions set forth therein and other terms and conditions customary for such agreements.

C. The parties hereto desire for the Company to provide the registration rights set out in this Agreement to the Sponsor in full and complete satisfaction of the Company's obligations under Section 5.1 of the Subscription Agreement.

AGREEMENT

NOW, THEREFORE, the parties to this Agreement hereby agree as follows:

SECTION I. DEMAND REGISTRATIONS

1.1 <u>Requests for Registration</u>. Subject to the other provisions of this <u>Section 1</u>, the Sponsor may (on behalf of itself and any Affiliate), at any time, initiate up to eight (8) registrations of all or part of its Registrable Securities on Form S-1 or any similar or successor long-form registration ("<u>Long-Form Registrations</u>") and, if available, an unlimited number of registrations of all or part of their Registrable Securities on Form S-3 or any similar or successor short-form registration ("<u>Short-Form Registrations</u>" and, collectively with any Long-Form Registration, the "<u>Demand Registrations</u>").

1.2 <u>Demand Notice</u>. All requests for Demand Registrations shall be made by giving written notice to the Company (a "<u>Demand Notice</u>"). Each Demand Notice shall specify the approximate number of Registrable Securities requested to be registered. Within ten (10) days after receipt of any such Demand Notice, the Company will give written notice of such requested registration to all other Stockholders, if any, and, subject to <u>Section 1.5</u>, the Company will include in any such registration (and in all related registrations and qualifications under blue sky laws or in compliance with other registration requirements and in any related underwriting), all Registrable Securities with respect to which the Company has received written requests from such other Stockholders, if any, for inclusion therein within fifteen (15) days after the delivery of the Company's notice.

1.3 <u>Demand Registration Expenses</u>. The Company will pay all Registration Expenses (as defined in <u>Section 3.2</u>) in connection with any registration initiated as a Demand Registration, whether or not it has become effective.

1.4 <u>Short-Form Registrations</u>. Demand Registrations will be Short-Form Registrations whenever the Company is permitted to use any applicable short-form (unless the managing underwriter(s) of such offering requests the Company to use a Long-Form Registration in order to sell all of the Registrable Securities requested to be sold). After the Company has become subject to the reporting requirements of the Exchange Act, the Company will use its reasonable best efforts to make Short-Form Registrations available for the sale of Registrable Securities. To the extent the Company is a well-known seasoned issuer (as defined in Rule 405 under the Securities Act) (a "<u>WKSI</u>") at the time any Demand Notice for a Short Form Registration is submitted to the Company in accordance with <u>Section 1.2</u>, the Company will use its reasonable best efforts to (i) if so requested, file an automatic shelf registration statement (as defined in Rule 405 under the Securities Act) (an "<u>automatic shelf registration statement</u>") to effect such registration, and (ii) remain a WKSI (and not become an ineligible issuer (as defined in Rule 405 under the Securities Act) under the Securities Act) during the period during which such automatic shelf registration statement is required to remain effective in accordance with this Agreement.

1.5 <u>Priority on Demand Registrations</u>. The Company shall not include in any Demand Registration any securities which are not Registrable Securities, other than securities of the Company to be offered by the Company solely for its own account (the "<u>Company Offered Securities</u>"), without the prior written consent of the Sponsor. If a Demand Registration is an underwritten offering and the managing underwriter(s) advises the Company in writing that in its opinion the number of Registrable Securities and, if permitted hereunder, other securities requested to be included in such offering exceeds the number of Registrable Securities and other securities, if any, which can be sold therein without adversely affecting the marketability of the offering, then the Company shall include in such registration, prior to the inclusion of any securities that are not Registrable Securities, other than Company Offered Securities, the number of Registrable Securities requested to be included in such offering that, in the opinion of such underwriter(s), can be sold without adversely affecting the marketability of the offering, pro rata among the respective holders thereof on the basis of the number of Registrable Securities owned by each such holder, and only after all such Registrable Securities are included in such offering, shall securities that are not Company Offered Securities or Registrable Securities, if any, be included in such offering.

1.6 <u>Selection of Underwriters</u>. The Sponsor shall have the right to select the underwriter or underwriters to administer the offering for any Demand Registration.

1.7 <u>Other Registration Rights</u>. The Company represents and warrants that it is not a party to, or otherwise subject to, any other agreement granting registration rights to any other Person with respect to any securities of the Company, other than this Agreement. Except as provided in this Agreement, for so long as any Registrable Securities are outstanding, the Company shall not grant to any Persons the right to request the Company to register any equity securities of the Company, or any securities convertible or exchangeable into or exercisable for such securities, without the approval of the Sponsor.

SECTION II. PIGGYBACK REGISTRATIONS.

2.1 Right to Piggyback.

(a) The Company will include in its Initial Public Offering all Registrable Securities held and requested by the Sponsor to be included in such registration (the "Initial Public Offering Piggyback Registration").

(b) Following the Initial Public Offering, whenever the Company proposes to register any of its equity securities under the Securities Act, including any registration pursuant to <u>Section 1.1</u> above (other in connection with registration on Form S-4 or Form S-8 or any successor or similar form) and the registration form to be used may be used for the registration of Registrable Securities (a "<u>Non-Initial Public Offering Piggyback</u> <u>Registration</u>"), the Company will give prompt written notice to all Stockholders of its intention to effect such a registration and, subject to <u>Sections 2.3</u> and <u>2.4</u> below, will include in such registration all Registrable Securities held by any Stockholders upon notice from such Stockholder within fifteen (15) days after the delivery of the Company's notice. Each such Company notice shall specify the approximate number of Company equity securities to be registered and the anticipated per share price range for such offering.

2.2 <u>Piggyback Expenses</u>. The Registration Expenses of the Stockholders will be paid by the Company in all Piggyback Registrations, whether or not any such registration becomes effective.

2.3 <u>Priority on Primary Registrations</u>. If a Piggyback Registration is an underwritten primary registration on behalf of the Company and the managing underwriter(s) advises the Company in writing (with a copy to each applicable Stockholder requesting registration of Registrable Securities) that in its opinion the number of securities requested to be included in such registration exceeds the number which can be sold in such offering without adversely affecting the marketability of such offering, the Company will include in such registration: (a) first, the securities the Company proposes to sell solely for its own account, (b) second, the Registrable Securities requested to be included in such registration, pro rata among the applicable holders of such Registrable Securities on the basis of the number of shares owned by each such holder, and (c) third, other securities requested to be included in such registration.

2.4 <u>Priority on Secondary Registrations</u>. If a Piggyback Registration is an underwritten secondary registration on behalf of holders of the Company's securities (other than the Stockholders with respect to Registrable Securities), and the managing underwriter(s) advises the Company in writing that in its opinion the number of securities requested to be included in such registration exceeds the number which can be sold in such offering without adversely affecting the marketability of the offering, the Company will include in such registration: (a) first, the Registrable Securities requested to be included in such registrable Securities on the basis of the number of Registrable Securities owned by each such holder, and (b) second, other such securities requested to be included in such registration.

SECTION III. REGISTRATION AND COORDINATION GENERALLY.

3.1 <u>Registration Procedures</u>. Whenever the Stockholders request that any Registrable Securities be registered pursuant to this Agreement, the Company will use its reasonable best efforts to effect the registration and the sale of such Registrable Securities in accordance with the intended method of disposition thereof and pursuant thereto the Company will as expeditiously as reasonably practicable:

(a) prepare and (within, in the case of a Long-Form Registration, forty-five (45) days, or, in the case of a Short-Form Registration, thirty (30) days, in each case, after the end of the period within which requests for inclusion in such registration may be given to the Company, if relevant) file with the Securities and Exchange Commission a registration statement with respect to such Registrable Securities and thereafter use its reasonable best efforts to cause such registration statement to become effective (provided that before filing a registration statement or prospectus or any amendments or supplements thereto, the Company will furnish to the counsel selected by the Sponsor for any registration in which the Sponsor participates (or counsel selected by the Stockholders holding a majority of the Registrable Securities for which registration is sought, for any registration in which the Sponsor does not participate and in which the other Stockholders do participate), copies of all such documents proposed to be filed, which documents will be subject to review by such counsel);

(b) prepare and file with the Securities and Exchange Commission such amendments and supplements to such registration statement and the prospectus used in connection therewith as may be necessary (i) to keep such registration statement effective (A) for at least ninety (90) days (subject to extension pursuant to <u>Section 3.3(b)</u>) or until each Stockholder participating in such registration has completed the distribution described in the registration statement relating to such distribution, whichever occurs first or, if such registration statement relates to an underwritten offering, such longer period as in the opinion of counsel for the underwriters a prospectus is required by law to be delivered in connection with sales of Registrable Securities by an underwriter or dealer, or (B) in the case of a Shelf Registration, until the earlier of (1) the date on which all Registrable Securities have been sold under the Shelf Registration or otherwise no longer qualify as Registrable Securities and (2) the latest date allowed by applicable law, and (ii) to comply with the provisions of the Securities Act with respect to the disposition of all securities covered by such registration statement until such time as all of such securities have been disposed of in accordance with the intended methods of disposition by the seller or sellers thereof set forth in such registration statement;

(c) furnish to each Stockholder participating in such registration such number of copies of such registration statement, each amendment and supplement thereto, the prospectus included in such registration statement (including each preliminary prospectus) and such other documents as such Stockholder may reasonably request in order to facilitate the disposition of the Registrable Securities owned by such Stockholder;

(d) use its reasonable best efforts to register or qualify such Registrable Securities under such other securities or blue sky laws of such jurisdictions as any Stockholder participating in such registration reasonably requests and do any and all other acts and things which may be reasonably necessary or advisable to enable such Stockholder to consummate the disposition in such jurisdictions of the Registrable Securities owned by such Stockholder (provided that the Company will not be required to (i) qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify but for this subsection, (ii) subject itself to taxation in respect of doing business in any such jurisdiction or (iii) consent to general service of process in any such jurisdiction);

(e) promptly notify each Stockholder participating in such registration at any time when a prospectus relating thereto is required to be delivered under the Securities Act, upon discovery that, or upon the discovery of the happening of any event as a result of which, the prospectus included in such registration statement contains an untrue statement of a material fact or omits any fact necessary to make the statements therein not misleading in the light of the circumstances under which they were made, and, at the request of any Stockholder participating in such registration, the Company will prepare and furnish to such Stockholder a reasonable number of copies of a supplement or amendment to such prospectus so that, as thereafter delivered to the prospective purchasers of such Registrable Securities, such prospectus will not contain an untrue statement of a material fact or omit to state any fact necessary to make the statements therein not misleading in the light of the circumstances under which they is a statement of the prospecture purchasers of such Registrable Securities, such prospectus will not contain an untrue statement of a material fact or omit to state any fact necessary to make the statements therein not misleading in the light of the circumstances under which they were made;

(f) cause all such Registrable Securities to be listed on each securities exchange on which similar securities issued by the Company are then listed;

(g) provide a transfer agent and registrar for all such Registrable Securities not later than the effective date of such registration statement;

(h) if at any time when the Company is required to re-evaluate its WKSI status for purposes of an automatic shelf registration statement used to effect a request for registration in accordance with <u>Section 1.4</u> (i) the Company determines that it is not a WKSI, (ii) the registration statement is required to be kept effective in accordance with this Agreement, and (iii) the registration rights of the Stockholders hereunder have not terminated, promptly amend the registration statement onto a form the Company is then eligible to use or file a new registration statement on such form, and keep such registration statement effective in accordance with the requirements otherwise applicable under this Agreement;

(i) if (i) a Shelf Registration is required to be kept effective in accordance with this Agreement after the third anniversary of the initial effective date of the Shelf Registration, (ii) the registration rights of the Stockholders hereunder have not terminated and (iii) the Company is eligible at such time to file a Shelf Registration, file a new registration statement with respect to any unsold Registrable Securities subject to the original request for registration prior to the end of the three year period after the initial effective date of the Shelf Registration, and keep such registration statement effective in accordance with the requirements otherwise applicable under this Agreement;

(j) enter into such customary agreements (including underwriting agreements in customary form) and perform the Company's obligations thereunder and take all such other actions as the Stockholders participating in such registration or the underwriters, if any, reasonably request in order to expedite or facilitate the disposition of such Registrable Securities (which might include effecting a stock split or a combination of shares);

(k) subject to reasonable confidentiality undertakings, make available for inspection by each Stockholder participating in such registration, any underwriter participating in any disposition pursuant to such registration statement and any attorney, accountant or other agent retained by such Stockholders or any such underwriter, all financial and other records, pertinent corporate documents and properties of the Company, and cause the Company's officers, directors, employees and independent accountants to supply all information reasonably requested by any such Stockholder, underwriter, attorney, accountant or agent in connection with such registration statement, and to cooperate and participate as reasonably requested by any such seller in road show presentations, in the preparation of the registration statement, each amendment and supplement thereto, the prospectus included therein, and other activities as such Stockholder may reasonably request in order to facilitate the disposition of the Registrable Securities owned by such Stockholder;

(1) otherwise use its reasonable best efforts to comply with all applicable rules and regulations of the Securities and Exchange Commission, and make available to its security holders, as soon as reasonably practicable, but not later than fifteen (15) months after the effective date of the registration statement, an earnings statement covering the period of at least twelve (12) months beginning with the first day of the Company's first full calendar quarter after the effective date of the registration statement, which earnings statement shall satisfy the provisions of Section 11(a) of the Securities Act and Rule 158 thereunder;

(m) in the event of the issuance of any stop order suspending the effectiveness of a registration statement, or of any order suspending or preventing the use of any related prospectus or suspending the qualification of any securities included in such registration statement for sale in any jurisdiction, the Company will use its reasonable best efforts promptly to obtain the withdrawal of such order;

(n) obtain one or more comfort letters, dated the effective date of such registration statement (and, if such registration includes an underwritten public offering, also dated the date of the closing under the underwriting agreement) addressed to the Persons participating in such offering, signed by the Company's independent public accountants in the then-current customary form and covering such matters of the type customarily covered from time to time by comfort letters as the Stockholders participating in such registration may reasonably request;

(o) provide one or more legal opinions of the Company's outside counsel, dated the effective date of such registration statement (and, if such registration includes an underwritten public offering, dated the date of the closing under the underwriting agreement and addressed to the underwriters), with respect to the registration statement, each amendment and supplement thereto, the prospectus included therein (including the preliminary prospectus) and such other documents relating thereto in the then-current customary form and covering such matters of the type customarily covered from time to time by legal opinions of such nature (in a form reasonably acceptable to the Stockholders participating in such registration);

(p) cooperate with the Stockholders participating in such registration and the managing underwriter(s), if any, to facilitate the timely preparation and delivery of certificates (not bearing any restrictive legends) representing securities to be sold under the registration statement, and enable such securities to be in such denominations and registered in such names as the managing underwriter(s), if any, or the Sponsor may request;

(q) notify counsel for the Stockholders participating in such registration and the managing underwriter(s), immediately, and confirm the notice in writing (i) when the registration statement, or any post-effective amendment to the registration statement, shall have become effective, or any supplement to the prospectus or any amendment prospectus shall have been filed, (ii) of the receipt of any comments from the Securities and Exchange Commission, (iii) of any request of the Securities and Exchange Commission to amend the registration statement or amend or supplement the prospectus or for additional information, and (iv) of the issuance by the Securities and Exchange Commission of any stop order suspending the effectiveness of the registration statement or of any order preventing or suspending the use of any preliminary prospectus, or of the suspension of the qualification of the registration statement for offering or sale in any jurisdiction, or of the institution or threatening of any proceedings for any of such purposes;

(r) use its reasonable best efforts to prevent the issuance of any stop order suspending the effectiveness of the registration statement or of any order preventing or suspending the use of any preliminary prospectus;

(s) if requested by the managing underwriter(s) or any Stockholder participating in such registration, promptly incorporate in a prospectus supplement or post-effective amendment such information as the managing underwriter(s) or such Stockholder reasonably requests to be included therein, including, without limitation, with respect to the number of Registrable Securities being sold by such Stockholder to such underwriter or agent, the purchase price being paid therefor by such underwriter or agent and with respect to any other terms of the underwritten offering of the Registrable Securities to be sold in such offering; and make all required filings of such prospectus supplement or post-effective amendment as soon as practicable after being notified of the matters incorporated in such prospectus supplement or post-effective amendment; and

(t) cooperate with each Stockholder participating in such registration and each underwriter or agent participating in the disposition of such Registrable Securities and their respective counsel in connection with any filings required to be made with the Financial Industry Regulatory Authority, Inc.

The Company may require each Stockholder participating in such registration to furnish the Company such information relating to the sale or registration of such securities regarding such Stockholder and the distribution of such securities as the Company hereunder may from time to time reasonably request in writing.

3.2 Registration Expenses.

(a) All expenses incident to the Company's performance of or compliance with this Agreement, including, without limitation, all registration, qualification and filing fees, fees and expenses of compliance with securities or blue sky laws, printing expenses, transfer agent's and registrar's fees, cost of distributing prospectuses in preliminary and final form as well as supplements thereto and fees and disbursements of counsel for the Company and all independent certified public accountants, underwriters (excluding discounts and commissions) and other Persons retained by the Company (all such expenses, as well as any expenses identified in Section 3.2(b), being herein called "<u>Registration Expenses</u>"), will be paid by the Company in respect of each Demand Registration and each Piggyback Registration, whether or not it has become effective, including that the Company will pay its internal expenses (including, without limitation, all salaries and expenses of its officers and employees performing legal or accounting duties), the expense of any annual audit or quarterly review, the expense of any liability insurance and the expenses and fees for listing the securities to be registered on each securities exchange on which similar securities issued by the Company are then listed; provided, that all underwriting discounts and commissions and transfer taxes, if any, attributable to the sale of equity securities of the Company shall not be included as Registration Expenses and shall be borne by the applicable seller of such equity securities of the Company.

(b) In connection with each Demand Registration and each Piggyback Registration, whether or not it has become effective, the Company will pay, and reimburse the Stockholders for the payment of, the reasonable fees and disbursements of one counsel selected by the Sponsor (or counsel selected by the Stockholders holding a majority of the Registrable Securities for which registration is sought, for any registration in which the Sponsor does not participate and in which the other Stockholders do participate), and such expenses shall be considered Registration Expenses hereunder.

3.3 Participation in Underwritten Offerings.

(a) No Stockholder may participate in any registration hereunder which is underwritten unless such Stockholder (i) agrees to sell its securities on the basis provided in any underwriting arrangements approved by the Sponsor (or the Stockholders holding a majority of the Registrable Securities for which registration is sought, for any registration in which the Sponsor does not participate and in which the other Stockholders do participate) including, without limitation, pursuant to the terms of any over-allotment or "green shoe" option requested by the managing underwriter(s), <u>provided</u> that such Stockholder will not be required to sell more than the number of Registrable Securities that such Stockholder has requested the Company to include in any registration, and (ii) completes and executes all questionnaires, powers of attorney, indemnities, underwriting agreements and other documents reasonably required under the terms of such underwriting arrangements.

(b) Each Stockholder agrees that, upon receipt of any notice from the Company of the happening of any event of the kind described in <u>Section 3.1(e)</u> above, such Stockholder will forthwith discontinue the disposition of its Registrable Securities pursuant to the registration statement until such Stockholder's receipt of the copies of a supplemented or amended prospectus as contemplated by such <u>Section 3.1(e)</u>. In the event the Company shall give any such notice, the applicable time period mentioned in <u>Section 3.1(b)</u> during which a Registration Statement is to remain effective shall be extended by the number of days during the period from and including the date of the giving of such notice pursuant to this paragraph to and including the date when such Stockholder shall have received the copies of the supplemented or amended prospectus contemplated by <u>Section 3.1(e)</u>.

3.4 <u>Company Holdback</u>. The Company shall not effect any public sale or distribution of its equity securities, or any securities convertible into or exchangeable or exercisable for such securities, with respect to any other underwritten Demand Registration or any underwritten Piggyback Registration in which Registrable Securities are included, during the seven (7) days prior to and the ninety (90)-day period beginning on the effective date of such registration, and (c) upon notice from the Sponsor that the Stockholders intend to effect an underwritten distribution of Registrable Securities pursuant to a Shelf Registration, the seven (7) days prior to and the ninety (90)-day period beginning on the date of such distribution, in each case except as part of such underwritten registration or pursuant to registrations on Form S-4 or Form S-8, and in each case unless the managing underwriter(s) otherwise agrees.

3.5 <u>Current Public Information</u>. At all times after the Company has filed a registration statement with the Securities and Exchange Commission pursuant to the requirements of either the Securities Act or the Exchange Act, the Company will use its reasonable best efforts to timely file all reports required to be filed by it under the Securities Act and the Exchange Act and the rules and regulations adopted by the Securities and Exchange Commission thereunder, and will take such further action as any holder or holders of Registrable Securities may reasonably request, all to the extent required to enable such holders to sell Registrable Securities pursuant to Rule 144.

3.6 <u>Shelf Take-Downs</u>. At any time that a Shelf Registration is effective, if a Stockholder delivers a notice to the Company (a "<u>Take-Down</u> <u>Notice</u>") stating that it intends to effect an offering of all or part of its Registrable Securities included by it on the Shelf Registration, whether such offering is underwritten or non-underwritten (a "<u>Shelf Offering</u>") and stating the number of the Registrable Securities to be included in the Shelf Offering, then, the Company shall amend or supplement the Shelf Registration as may be necessary in order to enable such Registrable Securities to be distributed pursuant to the Shelf Offering. In connection with any Shelf Offering, in the event that the managing underwriter(s), if any, advises the Company in writing that in its opinion the number of Registrable Securities to be included in such Shelf Offering exceeds the number of Registrable Securities which can be sold therein without adversely affecting the marketability of the offering, such managing underwriter(s), if any, may limit the number of shares which would otherwise be included in such Shelf Offering in the same manner as is described in <u>Section 1.5</u>.

SECTION IV. INDEMNIFICATION.

4.1 <u>Indemnification by the Company</u>. The Company agrees to indemnify and hold harmless, to the fullest extent permitted by law, each Stockholder and, as applicable, its officers, directors, trustees, employees, stockholders, holders of beneficial interests, members, and general and limited partners (collectively, such Stockholder's "<u>Indemnitees</u>") and each Person who controls such Stockholder (within the meaning of the Securities Act) against any and all losses, claims, damages, liabilities, joint or several, to which such Stockholder or any such Indemnitee may become subject under the Securities Act or otherwise, insofar as such losses, claims, damages or liabilities (or actions or proceedings, whether commenced or threatened, in respect thereof) arise out of or are based upon (a) any untrue or alleged untrue statement of material fact contained in any registration statement, prospectus or preliminary prospectus or any amendment thereof or supplement thereto, together with any documents incorporated therein by reference or, (b) any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, and the Company will reimburse such Stockholder and each of its Indemnitees for any legal or any other expenses, including any amounts paid in any settlement effected with the consent of the Company, which consent will not be unreasonably

withheld or delayed, incurred by them in connection with investigating or defending any such loss, claim, liability, action or proceeding; <u>provided</u>, <u>however</u>, that the Company shall not be liable in any such case to the extent that any such loss, claim, damage, liability (or action or proceeding in respect thereof) or expense arises out of or is based upon an untrue statement or alleged untrue statement, or omission or alleged omission, made in such registration statement, any such prospectus or preliminary prospectus or any amendment or supplement thereto, or in any application, in reliance upon, and in conformity with, written information prepared and furnished to the Company by such Stockholder expressly for use therein. In connection with an underwritten offering, the Company will indemnify such underwriters, their officers and directors and each Person who controls such underwriters (within the meaning of the Securities Act) to the same extent as provided above with respect to the indemnification of such Stockholder.

4.2 Indemnification by the Stockholders. In connection with any registration statement in which a Stockholder is participating, each such Stockholder will furnish to the Company in writing such information and affidavits as the Company reasonably requests for use in connection with any such registration statement or prospectus and, to the extent permitted by law, will indemnify and hold harmless the Company and its Indemnitees against any losses, claims, damages, liabilities, joint or several, to which the Company or any such Indemnitee may become subject under the Securities Act or otherwise, insofar as such losses, claims, damages or liabilities (or actions or proceedings, whether commenced or threatened, in respect thereof) arise out of or are based upon (a) any untrue or alleged untrue statement of material fact contained in the registration statement, prospectus or preliminary prospectus or any amendment thereof or supplement thereto or in any application, together with any documents incorporated therein by reference or (b) any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, but only to the extent that such untrue statement (or alleged untrue statement) or omission (or alleged omission) is made in such registration statement, any such prospectus or preliminary prospectus or any amendment or supplement thereto, or in any application, in reliance upon and in conformity with written information prepared and furnished to the Company by such Stockholder expressly for use therein, and such Stockholder will reimburse the Company and each such Indemnitee for any legal or any other expenses including any amounts paid in any settlement effected with the consent of such Stockholder, which consent will not be unreasonably withheld or delayed, incurred by them in connection with investigating or defending any such loss, claim, liability, action or proceeding; provided, however, that the obligation to indemnify will be individual (and not joint and several) to each Stockholder and will be limited to the net amount of proceeds received by such Stockholder from the sale of Registrable Securities pursuant to such registration statement, less any other amounts paid by such Stockholder in respect of such untrue statement, alleged untrue statement, omission or alleged omission.

4.3 <u>Procedure</u>. Any Person entitled to indemnification hereunder will (a) give prompt written notice to the indemnifying party of any claim with respect to which it seeks indemnification (<u>provided</u>, <u>however</u>, that the failure of any indemnified party to give such notice shall not relieve the indemnifying party of its obligations hereunder, except to the extent that the indemnifying party is actually prejudiced by such failure to give such notice), and (b) unless in such indemnified party's reasonable judgment a conflict of interest between such indemnified and indemnifying parties may exist with respect to such claim, permit such indemnifying party to assume the defense of such claim with counsel reasonably satisfactory to the indemnified party. If such defense is assumed, the indemnifying party will not be subject to any liability for any settlement made by the indemnified party without its consent (but such consent will not be unreasonably withheld). An indemnifying party who is not entitled to, or elects not to, assume the defense of a claim will not be obligated to pay the fees and expenses of more than one counsel for all parties indemnified by such indemnifying party with respect to such claim, unless in the reasonable judgment of any indemnified party a conflict of interest may exist between such indemnified party and any other of such claim.

4.4 <u>Entry of Judgment; Settlement</u>. The indemnifying party shall not, except with the approval of each indemnified party, consent to entry of any judgment or enter into any settlement which does not include as an unconditional term thereof the giving by the claimant or plaintiff to each indemnified party of a release from all liability in respect to such claim or litigation without any payment or consideration provided by such indemnified party.

4.5 <u>Contribution</u>. If the indemnification provided for in this <u>Section 4</u> is, other than expressly pursuant to its terms, unavailable to or is insufficient to hold harmless an indemnified party under the provisions above in respect of any losses, claims, damages or liabilities referred to therein, then each indemnifying party shall contribute to the amount paid or payable by such indemnified party as a result of such losses, claims, damages or liabilities (a) in such proportion as is appropriate to reflect the relative benefits received by the Company on the one hand and each Stockholder and any other sellers participating in the registration on the other hand from the sale of Registrable Securities pursuant to the registered offering of securities as to which indemnify is sought or (b) if the allocation provided by clause (a) above is not permitted by applicable law, in such proportion as is appropriate to reflect the relative fault of the Company on the one hand and of the Stockholders and any other sellers participating in the registration on the other hand in connection with the statement or omissions which resulted in such losses, claims, damages or liabilities, as well as any other relevant equitable considerations. The relative benefits received by the Company on the one hand and the Stockholders and any other sellers participating in the registration on the other hand shall be deemed to be in the same proportion as the total net proceeds from the offering (before deducting expenses) to the Company bear to the total net proceeds from the offering (before deducting expenses) to the Company on the one hand and of the Stockholders and any other sellers participating in the registration. The relative fault of the Company on the one hand and of the Stockholders and any other sellers participating in the registration. The relative fault of the Company on the one hand and of the Stockholders and any other sellers participating in the registration. The relative fault of the Company on the one hand and of the S

The Company and the Stockholders participating in the registration agree that it would not be just and equitable if contribution pursuant to this <u>Section 4</u> were determined by pro rata allocation (even if the Stockholders participating in the registration were treated as one entity for such purposes) or by any other method of allocation which does not take account of the equitable considerations referred to in the immediately preceding paragraph. The amount paid or payable by an indemnified party as a result of the losses, claims, damages and liabilities referred to in the immediately preceding paragraph shall be deemed to include, subject to the limitations set forth above, any legal or other expenses reasonably incurred by such indemnified party in connection with investigating or defending any such action or claim. Notwithstanding the provisions of this <u>Section 4</u>, no Stockholder participating in the registration shall be required to contribute any amount in excess of the net proceeds received by such Stockholder from the sale of Registrable Securities covered by the relevant registration statement filed pursuant hereto, less any other amounts paid by such Stockholder in respect of such untrue statement, alleged untrue statement, omission or alleged omission. No person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any person who was not guilty of such fraudulent misrepresentation.

4.6 <u>Other Rights</u>. The indemnification and contribution by any such party provided for under this Agreement shall be in addition to any other rights to indemnification or contribution which any indemnified party may have pursuant to law or contract and will remain in full force and effect regardless of any investigation made or omitted by or on behalf of the indemnified party or any officer, director or controlling Person of such indemnified party and will survive the transfer of securities.

SECTION V. TERMINATION.

The rights of the Sponsor or any other Stockholders, as the case may be, to request registration or inclusion of Registrable Securities in any registration pursuant to <u>Sections I</u> and <u>II</u> shall terminate upon the earlier to occur of:

(a) such time as Rule 144 or another similar exemption under the Securities Act is available for the sale of all of the Stockholder's shares without volume or other limitation during a three-month period without registration; and

(b) the tenth anniversary of the Initial Public Offering.

SECTION VI. DEFINITIONS.

"Affiliate" shall mean, with respect to any Person, (a) any other Person which directly or indirectly through one or more intermediaries controls, or is controlled by, or is under common control with, such specified Person; <u>provided</u>, <u>however</u>, that neither the Company nor any of its subsidiaries shall be deemed an Affiliate of the any of the Stockholders (and vice versa), and (b) if such specified Person is a private equity investment fund, any other private equity investment fund the primary investment advisor to which is the primary investment advisor to such specified Person or an Affiliate thereof.

"Agreement" has the meaning provided in the introduction.

"automatic shelf registration statement" has the meaning provided in Section 1.4.

"Board" shall mean the Board of Directors of the Company.

"Company" has the meaning provided in the introduction.

"Company Offered Securities" has the meaning provided in Section 1.5.

"Demand Notice" has the meaning provided in Section 1.2.

"Demand Registrations" shall mean Long-Form Registrations and Short-Form Registrations requested pursuant to Section 1.1.

"Discriminate" shall mean, with respect to a specified Person, to change the rights of such specified Person as compared to other applicable Persons in a manner that is, or is reasonably expected to be, materially and adversely different than the changes to the rights of the other applicable Persons.

"Effective Date" has the meaning provided in the introduction.

"Exchange Act" shall mean the Securities Exchange Act of 1934, as amended from time to time, or any successor federal law then in force.

"Indemnitees" has the meaning provided in Section 4.1.

"Initial Public Offering" shall mean the initial underwritten Public Offering registered on Form S-1 (or any successor form under the Securities Act).

"Long-Form Registrations" has the meaning provided in Section 1.1.

"Person" shall mean any individual, partnership, corporation, company, association, trust, joint venture, limited liability company, unincorporated organization, entity or division, or any government, governmental department or agency or political subdivision thereof.

"Piggyback Registration" shall mean any Initial Public Offering Registration and/or any Non-Initial Public Offering Registration.

"Public Offering" shall mean a public offering and sale of Common Stock for cash pursuant to an effective registration statement under the Securities Act.

"Registrable Securities" shall mean (a) the Shares, and (b) any common equity securities issued or issuable directly or indirectly with respect to any of the foregoing securities referred to in clause (a) by way of stock splits, stock dividends, reclassifications, reorganizations or similar transactions. As to any particular shares constituting Registrable Securities, such shares will cease to be Registrable Securities when they have been (x) effectively registered under the Securities Act and disposed of in accordance with the registration statement covering them, or (y) sold to the public pursuant to Securities Act Rule 144 or sold in a block sale to a financial institution in the ordinary course of its trading business, in each case in compliance with this Agreement. For purposes of this Agreement, a Person will be deemed to be a holder of Registrable Securities whenever such Person has the right to acquire directly or indirectly such Registrable Securities (upon conversion or exercise in connection with a transfer of securities or otherwise, but disregarding any restrictions or limitations upon the exercise of such right), whether or not such acquisition has actually been effected.

"Registration Expenses" has the meaning provided in Section 3.2.

"Rule 144" shall mean Securities and Exchange Commission Rule 144 under the Securities Act, as Rule 144 may be amended from time to time, or any similar successor rule that may be issued by the Securities and Exchange Commission.

"Securities Act" shall mean the Securities Act of 1933 and the rules promulgated thereunder, in each case as amended from time to time.

"Securities and Exchange Commission" includes any governmental body or agency succeeding to the functions thereof.

"Shares" has the meaning provided in the recitals.

"Shelf Offering" has the meaning provided in Section 3.6.

"Shelf Registration" shall mean the filing of a Short-Form Registration with the Securities and Exchange Commission in accordance with and pursuant to Rule 415 under the Securities Act (or any successor rule then in effect).

"Short-Form Registrations" has the meaning provided in Section 1.1.

"Sponsor" has the meaning provided in the introduction.

"Stock" shall mean the capital stock of the Company. For clarification purposes, as of the Effective Date, the Common Stock constitutes all of the Company's issued and outstanding capital stock.

"Stockholders" has the meaning provided in the introduction.

"Subscription Agreement" has the meaning provided in the recitals.

"Take-Down Notice" has the meaning provided in Section 3.6.

"Transfer" shall mean any sale, pledge, assignment, encumbrance or other transfer or disposition of any Registrable Securities (or any voting or economic interest therein) to any other Person, whether directly, indirectly, voluntarily, involuntarily, by operation of law, pursuant to judicial process or otherwise.

"WKSI" has the meaning provided in Section 1.4.

SECTION VII. MISCELLANEOUS.

7.1 <u>No Inconsistent Agreements</u>. The Company will not hereafter enter into any agreement with respect to its securities which is inconsistent with or violates the rights granted to the Stockholders in this Agreement.

7.2 <u>Adjustments Affecting Registrable Securities</u>. The Company will not take any action, or permit any change to occur, with respect to its securities which would materially and adversely affect the ability of the Stockholders to include its Registrable Securities in a registration undertaken pursuant to this Agreement (including, without limitation, effecting a stock split, stock dividend, reclassification, reorganization or similar transaction). If the Company or any Affiliate creates a new holding company (<u>Holdco</u>") or otherwise engages in any reorganization of any kind, the result of which is that the stockholders of the Company immediately before such event become all the stockholders of Holdco or such other successor entity, then in each instance the provisions of this Agreement will, in addition to applying to the Company, also apply to Holdco or such other successor entity in the same manner as if Holdco or such other successor entity were substituted for the Company throughout this Agreement.

7.3 <u>Amendment and Waiver</u>. This Agreement may be amended, modified, extended, terminated or waived (an "<u>Amendment</u>"), and the provisions hereof may be waived, only by an agreement in writing signed by the Company and the Sponsor. Each such Amendment shall be binding upon each party hereto. In addition, each party hereto and each Stockholder subject hereto may waive any right hereunder, as to itself, by an instrument in writing signed by such party. The failure of any party to enforce any provisions of this Agreement shall in no way be construed as a waiver of such provisions and shall not affect the right of such party thereafter to enforce each and every provision of this Agreement in accordance with its terms. To the extent the Amendment of any Section of this Agreement would require a specific consent pursuant to this <u>Section 7.3</u>, any Amendment to the definitions used in such Section as applied to such Section shall also require the same specified consent.

7.4 <u>Successors and Assigns; Transferees</u>. This Agreement shall be binding upon and inure to the benefit of and be enforceable by the parties hereto and their respective successors and assigns. Registrable Securities shall continue to be Registrable Securities after any Transfer pursuant to which the transferee is deemed to hold "restricted securities" as such term is defined in Rule 144(a)(3); provided, however, that Registrable Securities shall not continue to be Registrable Securities if such securities were effectively registered under the Securities Act and disposed of in accordance with the registration statement covering them. Any transferee receiving shares of Registrable Securities in a Transfer shall become a Stockholder party to this Agreement and subject to the terms and conditions of, and be entitled to enforce, this Agreement to the same extent, and in the same capacity, as the Person that Transfers such shares to such transferee. If the Sponsor Transfers 100% of its Registrable Securities, the term "Sponsor" hereunder shall be deemed to include all Stockholders, if any, and any exercise of rights by the Stockholders in such capacity shall be determined by Stockholders holding a majority of Registrable

Securities then outstanding. Prior to the Transfer of any Registrable Securities to any transferee, and as a condition thereto, each Stockholder effecting such Transfer shall cause such transferee to deliver to the Company a joinder agreement, in form and substance reasonably satisfactory to the Company, to be bound by the terms and conditions of this Agreement.

7.5 <u>Severability</u>. Whenever possible, each provision of this Agreement shall be interpreted in such manner as to be effective and valid under applicable law, but if any provision of this Agreement is held to be invalid, illegal or unenforceable in any respect under any applicable law or rule in any jurisdiction, such invalidity, illegality or unenforceability shall not affect any other provision or the effectiveness or validity of any provision in any other jurisdiction, and this Agreement shall be reformed, construed and enforced in such jurisdiction as if such invalid, illegal or unenforceable provision had never been contained herein.

7.6 <u>Counterparts</u>. This Agreement may be executed in separate counterparts (including by means of facsimile or electronic transmission in portable document format (pdf)), each of which shall be an original and all of which taken together shall constitute one and the same Agreement.

7.7 <u>Descriptive Headings</u>. The descriptive headings of this Agreement are inserted for convenience only and do not constitute a part of this Agreement.

7.8 <u>Notices</u>. Any notices and other communications required or permitted in this Agreement shall be effective if in writing and (a) delivered personally, (b) sent by facsimile or electronic transmission, or (c) sent by overnight courier, in each case, addressed as follows:

If to the Company, to:

Frontier Group Holdings, Inc. 4545 Airport Way Denver, CO 80239 Attn: General Counsel

If to the Sponsor, to:

Indigo Frontier Holdings, LLC C.O. Indigo Partners 2525 East Camelback Road, Suite 900 Phoenix, AZ 85016 Attn: William A. Franke Facsimile: (602) 224-1555

If to any other Stockholder, to it at the address set forth on <u>Schedule I</u>, or if not set forth thereon, in the records of the Company.

Notice to the holder of record of any shares of capital stock shall be deemed to be notice to the holder of such shares for all purposes hereof.

Unless otherwise specified herein, such notices or other communications shall be deemed effective (x) on the date received, if personally delivered, (y) on the date received if delivered by facsimile or electronic transmission on a business day, or if not delivered on a business day, on the first business day thereafter and (z) two (2) business days after being sent by overnight courier. Each of the parties hereto shall be entitled to specify a different address by giving notice as aforesaid to each of the other parties hereto.

7.9 <u>Governing Law</u>. This Agreement and all claims arising out of or based upon this Agreement or relating to the subject matter hereof shall be governed by and construed in accordance with the domestic substantive laws of the State of Delaware without giving effect to any choice or conflict of laws provision or rule that would cause the application of the domestic substantive laws of any other jurisdiction.

7.10 <u>Exercise of Rights and Remedies</u>. No delay of or omission in the exercise of any right, power or remedy accruing to any party as a result of any breach or default by any other party under this Agreement shall impair any such right, power or remedy, nor shall it be construed as a waiver of or acquiescence in any such breach or default, or of any similar breach or default occurring later; nor shall any such delay, omission nor waiver of any single breach or default be deemed a waiver of any other breach or default occurring before or after that waiver.

(SIGNATURE PAGES FOLLOW)

IN WITNESS WHEREOF, the parties have executed this Registration Rights Agreement on the day and year first above written.

FRONTIER GROUP HOLDINGS, INC.

By:

Name:Barry BiffleTitle:President and Chief Executive Officer

[Signature page to Registration Rights Agreement]

SPONSOR

INDIGO FRONTIER HOLDINGS COMPANY, LLC

By: Indigo Denver Management Company, LLC

By: Its: William A. Franke

Sole Member

<u>Schedule I</u>

Indigo Frontier Holdings Company, LLC c/o Indigo Partners 2525 East Camelback Road Suite 900 Phoenix, Arizona 85016 Attention: William A. Franke Facsimile: (602) 224-1555

Exhibit A

Joinder Agreement to Registration Rights Agreement

The undersigned hereby agrees, effective as of the date hereof, to be a party to the Registration Rights Agreement (the "<u>Agreement</u>") dated as of _______, 2021 by and among Frontier Group Holdings, Inc. and the other parties thereto and for all purposes of the Agreement, the undersigned shall be included with the term "Stockholder" (as defined in the Agreement). The address to which notices may be sent to the undersigned is as follows:

Notice Address:

[NAME OF UNDERSIGNED]

FRONTIER GROUP HOLDINGS, INC. 2021 INCENTIVE AWARD PLAN

ARTICLE I. PURPOSE

The Plan's purpose is to enhance the Company's ability to attract, retain and motivate persons who make (or are expected to make) important contributions to the Company by providing these individuals with equity ownership opportunities.

ARTICLE II. DEFINITIONS

As used in the Plan, the following words and phrases have the meanings specified below, unless the context clearly indicates otherwise:

2.1 "*Administrator*" means the Board or a Committee to the extent that the Board's powers or authority under the Plan have been delegated to such Committee. With reference to the Board's or a Committee's powers or authority under the Plan that have been delegated to one or more officers pursuant to Section 4.2, the term "Administrator" shall refer to such officer(s) unless and until such delegation has been revoked.

2.2 "*Applicable Law*" means any applicable law, including without limitation: (a) provisions of the Code, the Securities Act, the Exchange Act and any rules or regulations thereunder; (b) corporate, securities, tax or other laws, statutes, rules, requirements or regulations, whether U.S. or non-U.S. federal, state, or local; and (c) rules of any securities exchange or automated quotation system on which the Shares are listed, quoted or traded.

2.3 "*Award*" means an Option, Stock Appreciation Right, Restricted Stock award, Restricted Stock Unit award, Performance Bonus Award, Performance Stock Unit award, Dividend Equivalents award or Other Stock or Cash Based Award granted to a Participant under the Plan.

2.4 "*Award Agreement*" means an agreement evidencing an Award, which may be written or electronic, that contains such terms and conditions as the Administrator determines, consistent with and subject to the terms and conditions of the Plan.

2.5 "Board" means the Board of Directors of the Company.

2.6 "*Cause*" means, if the Participant is a party to a written employment or consulting agreement with the Company or any of its Subsidiaries or an Award Agreement in which the term "cause" is defined, "Cause" shall be as defined in such agreement, or if no such agreement exists, (i) any willful, material violation by the Participant of any law or regulation applicable to the business of the Company or a Subsidiary or other affiliate of the Company, (ii) the Participant's conviction for, or guilty plea to, a felony (or crime of similar magnitude under Applicable Law outside the United States), or any willful perpetration by the Participant of a common law fraud, (iii) the Participant's commission of an act of personal dishonesty which involves personal profit in connection with the Company or any other entity having a business relationship with the Company, (iv) any material breach or violation by the Participant of any greement or understanding between the Company or any Subsidiary or other affiliate of the Company and the Participant regarding the terms of the Participant's service as an employee, officer, director or consultant to the Company or a Subsidiary or other affiliate of the Company and the Participant regarding the terms of the Participant's service as an employee, officer, director or consultant to the Company or a Subsidiary or other affiliate of the Company, including without limitation, the willful and continued failure or refusal of the Participant to perform the material duties required of such

Participant as an employee, officer, director or consultant of the Company or a Subsidiary or other affiliate of the Company, other than as a result of having a Disability, or a breach of any applicable invention assignment and confidentiality agreement or similar agreement between the Company or a Subsidiary or other affiliate of the Company and the Participant, (v) the Participant's willful violation of the Company's code of ethics, (vi) the Participant's willful disregard of the policies of the Company or any Subsidiary or other affiliate of the Company so as to cause loss, harm, damage or injury to the property, reputation or employees of the Company or a Subsidiary or other affiliate of the Company, or (vii) any other willful misconduct by the Participant which is injurious to the financial condition or business reputation of, or is otherwise injurious to, the Company or a Subsidiary or other affiliate of the Company or a Subsidiary or other affiliate of the Company or a Subsidiary or other affiliate of the Company or a Subsidiary or other affiliate of the Company or a Subsidiary or other affiliate of the Company or a Subsidiary or other affiliate of the Company or a Subsidiary or other affiliate of the Company or a Subsidiary or other affiliate of the Company or a Subsidiary or other affiliate of the Company or a Subsidiary or other affiliate of the Company or a Subsidiary or other affiliate of the Company or a Subsidiary or other affiliate of the Company or a Subsidiary or other affiliate of the Company or a Subsidiary or other affiliate of the Company.

2.7 "Change in Control" means any of the following:

(a) A transaction or series of transactions (other than an offering of Common Stock to the general public through a registration statement filed with the Securities and Exchange Commission) whereby any "person" or related "group" of "persons" (as such terms are used in Sections 13(d) and 14(d)(2) of the Exchange Act) directly or indirectly acquires beneficial ownership (within the meaning of Rules 13d-3 and 13d-5 under the Exchange Act) of the Company's securities possessing more than 50% of the total combined voting power of the Company's securities outstanding immediately after such acquisition; provided, however, that the following acquisitions shall not constitute a Change in Control: (i) any acquisition by the Company or any of its Subsidiaries; (ii) any acquisition by an employee benefit plan maintained by the Company or any of its Subsidiaries, (iii) any acquisition by an employee benefit plan maintained by the Company or any of its Subsidiaries, (iii) any acquisition by an employee benefit plan maintained by the Participant, any acquisition by the Participant or any group of persons including the Participant (or any entity controlled by the Participant or any group of persons including the Participant (or any entity controlled by the Participant or any group of persons including the Participant);

(b) The Incumbent Directors cease for any reason to constitute a majority of the Board;

(c) The consummation by the Company (whether directly involving the Company or indirectly involving the Company through one or more intermediaries) of (x) a merger, consolidation, reorganization, or business combination, (y) a sale or other disposition of all or substantially all of the Company's assets in any single transaction or series of related transactions or (z) the acquisition of assets or stock of another entity, in each case other than a transaction:

(i) which results in the Company's voting securities outstanding immediately before the transaction continuing to represent (either by remaining outstanding or by being converted into voting securities of the Company or the person that, as a result of the transaction, controls, directly or indirectly, the Company or owns, directly or indirectly, all or substantially all of the Company's assets or otherwise succeeds to the business of the Company (the Company or such person, the "*Successor Entity*")) directly or indirectly, at least a majority of the combined voting power of the Successor Entity's outstanding voting securities immediately after the transaction;

(ii) after which no person or group beneficially owns voting securities representing 50% or more of the combined voting power of the Successor Entity; provided, however, that no person or group shall be treated for purposes of this Section 2.7(c)(ii) as beneficially owning 50% or more of the combined voting power of the Successor Entity solely as a result of the voting power held in the Company prior to the consummation of the transaction; and

(iii) after which at least a majority of the members of the board of directors (or the analogous governing body) of the Successor Entity were Board members at the time of the Board's approval of the execution of the initial agreement providing for such transaction; or (d) The completion of a liquidation or dissolution of the Company.

Notwithstanding the foregoing, if a Change in Control constitutes a payment event with respect to any Award (or any portion of an Award) that provides for the deferral of compensation that is subject to Section 409A, to the extent required to avoid the imposition of additional taxes under Section 409A, the transaction or event described in subsection (a), (b), (c) or (d) of this Section 2.7 with respect to such Award (or portion thereof) shall only constitute a Change in Control for purposes of the payment timing of such Award if such transaction also constitutes a "change in control event," as defined in Treasury Regulation Section 1.409A-3(i)(5).

The Administrator shall have full and final authority, which shall be exercised in its sole discretion, to determine conclusively whether a Change in Control has occurred pursuant to the above definition, the date of such Change in Control and any incidental matters relating thereto; provided that any exercise of authority in conjunction with a determination of whether a Change in Control is a "change in control event" as defined in Treasury Regulation Section 1.409A-3(i)(5) shall be consistent with such regulation.

2.8 "*Code*" means the U.S. Internal Revenue Code of 1986, as amended, and all regulations, guidance, compliance programs and other interpretative authority issued thereunder.

2.9 "*Committee*" means one or more committees or subcommittees of the Board, which may include one or more Company directors or executive officers, to the extent permitted by Applicable Law. To the extent required to comply with the provisions of Rule 16b-3, it is intended that each member of the Committee will be, at the time the Committee takes any action with respect to an Award that is subject to Rule 16b-3, a "non-employee director" within the meaning of Rule 16b-3; however, a Committee member's failure to qualify as a "non-employee director" within the meaning of Rule 16b-3; however, a Committee that is otherwise validly granted under the Plan.

2.10 "Common Stock" means the common stock of the Company.

2.11 "Company" means Frontier Group Holdings, Inc., a Delaware corporation, or any successor.

2.12 "*Consultant*" means any person, including any adviser, engaged by the Company or a Subsidiary to render services to such entity if the consultant or adviser: (i) renders bona fide services to the Company or a Subsidiary; (ii) renders services not in connection with the offer or sale of securities in a capital-raising transaction and does not directly or indirectly promote or maintain a market for the Company's securities; and (iii) is a natural person.

2.13 "*Designated Beneficiary*" means, if permitted by the Company, the beneficiary or beneficiaries the Participant designates, in a manner the Company determines, to receive amounts due or exercise the Participant's rights if the Participant dies. Without a Participant's effective designation, "Designated Beneficiary" will mean the Participant's estate or legal heirs.

2.14 "*Director*" means a Board member.

2.15 "Disability" means a permanent and total disability under Section 22(e)(3) of the Code.

2.16 "*Dividend Equivalents*" means a right granted to a Participant to receive the equivalent value (in cash or Shares) of dividends paid on a specified number of Shares. Such Dividend Equivalent shall be converted to cash or additional Shares, or a combination of cash and Shares, by such formula and at such time and subject to such limitations as may be determined by the Administrator.

2.17 "**DRO**" means a "domestic relations order" as defined by the Code or Title I of the U.S. Employee Retirement Income Security Act of 1974, as amended, or the rules thereunder.

2.18 "Effective Date" has the meaning set forth in Section 11.3.

2.19 "*Employee*" means any employee of the Company or any of its Subsidiaries.

2.20 "*Equity Restructuring*" means a nonreciprocal transaction between the Company and its stockholders, such as a stock dividend, stock split (including a reverse stock split), spin-off or recapitalization through a large, nonrecurring cash dividend, that affects the number or kind of Shares (or other Company securities) or the share price of Common Stock (or other Company securities) and causes a change in the per share value of the Common Stock underlying outstanding Awards.

2.21 "*Exchange Act*" means the U.S. Securities Exchange Act of 1934, as amended, and all regulations, guidance and other interpretative authority issued thereunder.

2.22 "*Fair Market Value*" means, as of any date, the value of a Share determined as follows: (i) if the Common Stock is listed on any established stock exchange, the value of a Share will be the closing sales price for a Share as quoted on such exchange for such date, or if no sale occurred on such date, the last day preceding such date during which a sale occurred, as reported in *The Wall Street Journal* or another source the Administrator deems reliable; (ii) if the Common Stock is not listed on an established stock exchange but is quoted on a national market or other quotation system, the value of a Share will be the closing sales price for a Share on such date, or if no sales occurred on such date, then on the last date preceding such date during which a sale occurred, as reported in The Wall Street Journal or another source the Administrator deems reliable; or (iii) if the Common Stock is not listed on a national market or other quotation system, the value of a Share will be the closing sales price for a Share on such date, or if no sales occurred on such date, then on the last date preceding such date during which a sale occurred, as reported in The Wall Street Journal or another source the Administrator deems reliable; or (iii) if the Common Stock is not listed on any established stock exchange or quoted on a national market or other quotation system, the value established by the Administrator in its sole discretion. Notwithstanding the foregoing, with respect to any Award granted on or after the effectiveness of the Company's registration statement relating to its initial public offering and prior to the Public Trading Date, the Fair Market Value means the initial public offering price of a Share as set forth in the Company's final prospectus relating to its initial public offering filed with the Securities and Exchange Commission.

2.23 "*Greater Than 10% Stockholder*" means an individual then owning (within the meaning of Section 424(d) of the Code) more than 10% of the total combined voting power of all classes of stock of the Company or any parent corporation or subsidiary corporation of the Company, as determined in accordance with in Section 424(e) and (f) of the Code, respectively.

2.24 "*Incentive Stock Option*" means an Option that meets the requirements to qualify as an "incentive stock option" as defined in Section 422 of the Code.

2.25 "*Incumbent Directors*" means, for any period of 12 consecutive months, individuals who, at the beginning of such period, constitute the Board together with any new Director(s) (other than a Director designated by a person who shall have entered into an agreement with the Company to effect a transaction described in Section 2.7(a) or 2.7(c)) whose election or nomination for election to the Board was approved by a vote of at least a majority (either by a specific vote or by approval of the proxy statement of the Company in which such person is named as a nominee for Director without objection to such nomination) of the Directors then still in office who either were Directors at the beginning of the 12-month period or whose election or nomination for election contest with respect to Directors or as a result of any other actual or threatened solicitation of proxies by or on behalf of any person other than the Board shall be an Incumbent Director.

2.26 "Non-Employee Director" means a Director who is not an Employee.

2.27 "Nonqualified Stock Option" means an Option that is not an Incentive Stock Option.

2.28 "*Option*" means a right granted under Article VI to purchase a specified number of Shares at a specified price per Share during a specified time period. An Option may be either an Incentive Stock Option or a Nonqualified Stock Option.

2.29 "Other Stock or Cash Based Awards" means cash awards, awards of Shares, and other awards valued wholly or partially by referring to, or are otherwise based on, Shares or other property.

2.30 "**Overall Share Limit**" means the sum of (i) [_____] Shares; (ii) any Shares that are subject to Prior Plan Awards that become available for issuance under the Plan pursuant to Article V; and (iii) an annual increase on the first day of each year beginning on January 1, 2022 and annually thereafter ending in 2031, equal to the lesser of (A) 1% of the Shares outstanding on the last day of the immediately preceding fiscal year and (B) such smaller number of Shares as determined by the Board or the Committee.

2.31 "Participant" means a Service Provider who has been granted an Award.

2.32 "Performance Bonus Award" has the meaning set forth in Section 8.3.

2.33 "**Performance Stock Unit**" means a right granted to a Participant pursuant to Section 8.1 and subject to Section 8.2, to receive cash or Shares, the payment of which is contingent upon achieving certain performance goals or other performance-based targets established by the Administrator.

2.34 "*Permitted Transferee*" means, with respect to a Participant, any "family member" of the Participant, as defined in the General Instructions to Form S-8 Registration Statement under the Securities Act (or any successor form thereto), or any other transferee specifically approved by the Administrator after taking into account Applicable Law.

2.35 "Plan" means this 2021 Incentive Award Plan.

2.36 "Prior Plan" means the Company's 2014 Equity Incentive Plan, as amended.

2.37 "Prior Plan Award" means an award outstanding under the Prior Plan as of the Effective Date.

2.38 "*Public Trading Date*" means the first date upon which Common Stock is listed (or approved for listing) upon notice of issuance on any securities exchange or designated (or approved for designation) upon notice of issuance as a national market security on an interdealer quotation system.

2.39 "Restricted Stock" means Shares awarded to a Participant under Article VII, subject to certain vesting conditions and other restrictions.

2.40 "*Restricted Stock Unit*" means an unfunded, unsecured right to receive, on the applicable settlement date, one Share or an amount in cash or other consideration determined by the Administrator to be of equal value as of such settlement date, subject to certain vesting conditions and other restrictions.

2.41 "*Rule 16b-3*" means Rule 16b-3 promulgated under the Exchange Act.

2.42 "*Section 409A*" means Section 409A of the Code and the regulations promulgated thereunder by the United States Treasury Department, as amended or as may be amended from time to time.

2.43 "Securities Act" means the U.S. Securities Act of 1933, as amended, and all regulations, guidance and other interpretative authority issued thereunder.

2.44 "Service Provider" means an Employee, Consultant or Director.

2.45 "Shares" means shares of Common Stock.

2.46 "*Stock Appreciation Right*" or "*SAR*" means a right granted under Article VI to receive a payment equal to the excess of the Fair Market Value of a specified number of Shares on the date the right is exercised over the exercise price set forth in the applicable Award Agreement.

2.47 "*Subsidiary*" means any entity (other than the Company), whether U.S. or non-U.S., in an unbroken chain of entities beginning with the Company if each of the entities other than the last entity in the unbroken chain beneficially owns, at the time of the determination, securities or interests representing at least 50% of the total combined voting power of all classes of securities or interests in one of the other entities in such chain.

2.48 "*Substitute Awards*" means Awards granted or Shares issued by the Company in assumption of, or in substitution or exchange for, awards previously granted, or the right or obligation to make future awards, in each case by a company or other entity acquired by the Company or any Subsidiary or with which the Company or any Subsidiary combines.

2.49 "*Tax-Related Items*" means any U.S. and non-U.S. federal, state and/or local taxes (including, without limitation, income tax, social insurance contributions, fringe benefit tax, employment tax, stamp tax and any employer tax liability which has been transferred to a Participant) for which a Participant is liable in connection with Awards and/or Shares.

2.50 "Termination of Service" means:

(a) As to a Consultant, the time when the engagement of a Participant as a Consultant to the Company or a Subsidiary is terminated for any reason, with or without Cause, including, without limitation, by resignation, discharge, death or retirement, but excluding terminations where the Consultant simultaneously commences or remains in employment or service with the Company or any Subsidiary.

(b) As to a Non-Employee Director, the time when a Participant who is a Non-Employee Director ceases to be a Director for any reason, including, without limitation, a termination by resignation, failure to be elected, death or retirement, but excluding terminations where the Participant simultaneously commences or remains in employment or service with the Company or any Subsidiary.

(c) As to an Employee, the time when the employee-employer relationship between a Participant and the Company or any Subsidiary is terminated for any reason, including, without limitation, a termination by resignation, discharge, death, disability or retirement; but excluding terminations where the Participant simultaneously commences or remains in employment or service with the Company or any Subsidiary at an equivalent or higher level position.

The Company, in its sole discretion, shall determine the effect of all matters and questions relating to any Termination of Service, including, without limitation, whether a Termination of Service has occurred, whether a Termination of Service resulted from a discharge for Cause and all questions of whether particular leaves of absence constitute a Termination of Service. For purposes of the Plan, a Participant's employee-employer relationship or consultancy relationship shall be deemed to be terminated in the event that the Subsidiary employing or contracting with such Participant ceases to remain a Subsidiary following any merger, sale of stock or other corporate transaction or event (including, without limitation, a spin-off), even though the Participant may subsequently continue to perform services for that entity.

ARTICLE III. ELIGIBILITY

Service Providers are eligible to be granted Awards under the Plan, subject to the limitations described herein. No Service Provider shall have any right to be granted an Award pursuant to the Plan and neither the Company nor the Administrator is obligated to treat Service Providers, Participants or any other persons uniformly.

ARTICLE IV. ADMINISTRATION AND DELEGATION

4.1 Administration.

(a) The Plan is administered by the Administrator. The Administrator has authority to determine which Service Providers receive Awards, grant Awards and set Award terms and conditions, subject to the conditions and limitations in the Plan. The Administrator also has the authority to take all actions and make all determinations under the Plan, to interpret the Plan and Award Agreements and to adopt, amend and repeal Plan administrative rules, guidelines and practices as it deems advisable. The Administrator may correct defects and ambiguities, supply omissions, reconcile inconsistencies in the Plan or any Award and make all other determinations that it deems necessary or appropriate to administer the Plan and any Awards. The Administrator (and each member thereof) is entitled to, in good faith, rely or act upon any report or other information furnished to it, him or her by any officer or other employee of the Company or any Subsidiary, the Company's independent certified public accountants, or any executive compensation consultant or other professional retained by the Company to assist in the administration of the Plan. The Administrator's determinations under the Plan are in its sole discretion and will be final, binding and conclusive on all persons having or claiming any interest in the Plan or any Award.

(b) Without limiting the foregoing, the Administrator has the exclusive power, authority and sole discretion to: (i) designate Participants; (ii) determine the type or types of Awards to be granted to each Participant; (iii) determine the number of Awards to be granted and the number of Shares to which an Award will relate; (iv) subject to the limitations in the Plan, determine the terms and conditions of any Award and related Award Agreement, including, but not limited to, the exercise price, grant price, purchase price, any performance criteria, any restrictions or limitations on the Award, any schedule for vesting, lapse of forfeiture restrictions or restrictions on the exercisability of an Award, and accelerations, waivers or amendments thereof; (v) determine whether, to what extent, and under what circumstances an Award may be settled in, or the exercise price of an Award may be paid in cash, Shares, or other property, or an Award may be canceled, forfeited, or surrendered; and (vi) make all other decisions and determinations that may be required pursuant to the Plan or as the Administrator deems necessary or advisable to administer the Plan.

4.2 <u>Delegation of Authority</u>. To the extent permitted by Applicable Law, the Board or any Committee may delegate any or all of its powers under the Plan to one or more Committees or officers of the Company or any of its Subsidiaries; provided, however, that in no event shall an officer of the Company or any of its Subsidiaries be delegated the authority to grant Awards to, or amend Awards held by, the following individuals: (a) individuals who are subject to Section 16 of the Exchange Act, or (b) officers of the Company or any of its Subsidiaries or Directors to whom authority to grant or amend Awards has been delegated hereunder. Any delegation hereunder shall be subject to the restrictions and limits that the Board or Committee specifies at the time of such delegation or that are otherwise included in the applicable organizational documents, and the Board or Committee, as applicable, may at any time rescind the authority so delegated or appoint a new delegatee. At all times, the delegatee appointed under this Section 4.2 shall serve in such capacity at the pleasure of the Board or the Committee, as applicable, and the Board or the Committee may abolish any committee at any time and re-vest in itself any previously delegated authority. Further, regardless of any delegation, the Board or a Committee may, in its discretion, exercise any and all rights and duties as the Administrator under the Plan delegated thereby, except with respect to Awards that are required to be determined in the sole discretion of the Committee under the rules of any securities exchange or automated quotation system on which the Shares are listed, quoted or traded.

ARTICLE V. STOCK AVAILABLE FOR AWARDS

5.1 <u>Number of Shares</u>. Subject to adjustment under Article IX and the terms of this Article V, Awards may be made under the Plan covering up to the Overall Share Limit. As of the Effective Date, the Company will cease granting awards under the Prior Plan; however, Prior Plan Awards will remain subject to the terms of the Prior Plan. Shares issued or delivered under the Plan may consist of authorized but unissued Shares, Shares purchased on the open market or treasury Shares.

5.2 Share Recycling.

(a) If all or any part of an Award or Prior Plan Award expires, lapses or is terminated, converted into an award in respect of shares of another entity in connection with a spin-off or other similar event, exchanged or settled for cash, surrendered, repurchased, canceled without having been fully exercised or forfeited, in any case, in a manner that results in the Company acquiring Shares covered by the Award or Prior Plan Award at a price not greater than the price (as adjusted to reflect any Equity Restructuring) paid by the Participant for such Shares or not issuing any Shares covered by the Award or Prior Plan Award, the unused Shares covered by the Award or Prior Plan Award will, as applicable, become or again be available for Awards under the Plan. The payment of Dividend Equivalents in cash in conjunction with any outstanding Awards or Prior Plan Awards shall not count against the Overall Share Limit.

(b) In addition, the following Shares shall be available for future grants of Awards: (i) Shares tendered by a Participant or withheld by the Company in payment of the exercise price of an Option or any stock option granted under the Prior Plan; (ii) Shares tendered by the Participant or withheld by the Company to satisfy any tax withholding obligation with respect to an Award or any Prior Plan Award; and (iii) Shares subject to a Stock Appreciation Right that are not issued in connection with the stock settlement of the Stock Appreciation Right on exercise thereof. Notwithstanding the provisions of this Section 5.2(b), no Shares may again be optioned, granted or awarded pursuant to an Incentive Stock Option if such action would cause such Option to fail to qualify as an incentive stock option under Section 422 of the Code.

5.3 <u>Incentive Stock Option Limitations</u>. Notwithstanding anything to the contrary herein, no more than [_____]Shares (as adjusted to reflect any Equity Restructuring) may be issued pursuant to the exercise of Incentive Stock Options.

5.4 <u>Substitute Awards</u>. In connection with an entity's merger or consolidation with the Company or any Subsidiary or the Company's or any Subsidiary's acquisition of an entity's property or stock, the Administrator may grant Awards in substitution for any options or other stock or stockbased awards granted before such merger or consolidation by such entity or its affiliate. Substitute Awards may be granted on such terms and conditions as the Administrator deems appropriate, notwithstanding limitations on Awards in the Plan. Substitute Awards will not count against the Overall Share Limit (nor shall Shares subject to a Substitute Award be added to the Shares available for Awards under the Plan as provided above), except that Shares acquired by exercise of substitute Incentive Stock Options will count against the maximum number of Shares that may be issued pursuant to the exercise of Incentive Stock Options under the Plan. Additionally, in the event that a company acquired by the Company or any Subsidiary or with which the Company or any Subsidiary combines has shares available under a pre-existing plan approved by stockholders and not adopted in contemplation of such acquisition or combination, the shares available for grant pursuant to the terms of such pre-existing plan (as appropriately adjusted to reflect the transaction) may be used for Awards under the Plan and shall not reduce the Shares authorized for grant under the Plan (and Shares subject to such Awards may again become available for Awards under the Plan as provided under Section 5.2 above); provided that Awards using such available shares shall not be made after the date awards or grants could have been made under the terms of the pre-existing plan, absent the acquisition or combination, and shall only be made to individuals who were not employees or directors of the Company or any of its Subsidiaries prior to such acquisition or combination.

5.5 <u>Non-Employee Director Award Limit</u>. Notwithstanding any provision to the contrary in the Plan or in any policy of the Company regarding non-employee director compensation, the sum of the grant date fair value (determined as of the grant date in accordance with Financial Accounting Standards Board Accounting Standards Codification Topic 718, or any successor thereto) of all equity-based Awards and the maximum amount that may become payable pursuant to all cash-based Awards that may be granted to a Service Provider as compensation for services as a Non-Employee Director during any calendar year shall not exceed \$[1,500,000] for such Service Provider's first year of service as a Non-Employee Director and \$[1,000,000] for each year thereafter.

ARTICLE VI. STOCK OPTIONS AND STOCK APPRECIATION RIGHTS

6.1 General. The Administrator may grant Options or Stock Appreciation Rights to one or more Service Providers, subject to such terms and conditions not inconsistent with the Plan as the Administrator shall determine. The Administrator will determine the number of Shares covered by each Option and Stock Appreciation Right, the exercise price of each Option and Stock Appreciation Right and the conditions and limitations applicable to the exercise of each Option and Stock Appreciation Right. A Stock Appreciation Right will entitle the Participant (or other person entitled to exercise the Stock Appreciation Right) to receive from the Company upon exercise of the exercisable portion of the Stock Appreciation Right an amount determined by multiplying the excess, if any, of the Fair Market Value of one Share on the date of exercise over the exercise price per Share of the Stock Appreciation Right by the number of Shares with respect to which the Stock Appreciation Right is exercised, subject to any limitations of the Plan or that the Administrator may impose and payable in cash, Shares valued at Fair Market Value on the date of exercise or a combination of the two as the Administrator may determine or provide in the Award Agreement.

6.2 Exercise Price. The Administrator will establish each Option's and Stock Appreciation Right's exercise price and specify the exercise price in the Award Agreement. Subject to Section 6.6, the exercise price will not be less than 100% of the Fair Market Value on the grant date of the Option or Stock Appreciation Right. Notwithstanding the foregoing, in the case of an Option or Stock Appreciation Right that is a Substitute Award, the exercise price per share of the Shares subject to such Option or Stock Appreciation Right, as applicable, may be less than the Fair Market Value per share on the date of grant; provided that the exercise price of any Substitute Award shall be determined in accordance with the applicable requirements of Sections 424 and 409A of the Code.

6.3 <u>Duration of Options</u>. Subject to Section 6.6, each Option or Stock Appreciation Right will be exercisable at such times and as specified in the Award Agreement, provided that the term of an Option or Stock Appreciation Right will not exceed ten years; provided, further, that, unless otherwise determined by the Administrator or specified in the Award Agreement, (a) no portion of an Option or Stock Appreciation Right which is unexercisable at a Participant's Termination of Service shall thereafter become exercisable and (b) the portion of an Option or Stock Appreciation Right that is unexercisable at a Participant's Termination of Service shall automatically expire on the date of such Termination of Service. In addition, in no event shall an Option or Stock Appreciation Right granted to an Employee who is a non-exempt employee for purposes of overtime pay under the U.S. Fair Labor Standards Act of 1938 be exercisable earlier than six (6) months after its date of grant. Notwithstanding the foregoing, if the Participant, prior to the end of the term of an Option or Stock Appreciation Right, commits an act of Cause (as determined by the Administrator), or violates any non-competition, non-solicitation or confidentiality provisions of any employment contract, confidentiality and nondisclosure agreement or other agreement between the Participant and the Company or any of its Subsidiaries, the right to exercise the Option or Stock Appreciation Right, as applicable, shall be terminated, unless otherwise determined by the Company, and the Company may suspend the Participant's right to exercise the Option or Stock Appreciation Right when it reasonably believes that the Participant may have participated in any such act or violation.

6.4 Exercise. Options and Stock Appreciation Rights may be exercised by delivering to the Company (or such other person or entity designated by the Administrator) a notice of exercise, in a form and manner the Company approves (which may be written, electronic or telephonic and may contain representations and warranties deemed advisable by the Administrator), signed or authenticated by the person authorized to exercise the Option or Stock Appreciation Right, together with, as applicable, (a) payment in full of the exercise price for the number of Shares for which the Option is exercised in a manner specified in Section 6.5 and (b) satisfaction in full of any withholding obligation for Tax-Related Items in a manner specified in Section 10.5. The Administrator may, in its discretion, limit exercise with respect to fractional Shares and require that any partial exercise of an Option or Stock Appreciation Right be with respect to a minimum number of Shares.

6.5 <u>Payment Upon Exercise</u>. The Administrator shall determine the methods by which payment of the exercise price of an Option shall be made, including, without limitation:

(a) Cash, check or wire transfer of immediately available funds; provided that the Company may limit the use of one of the foregoing methods if one or more of the methods below is permitted;

(b) If there is a public market for Shares at the time of exercise, unless the Company otherwise determines, (A) delivery (including electronically or telephonically to the extent permitted by the Company) of a notice that the Participant has placed a market sell order with a broker acceptable to the Company with respect to Shares then issuable upon exercise of the Option and that the broker has been directed to deliver promptly to the Company funds sufficient to pay the exercise price, or (B) the Participant's delivery to the Company of a copy of irrevocable and unconditional instructions to a broker acceptable to the Company to deliver promptly to the Company an amount sufficient to pay the exercise price by cash, wire transfer of immediately available funds or check; provided that such amount is paid to the Company at such time as may be required by the Company;

(c) To the extent permitted by the Administrator, delivery (either by actual delivery or attestation) of Shares owned by the Participant valued at their Fair Market Value on the date of delivery;

(d) To the extent permitted by the Administrator, surrendering Shares then issuable upon the Option's exercise valued at their Fair Market Value on the exercise date;

(e) To the extent permitted by the Administrator, delivery of a promissory note or any other lawful consideration; or

(f) To the extent permitted by the Administrator, any combination of the above payment forms.

6.6 <u>Additional Terms of Incentive Stock Options</u>. The Administrator may grant Incentive Stock Options only to employees of the Company, any of its present or future parent or subsidiary corporations, as defined in Sections 424(e) or (f) of the Code, respectively, and any other entities the employees of which are eligible to receive Incentive Stock Options under the Code. If an Incentive Stock Option is granted to a Greater Than 10% Stockholder, the exercise price will not be less than 110% of the Fair Market Value on the Option's grant date, and the term of the Option will not exceed five years. All Incentive Stock Options (and Award Agreements related thereto) will be subject to and construed consistently with Section 422 of the Code. By accepting an Incentive Stock Option, the Participant agrees to give prompt notice to the Company of dispositions or other transfers (other than in connection with a Change in Control) of Shares acquired under the Option made within (a) two years from the grant date of the Option or (b) one year after the transfer of such Shares to the Participant, specifying the date of the disposition or other transfer. Neither the Company nor the Administrator will be liable to a Participant, or any other party, if an Incentive Stock Option fails or ceases to qualify as an "incentive stock option" under Section 422 of the Code. Any Incentive Stock Option or portion thereof that fails to qualify as an "incentive stock option" under Treasury Regulation Section 1.422-4, will be a Nonqualified Stock Option.

ARTICLE VII. RESTRICTED STOCK; RESTRICTED STOCK UNITS

7.1 <u>General</u>. The Administrator may grant Restricted Stock, or the right to purchase Restricted Stock, to any Service Provider, subject to forfeiture or the Company's right to repurchase all or part of such shares at their issue price or other stated or formula price from the Participant if conditions the Administrator specifies in the Award Agreement are not satisfied before the end of the applicable restriction period or periods that the Administrator establishes for such Award. In addition, the Administrator may grant Restricted Stock Units, which may be subject to vesting and forfeiture conditions during the applicable restriction period or periods, as set forth in an Award Agreement, to Service Providers. The Administrator shall establish the purchase price, if any, and form of payment for Restricted Stock and Restricted Stock Units; provided, however, that if a purchase price is charged, such purchase price shall be no less than the par value, if any, of the Shares to be purchased, unless otherwise permitted by Applicable Law. In all cases, legal consideration shall be required for each issuance of Restricted Stock Units shall set forth the terms and conditions not inconsistent with the Plan as the Administrator shall determine.

7.2 Restricted Stock.

(a) *Stockholder Rights*. Unless otherwise determined by the Administrator, each Participant holding shares of Restricted Stock will be entitled to all the rights of a stockholder with respect to such Shares, subject to the restrictions in the Plan and the applicable Award Agreement, including the right to receive all dividends and other distributions paid or made with respect to the Shares to the extent such dividends and other distributions have a record date that is on or after the date on which such Participant becomes the record holder of such Shares; provided, however, that with respect to a share of Restricted Stock subject to restrictions or vesting conditions, except in connection with a spin-off or other similar event as otherwise permitted under Section 9.2, dividends which are paid to Company stockholders prior to the removal of restrictions and satisfaction of vesting conditions shall only be paid to the Participant to the extent that the restrictions are subsequently removed and the vesting conditions are subsequently satisfied and the share of Restricted Stock vests.

(b) *Stock Certificates*. The Company may require that the Participant deposit in escrow with the Company (or its designee) any stock certificates issued in respect of shares of Restricted Stock, together with a stock power endorsed in blank.

(c) *Section 83(b) Election*. If a Participant makes an election under Section 83(b) of the Code to be taxed with respect to the Restricted Stock as of the date of transfer of the Restricted Stock rather than as of the date or dates upon which such Participant would otherwise be taxable under Section 83(a) of the Code, such Participant shall be required to deliver a copy of such election to the Company promptly after filing such election with the Internal Revenue Service along with proof of the timely filing thereof.

7.3 <u>Restricted Stock Units</u>. The Administrator may provide that settlement of Restricted Stock Units will occur upon or as soon as reasonably practicable after the Restricted Stock Units vest or will instead be deferred, on a mandatory basis or at the Participant's election, subject to compliance with Applicable Law.

ARTICLE VIII. OTHER TYPES OF AWARDS

8.1 <u>General</u>. The Administrator may grant Performance Stock Unit awards, Performance Bonus Awards, Dividend Equivalents or Other Stock or Cash Based Awards, to one or more Service Providers, in such amounts and subject to such terms and conditions not inconsistent with the Plan as the Administrator shall determine.

8.2 <u>Performance Stock Unit Awards</u>. Each Performance Stock Unit award shall be denominated in a number of Shares or in unit equivalents of Shares or units of value (including a dollar value of Shares) and may be linked to any one or more of performance or other specific criteria, including service to the Company or Subsidiaries, determined to be appropriate by the Administrator, in each case on a specified date or dates or over any period or periods determined by the Administrator. In making such determinations, the Administrator may consider (among such other factors as it deems relevant in light of the specific type of award) the contributions, responsibilities and other compensation of the particular Participant.

8.3 <u>Performance Bonus Awards</u>. Each right to receive a bonus granted under this Section 8.3 shall be denominated in the form of cash (but may be payable in cash, stock or a combination thereof) (a "*Performance Bonus Award*") and shall be payable upon the attainment of performance goals that are established by the Administrator and relate to one or more of performance or other specific criteria, including service to the Company or Subsidiaries, in each case on a specified date or dates or over any period or periods determined by the Administrator.

8.4 <u>Dividend Equivalents</u>. If the Administrator provides, an Award (other than an Option or Stock Appreciation Right) may provide a Participant with the right to receive Dividend Equivalents. Dividend Equivalents may be paid currently or credited to an account for the Participant, settled in cash or Shares and subject to the same restrictions on transferability and forfeitability as the Award with respect to which the Dividend Equivalents are granted and subject to other terms and conditions as set forth in the Award Agreement. Notwithstanding anything to the contrary herein, Dividend Equivalents with respect to an Award subject to vesting shall either (i) to the extent permitted by Applicable Law, not be paid or credited or (ii) be accumulated and subject to vesting to the same extent as the related Award. All such Dividend Equivalents shall be paid at such time as the Administrator shall specify in the applicable Award Agreement.

8.5 <u>Other Stock or Cash Based Awards</u>. Other Stock or Cash Based Awards may be granted to Participants, including Awards entitling Participants to receive cash or Shares to be delivered in the future and annual or other periodic or long-term cash bonus awards (whether based on specified performance criteria or otherwise), in each case subject to any conditions and limitations in the Plan. Such Other Stock or Cash Based Awards will also be available as a payment form in the settlement of other Awards, as standalone payments and as payment in lieu of compensation to which a Participant is otherwise entitled. Other Stock or Cash Based Awards may be paid in Shares, cash or other property, as the Administrator determines. Subject to the provisions of the Plan, the Administrator will determine the terms and conditions of each Other Stock or Cash Based Award, including any purchase price, performance goal(s), transfer restrictions, and vesting conditions, which will be set forth in the applicable Award Agreement. Except in connection with a spin-off or other similar event as otherwise permitted under Article IX, dividends that are paid prior to vesting of any Other Stock or Cash Based Award shall only be paid to the applicable Participant to the extent that the vesting conditions are subsequently satisfied and the Other Stock or Cash Based Award vests.

ARTICLE IX. ADJUSTMENTS FOR CHANGES IN COMMON STOCK AND CERTAIN OTHER EVENTS

9.1 Equity Restructuring(a). In connection with any Equity Restructuring, notwithstanding anything to the contrary in this Article IX the Administrator will equitably adjust the terms of the Plan and each outstanding Award as it deems appropriate to reflect the Equity Restructuring, which may include (i) adjusting the number and type of securities subject to each outstanding Award or with respect to which Awards may be granted under the Plan (including, but not limited to, adjustments of the limitations in Article V hereof on the maximum number and kind of shares that may be issued); (ii) adjusting the terms and conditions of (including the grant or exercise price), and the performance goals or other criteria included in, outstanding Awards; and (iii) granting new Awards or making cash payments to Participants. The adjustments provided under this Section 9.1 will be nondiscretionary and final and binding on all interested parties, including the affected Participant and the Company; provided that the Administrator will determine whether an adjustment is equitable.

9.2 <u>Corporate Transactions</u>. In the event of any dividend or other distribution (whether in the form of cash, Common Stock, other securities, or other property), reorganization, merger, consolidation, split-up, spin off, combination, amalgamation, repurchase, recapitalization, liquidation, dissolution, or sale, transfer, exchange or other disposition of all or substantially all of the assets of the Company, or sale or exchange of Common Stock or other securities of the Company, Change in Control, issuance of warrants or other rights to purchase Common Stock or other securities of the Company, other similar corporate transaction or event, other unusual or nonrecurring transaction or event affecting the Company or its financial statements or any change in any Applicable Law or accounting principles, the Administrator, on such terms and conditions as it deems appropriate, either by the terms of the Award or by action taken prior to the occurrence of such transaction or event (except that action to give effect to a change in Applicable Law or accounting principles may be made within a reasonable period of time after such change) and either automatically or upon the Participant's request, is hereby authorized to take any one or more of the following actions whenever the Administrator determines that such action is appropriate in order to (x) prevent dilution or enlargement of the benefits or potential benefits intended by the Company to be made available under the Plan or with respect to any Award granted or issued under the Plan, (y) to facilitate such transaction or event or (z) give effect to such changes in Applicable Law or accounting principles:

(a) To provide for the cancellation of any such Award in exchange for either an amount of cash or other property with a value equal to the amount that could have been obtained upon the exercise or settlement of the vested portion of such Award or realization of the Participant's rights under the vested portion of such Award, as applicable; provided that, if the amount that could have been obtained upon the exercise or settlement of the vested portion of such Award or realization of the Participant's rights, in any case, is equal to or less than zero, then the Award may be terminated without payment;

(b) To provide that such Award shall vest and, to the extent applicable, be exercisable as to all Shares (or other property) covered thereby, notwithstanding anything to the contrary in the Plan or the provisions of such Award;

(c) To provide that such Award be assumed by the successor or survivor corporation or entity, or a parent or subsidiary thereof, or shall be substituted for by awards covering the stock of the successor or survivor corporation or entity, or a parent or subsidiary thereof, with appropriate adjustments as to the number and kind of shares and applicable exercise or purchase price, in all cases, as determined by the Administrator;

(d) To make adjustments in the number and type of shares of Common Stock (or other securities or property) subject to outstanding Awards or with respect to which Awards may be granted under the Plan (including, but not limited to, adjustments of the limitations in Article V hereof on the maximum number and kind of shares which may be issued) or in the terms and conditions of (including the grant or exercise price), and the criteria included in, outstanding Awards;

(e) To replace such Award with other rights or property selected by the Administrator; or

(f) To provide that the Award will terminate and cannot vest, be exercised or become payable after the applicable event.

9.3 Change in Control.

(a) Notwithstanding any other provision of the Plan, in the event of a Change in Control, unless the Administrator elects to (i) terminate an Award in exchange for cash, rights or property, or (ii) cause an Award to become fully exercisable and no longer subject to any forfeiture restrictions prior to the consummation of a Change in Control, pursuant to Section 9.2, (A) such Award (other than any portion subject to performance-based vesting) shall continue in effect or be assumed or an equivalent Award substituted by the successor corporation or a parent or subsidiary of the successor corporation and (B) the portion of such Award subject to performance-based vesting shall be subject to the terms and conditions of the applicable Award Agreement and, in the absence of applicable terms and conditions, the Administrator's discretion.

(b) In the event that the successor corporation in a Change in Control refuses to assume or substitute for an Award (other than any portion subject to performance-based vesting), the Administrator shall cause such Award to become fully vested and, if applicable, exercisable immediately prior to the consummation of such transaction and all forfeiture restrictions on such Award to lapse and, to the extent unexercised upon the consummation of such transaction, to terminate in exchange for cash, rights or other property. The Administrator shall notify the Participant of any Award that becomes exercisable pursuant to the preceding sentence that such Award shall be fully exercisable for a period of time as determined by the Administrator from the date of such notice (which shall be fifteen (15) days if no period is determined by the Administrator), contingent upon the consummation of the Change in Control, and such Award shall terminate upon the consummation of the Change in Control in accordance with the preceding sentence.

(c) For the purposes of this Section 9.3, an Award shall be considered assumed if, following the Change in Control, the Award confers the right to purchase or receive, for each Share subject to the Award immediately prior to the Change in Control, the consideration (whether stock, cash, or other securities or property) received in the Change in Control by holders of Common Stock for each Share held on the effective date of the transaction (and if holders were offered a choice of consideration, the type of consideration chosen by the holders of a majority of the outstanding Shares); provided, however, that if such consideration received in the Change in Control was not solely common stock of the successor corporation or its parent, the Administrator may, with the consent of the successor corporation, provide for the consideration to be received upon the exercise of the Award, for each Share subject to an Award, to be solely common stock of the successor corporation or its parent equal in fair market value to the per-share consideration received by holders of Common Stock in the Change in Control.

(d) For the avoidance of doubt, each Award shall be subject to any accelerated vesting provisions pursuant to any written agreement between Participant and the Company in accordance with the terms therein.

9.4 <u>Administrative Stand Still</u>. In the event of any pending stock dividend, stock split, combination or exchange of shares, merger, consolidation or other distribution (other than normal cash dividends) of Company assets to stockholders, or any other extraordinary transaction or change affecting the Shares or the share price of Common Stock (including any Equity Restructuring or any securities offering or other similar transaction) or for reasons of administrative convenience or to facilitate compliance with any Applicable Law, the Company may refuse to permit the exercise or settlement of one or more Awards for such period of time as the Company may determine to be reasonably appropriate under the circumstances.

9.5 General. Except as expressly provided in the Plan or the Administrator's action under the Plan, no Participant will have any rights due to any subdivision or consolidation of Shares of any class, dividend payment, increase or decrease in the number of Shares of any class or dissolution, liquidation, merger, or consolidation of the Company or other corporation. Except as expressly provided with respect to an Equity Restructuring under Section 9.1 above or the Administrator's action under the Plan, no issuance by the Company of Shares of any class, or securities convertible into Shares of any class, will affect, and no adjustment will be made regarding, the number of Shares subject to an Award or the Award's grant or exercise price. The existence of the Plan, any Award Agreements and the Awards granted hereunder will not affect or restrict in any way the Company's right or power to make or authorize (i) any adjustment, recapitalization, reorganization or other change in the Company's capital structure or its business, (ii) any merger, consolidation, spinoff, dissolution or liquidation of the Company or sale of Company assets or (iii) any sale or issuance of securities, including securities with rights superior to those of the Shares or securities convertible into or exchangeable for Shares.

ARTICLE X. PROVISIONS APPLICABLE TO AWARDS

10.1 Transferability.

(a) No Award may be sold, assigned, transferred, pledged or otherwise encumbered, either voluntarily or by operation of law, except by will or the laws of descent and distribution, or, subject to the Administrator's consent, pursuant to a DRO, unless and until such Award has been exercised or the Shares underlying such Award have been issued, and all restrictions applicable to such Shares have lapsed. During the life of a Participant, Awards will be exercisable only by the Participant, unless it has been disposed of pursuant to a DRO. After the death of a Participant, any exercisable portion of an Award may, prior to the time when such portion becomes unexercisable under the Plan or the applicable Award Agreement, be exercised by the Participant's personal representative or by any person empowered to do so under the deceased Participant's will or under the then-Applicable Law of descent and distribution. References to a Participant, to the extent relevant in the context, will include references to a transferee approved by the Administrator.

(b) Notwithstanding Section 10.1(a), the Administrator, in its sole discretion, may determine to permit a Participant or a Permitted Transferee of such Participant to transfer an Award other than an Incentive Stock Option (unless such Incentive Stock Option is intended to become a Nonqualified Stock Option) to any one or more Permitted Transferees of such Participant, subject to the following terms and conditions: (i) an Award transferred to a Permitted Transferee shall not be assignable or transferable by the Permitted Transferee other than (A) to another Permitted Transferee of the applicable Participant or (B) by will or the laws of descent and distribution or, subject to the consent of the Administrator, pursuant to a domestic relations order; (ii) an Award transferred to a Permitted Transferee shall continue to be subject to all the terms and conditions of the Award as applicable to the original Participant (other than the ability to further transferee) and the receiving Permitted Transferee shall execute any and all documents requested by the Administrator, including, without limitation documents to (A) confirm the status of the transfere as a Permitted Transferee, (B) satisfy any requirements for an exemption for the transfer under Applicable Law and (C) evidence the transfer; and (iv) any transfer of an Award to a Permitted Transferee shall be without consideration, except as required by Applicable Law. In addition, and further notwithstanding Section 10.1(a), the Administrator, in its sole discretion, may determine to permit a Participant to transfer Incentive Stock Options to a trust that constitutes a Permitted Transferee if, under Section 671 of the Code and other Applicable Law, the Participant is considered the sole beneficial owner of the Incentive Stock Option while it is held in the trust.

(c) Notwithstanding Section 10.1(a), if permitted by the Administrator, a Participant may, in the manner determined by the Administrator, designate a Designated Beneficiary. A Designated Beneficiary, legal guardian, legal representative, or other person claiming any rights pursuant to the Plan is subject to all terms and conditions of the Plan and any Award Agreement applicable to the Participant and any additional restrictions deemed necessary or appropriate by the Administrator. If the Participant is married or a domestic partner in a domestic partnership qualified under Applicable Law and resides in a community property state, a designation of a person other than the Participant's spouse or domestic partner, as applicable, as the Participant's Designated Beneficiary with respect to more than 50% of the Participant's interest in the Award shall not be effective without the prior written or electronic consent of the Participant's spouse or domestic partner. Subject to the foregoing, a beneficiary designation may be changed or revoked by a Participant at any time; provided that the change or revocation is delivered in writing to the Administrator prior to the Participant's death.

10.2 Documentation. Each Award will be evidenced in an Award Agreement in such form as the Administrator determines in its discretion. Each Award may contain such terms and conditions as are determined by the Administrator in its sole discretion, to the extent not inconsistent with those set forth in the Plan.

10.3 <u>Discretion</u>. Except as the Plan otherwise provides, each Award may be made alone or in addition or in relation to any other Award. The terms of each Award to a Participant need not be identical, and the Administrator need not treat Participants or Awards (or portions thereof) uniformly.

10.4 <u>Changes in Participant's Status</u>. The Administrator will determine how the Disability, death, retirement, authorized leave of absence or any other change or purported change in a Participant's Service Provider status affects an Award and the extent to which, and the period during which, the Participant, the Participant's legal representative, conservator, guardian or Designated Beneficiary may exercise rights under the Award, if applicable. Except to the extent otherwise required by Applicable Law or expressly authorized by the Company or by the Company's written policy on leaves of absence, no service credit shall be given for vesting purposes for any period the Participant is on a leave of absence.

10.5 <u>Withholding</u>. Each Participant must pay the Company or a Subsidiary, as applicable, or make provision satisfactory to the Administrator for payment of, any Tax-Related Items required by Applicable Law to be withheld in connection with such Participant's Awards and/or Shares by the date of the event creating the liability for Tax-Related Items. At the Company's discretion and subject to any Company insider trading policy (including black-out periods), any withholding obligation for Tax-Related Items may be satisfied by (i) deducting an amount sufficient to satisfy such withholding obligation from any payment of any kind otherwise due to a Participant; (ii) accepting a payment from the Participant in cash, by wire transfer of immediately available funds, or by check made payable to the order of the Company or a Subsidiary, as applicable; (iii) accepting the delivery of Shares, including Shares delivered by attestation; (iv) retaining Shares from the Award creating the withholding obligation for Tax-Related Items, valued on the date of delivery, (v) if there is a public market for Shares at the time the withholding obligation for Tax-Related Items, selling Shares issued pursuant to the Award creating the withholding obligation for Tax-Related Items, either voluntarily by the Participant or mandatorily by the Company; (vi) accepting delivery of a promissory note or any other lawful consideration; or (vii) any combination of the foregoing payment forms. The amount withheld pursuant to any of the foregoing payment forms shall be determined by the Company and may be up to, but no greater than, the aggregate amount of such obligations based on the maximum statutory withholding rates in the applicable Participant's jurisdiction for all Tax-Related Items that are applicable to such taxable income. If any tax withholding obligation will be satisfied under clause (v) of the preceding paragraph, each Participant's acceptance of an Award under the Plan will constitute the Participant's authorization to the Company and

10.6 <u>Amendment of Award; Repricing</u>. The Administrator may amend, modify or terminate any outstanding Award, including by substituting another Award of the same or a different type, changing the exercise or settlement date, and converting an Incentive Stock Option to a Nonqualified Stock Option. The Participant's consent to such action will be required unless (i) the action, taking into account any related action, does not materially and adversely affect the Participant's rights under the Award, or (ii) the change is permitted under Article IX or pursuant to Section 11.6. In addition, the Administrator shall, without the approval of the stockholders of the Company, have the authority to (a) amend any outstanding Option or Stock Appreciation Right to reduce its exercise price per Share, or (b) cancel any Option or Stock Appreciation Right in exchange for cash or another Award.

10.7 <u>Conditions on Delivery of Stock</u>. The Company will not be obligated to deliver any Shares under the Plan or remove restrictions from Shares previously delivered under the Plan until (i) all Award conditions have been met or removed to the Company's satisfaction, (ii) as determined by the Company, all other legal matters regarding the issuance and delivery of such Shares have been satisfied, including, without limitation, any applicable securities laws and stock exchange or stock market rules and regulations, (iii) any approvals from governmental agencies that the Company determines are necessary or advisable have been obtained, and (iv) the Participant has executed and delivered to the Company such representations or agreements as the Administrator deems necessary or appropriate to satisfy Applicable Law. The inability or impracticability of the Company to obtain or maintain authority from any regulatory body having jurisdiction, which authority is deemed by the Company's counsel to be necessary to the lawful issuance and sale of any Shares hereunder, shall relieve the Company of any liability in respect of the failure to issue or sell such Shares as to which such requisite authority shall not have been obtained, and shall constitute circumstances in which the Administrator may determine to amend or cancel Awards pertaining to such Shares, with or without consideration to the Participant.

10.8 <u>Acceleration</u>. The Administrator may at any time provide that any Award will become immediately vested and fully or partially exercisable, free of some or all restrictions or conditions, or otherwise fully or partially realizable.

ARTICLE XI. MISCELLANEOUS

11.1 <u>No Right to Employment or Other Status</u>. No person will have any claim or right to be granted an Award, and the grant of an Award will not be construed as giving a Participant the right to continue employment or any other relationship with the Company or a Subsidiary. The Company and its Subsidiaries expressly reserve the right at any time to dismiss or otherwise terminate its relationship with a Participant free from any liability or claim under the Plan or any Award, except as expressly provided in an Award Agreement or other written agreement between the Participant and the Company or any Subsidiary.

11.2 <u>No Rights as Stockholder; Certificates</u>. Subject to the Award Agreement, no Participant or Designated Beneficiary will have any rights as a stockholder with respect to any Shares to be distributed under an Award until becoming the record holder of such Shares. Notwithstanding any other provision of the Plan, unless the Administrator otherwise determines or Applicable Law requires, the Company will not be required to deliver to any Participant certificates evidencing Shares issued in connection with any Award and instead such Shares may be recorded in the books of the Company (or, as applicable, its transfer agent or stock plan administrator). The Company may place legends on any share certificate or book entry to reference restrictions applicable to the Shares (including, without limitation, restrictions applicable to Restricted Stock).

11.3 <u>Effective Date</u>. The Plan will become effective on the date immediately prior to the date the Company's registration statement relating to the initial public offering of its Common Stock becomes effective (the "*Effective Date*"). No Incentive Stock Option may be granted pursuant to the Plan after the tenth anniversary of the earlier of (i) the date the Plan was approved by the Board and (ii) the date the Plan was approved by the Company's stockholders.

11.4 <u>Amendment of Plan</u>. The Board may amend, suspend or terminate the Plan at any time and from time to time; provided that (a) no amendment requiring stockholder approval to comply with Applicable Law shall be effective unless approved by the Board, and (b) no amendment, other than an increase to the Overall Share Limit or pursuant to Article IX or Section 11.6, may materially and adversely affect any Award outstanding at the time of such amendment without the affected Participant's consent. No Awards may be granted under the Plan during any suspension period or after Plan termination. Awards outstanding at the time of any Plan suspension or termination will continue to be governed by the Plan and the Award Agreement, as in effect before such suspension or termination. The Board will obtain stockholder approval of any Plan amendment to the extent necessary to comply with Applicable Law.

11.5 <u>Provisions for Foreign Participants</u>. The Administrator may modify Awards granted to Participants who are nationals of a country other than the United States or employed or residing outside the United States, establish subplans or procedures under the Plan or take any other necessary or appropriate action to address Applicable Law, including (a) differences in laws, rules, regulations or customs of such jurisdictions with respect to tax, securities, currency, employee benefit or other matters, (b) listing and other requirements of any non-U.S. securities exchange, and (c) any necessary local governmental or regulatory exemptions or approvals.

11.6 Section 409A.

(a) *General*. The Company intends that all Awards be structured to comply with, or be exempt from, Section 409A, such that no adverse tax consequences, interest, or penalties under Section 409A apply. Notwithstanding anything in the Plan or any Award Agreement to the contrary, the Administrator may, without a Participant's consent, amend this Plan or Awards, adopt policies and procedures, or take any other actions (including amendments, policies, procedures and retroactive actions) as are necessary or appropriate to preserve the intended tax treatment of Awards, including any such actions intended to (A) exempt this Plan or any Award from Section 409A, or (B) comply with Section 409A, including regulations, guidance, compliance programs and other interpretative authority that may be issued after an Award's grant date. The Company makes no representations or warranties as to an Award's tax treatment under Section 409A or otherwise. The Company will have no obligation under this Section 11.6 or otherwise to avoid the taxes, penalties or interest under Section 409A with respect to any Award and will have no liability to any Participant or any other person if any Award, compensation or other benefits under the Plan are determined to constitute noncompliant "nonqualified deferred compensation" subject to taxes, penalties or interest under Section 409A.

(b) *Separation from Service*. If an Award constitutes "nonqualified deferred compensation" under Section 409A, any payment or settlement of such Award upon a Participant's Termination of Service will, to the extent necessary to avoid taxes under Section 409A, be made only upon the Participant's "separation from service" (within the meaning of Section 409A), whether such "separation from service" occurs upon or after the Participant's Termination of Service. For purposes of this Plan or any Award Agreement relating to any such payments or benefits, references to a "termination," "termination of employment" or like terms means a "separation from service."

(c) *Payments to Specified Employees*. Notwithstanding any contrary provision in the Plan or any Award Agreement, any payment(s) of "nonqualified deferred compensation" required to be made under an Award to a "specified employee" (as defined under Section 409A and as the Administrator determines) due to his or her "separation from service" will, to the extent necessary to avoid taxes under Section 409A(a)(2)(B)(i) of the Code, be delayed for the six-month period immediately following such "separation from service" (or, if earlier, until the specified employee's death) and will instead be paid (as set forth in the Award Agreement) on the day immediately following such six-month period or as soon as administratively practicable thereafter (without interest). Any payments of "nonqualified deferred compensation" under such Award payable more than six months following the Participant's "separation from service" will be paid at the time or times the payments are otherwise scheduled to be made.

11.7 Limitations on Liability. Notwithstanding any other provisions of the Plan, no individual acting as a director, officer or other employee of the Company or any Subsidiary will be liable to any Participant, former Participant, spouse, beneficiary, or any other person for any claim, loss, liability, or expense incurred in connection with the Plan or any Award, and such individual will not be personally liable with respect to the Plan because of any contract or other instrument executed in his or her capacity as an Administrator, director, officer or other employee of the Company or any Subsidiary that has been or will be granted or delegated any duty or power relating to the Plan's administration or interpretation, against any cost or expense (including attorneys' fees) or liability (including any sum paid in settlement of a claim with the Administrator's approval) arising from any act or omission concerning this Plan unless arising from such person's own fraud or bad faith; provided that he or she gives the Company an opportunity, at its own expense, to handle and defend the same before he or she undertakes to handle and defend it on his or her own behalf.

11.8 <u>Severability</u>. If any portion of the Plan or any action taken under it is held illegal or invalid for any reason, the illegality or invalidity will not affect the remaining parts of the Plan, and the Plan will be construed and enforced as if the illegal or invalid provisions had been excluded, and the illegal or invalid action will be null and void.

11.9 Data Privacy. As a condition for receiving any Award, each Participant explicitly and unambiguously consents to the collection, use and transfer, in electronic or other form, of personal data as described in this Section by and among the Company and its Subsidiaries and affiliates exclusively for implementing, administering and managing the Participant's participation in the Plan. The Company and its Subsidiaries and affiliates may hold certain personal information about a Participant, including the Participant's name, address and telephone number; birthdate; social security, insurance number or other identification number; salary; nationality; job title(s); any Shares held in the Company or its Subsidiaries and affiliates; and Award details, to implement, manage and administer the Plan and Awards (the "Data"). The Company and its Subsidiaries and affiliates may transfer the Data amongst themselves as necessary to implement, administer and manage a Participant's participation in the Plan, and the Company and its Subsidiaries and affiliates may transfer the Data to third parties assisting the Company with Plan implementation, administration and management. These recipients may be located in the Participant's country, or elsewhere, and the Participant's country may have different data privacy laws and protections than the recipients' country. By accepting an Award, each Participant authorizes such recipients to receive, possess, use, retain and transfer the Data, in electronic or other form, to implement, administer and manage the Participant's participation in the Plan, including any required Data transfer to a broker or other third party with whom the Company or the Participant may elect to deposit any Shares. The Data related to a Participant will be held only as long as necessary to implement, administer, and manage the Participant's participation in the Plan. A Participant may, at any time, view the Data that the Company holds regarding such Participant, request additional information about the storage and processing of the Data regarding such Participant, recommend any necessary corrections to the Data regarding the Participant or refuse or withdraw the consents in this Section 11.9 in writing, without cost, by contacting the local human resources representative. The Company may cancel Participant's ability to participate in the Plan and, in the Administrator's sole discretion, the Participant may forfeit any outstanding Awards if the Participant refuses or withdraws the consents in this Section 11.9. For more information on the consequences of refusing or withdrawing consent, Participants may contact their local human resources representative.

11.10 <u>Governing Documents</u>. If any contradiction occurs between the Plan and any Award Agreement or other written agreement between a Participant and the Company (or any Subsidiary), the Plan will govern, unless such Award Agreement or other written agreement was approved by the Administrator and expressly provides that a specific provision of the Plan will not apply.

11.11 <u>Governing Law</u>. The Plan and all Awards will be governed by and interpreted in accordance with the laws of the State of Delaware, without regard to the conflict of law rules thereof or of any other jurisdiction.

11.12 <u>Clawback Provisions</u>. All Awards (including the gross amount of any proceeds, gains or other economic benefit the Participant actually or constructively receives upon receipt or exercise of any Award or the receipt or resale of any Shares underlying the Award) will be subject to recoupment by the Company to the extent required to comply with Applicable Law or any policy of the Company providing for the reimbursement of incentive compensation, whether or not such policy was in place at the time of grant of an Award.

11.13 <u>Titles and Headings</u>. The titles and headings in the Plan are for convenience of reference only and, if any conflict, the Plan's text, rather than such titles or headings, will control.

11.14 <u>Conformity to Applicable Law</u>. Participant acknowledges that the Plan is intended to conform to the extent necessary with Applicable Law. Notwithstanding anything herein to the contrary, the Plan and all Awards will be administered only in a manner intended to conform with Applicable Law. To the extent Applicable Law permits, the Plan and all Award Agreements will be deemed amended as necessary to conform to Applicable Law.

11.15 <u>Relationship to Other Benefits</u>. No payment under the Plan will be taken into account in determining any benefits under any pension, retirement, savings, profit sharing, group insurance, welfare or other benefit plan of the Company or any Subsidiary, except as expressly provided in writing in such other plan or an agreement thereunder.

11.16 <u>Unfunded Status of Awards</u>. The Plan is intended to be an "unfunded" plan for incentive compensation. With respect to any payments not yet made to a Participant pursuant to an Award, nothing contained in the Plan or Award Agreement shall give the Participant any rights that are greater than those of a general creditor of the Company or any Subsidiary.

11.17 <u>Limitations Applicable to Section 16 Persons</u>. Notwithstanding any other provision of the Plan, the Plan and any Award granted or awarded to any individual who is then subject to Section 16 of the Exchange Act shall be subject to any additional limitations set forth in any applicable exemptive rule under Section 16 of the Exchange Act (including Rule 16b-3 of the Exchange Act and any amendments thereto) that are requirements for the application of such exemptive rule. To the extent permitted by Applicable Law, the Plan and Awards granted or awarded hereunder shall be deemed amended to the extent necessary to conform to such applicable exemptive rule.

11.18 <u>Prohibition on Executive Officer and Director Loans</u>. Notwithstanding any other provision of the Plan to the contrary, no Participant who is a Director or an "executive officer" of the Company within the meaning of Section 13(k) of the Exchange Act shall be permitted to make payment with respect to any Awards granted under the Plan, or continue any extension of credit with respect to such payment, with a loan from the Company or a loan arranged by the Company in violation of Section 13(k) of the Exchange Act.

11.19 <u>Broker-Assisted Sales</u>. In the event of a broker-assisted sale of Shares in connection with the payment of amounts owed by a Participant under or with respect to the Plan or Awards, including amounts to be paid under Section 10.5: (a) any Shares to be sold through the broker-assisted sale will be sold on the day the payment first becomes due, or as soon thereafter as practicable; (b) such Shares may be sold as part of a block trade with other Participants in the Plan in which all Participants receive an average price; (c) the applicable Participant will be responsible for all broker's fees and other costs of sale, and by accepting an Award, each Participant agrees to indemnify and hold the Company harmless from any losses, costs, damages, or expenses relating to any such sale; (d) to the extent the Company or its designee receives proceeds of such sale that exceed the amount owed, the Company will pay such excess in cash to the applicable Participant as soon as reasonably practicable; (e) the Company and its designees are under no obligation to arrange for such sale at any particular price; and (f) in the event the proceeds of such sale are insufficient to satisfy the Participant's applicable obligation, the Participant may be required to pay immediately upon demand to the Company or its designee an amount in cash sufficient to satisfy any remaining portion of the Participant's obligation.

* * * * *

FRONTIER GROUP HOLDINGS, INC. 2021 INCENTIVE AWARD PLAN STOCK OPTION GRANT NOTICE

Frontier Group Holdings, Inc., a Delaware corporation, (the "<u>Company</u>"), pursuant to its 2021 Incentive Award Plan, as may be amended from time to time (the "<u>Plan</u>"), hereby grants to the holder listed below ("<u>Participant</u>"), an option to purchase the number of shares of the Company's Common Stock (the "<u>Shares</u>"), set forth below (the "<u>Option</u>"). This Option is subject to all of the terms and conditions set forth herein, as well as in the Plan and the Stock Option Agreement attached hereto as <u>Exhibit A</u> (the "<u>Stock Option Agreement</u>"), each of which are incorporated herein by reference. Unless otherwise defined herein, the terms defined in the Plan shall have the same defined meanings in this Stock Option Grant Notice (the "<u>Grant</u> <u>Notice</u>") and the Stock Option Agreement. If the Company uses an electronic capitalization system and the fields below are blank or the information is otherwise provided in a different format electronically, the blank fields and other information shall be deemed incorporated herein from the electronic capitalization system and considered part of this Grant Notice.

Participant:	[]
Grant Date:	[]
Vesting Commencement Date:	[]
Exercise Price per Share:	\$[]
Total Exercise Price:	[]
Total Number of Shares Subject to the Option:	[] shares
Expiration Date:	[]
Vesting Schedule:	[]

Type of Option:

 \Box Incentive Stock Option

□ Nonqualified Stock Option

By Participant's signature below or by electronic acceptance or authentication in a form authorized by the Company, Participant agrees to be bound by the terms and conditions of the Plan, the Stock Option Agreement, and this Grant Notice. Participant has reviewed the Stock Option Agreement, the Plan and this Grant Notice in their entirety, has had an opportunity to obtain the advice of counsel prior to Participant's execution of this Grant Notice or electronic acceptance or authentication and fully understands all provisions of this Grant Notice, the Stock Option Agreement and the Plan. Participant hereby agrees to accept as binding, conclusive and final all decisions or interpretations of the Administrator upon any questions arising under the Plan, this Grant Notice or the Stock Option Agreement.

FRONTIER	GROUP	HOLDINGS	. INC.:
I ICOI (I ILIC	oncour	nonputto	,

By: Print Name:

ву:		
Print Name:		
Title:		

EXHIBIT A TO STOCK OPTION GRANT NOTICE

FRONTIER GROUP HOLDINGS, INC. STOCK OPTION AGREEMENT

Pursuant to the Stock Option Grant Notice (the "<u>Grant Notice</u>") to which this Stock Option Agreement (this "<u>Agreement</u>") is attached, Frontier Group Holdings, Inc., a Delaware corporation (the "<u>Company</u>"), has granted to the Participant an Option under the Company's 2021 Incentive Award Plan, as may be amended from time to time (the "<u>Plan</u>"), to purchase the number of Shares indicated in the Grant Notice.

ARTICLE 1.

GENERAL

1.1 <u>Defined Terms</u>. Wherever the following terms are used in this Agreement they shall have the meanings specified below, unless the context clearly indicates otherwise. Capitalized terms not specifically defined herein shall have the meanings specified in the Plan and the Grant Notice.

1.2 <u>Incorporation of Terms of Plan</u>. The Option is subject to the terms and conditions of the Plan which are incorporated herein by reference. In the event of any inconsistency between the Plan and this Agreement, the terms of the Plan shall control.

ARTICLE 2.

GRANT OF OPTION

2.1 <u>Grant of Option</u>. In consideration of the Participant's past and/or continued employment with or service to the Company or any Subsidiary and for other good and valuable consideration, effective as of the Grant Date set forth in the Grant Notice (the "<u>Grant Date</u>"), the Company irrevocably grants to the Participant the Option to purchase any part or all of an aggregate of the number of Shares set forth in the Grant Notice, upon the terms and conditions set forth in the Plan and this Agreement, subject to adjustments as provided in Article IX of the Plan. Unless designated as a Nonqualified Stock Option in the Grant Notice, the Option shall be an Incentive Stock Option to the maximum extent permitted by law.

2.2 <u>Exercise Price</u>. The exercise price of the Shares subject to the Option shall be as set forth in the Grant Notice, without commission or other charge; *provided*, *however*, that the price per share of the Shares subject to the Option shall not be less than 100% of the Fair Market Value of a Share on the Grant Date. Notwithstanding the foregoing, if this Option is designated as an Incentive Stock Option and the Participant is a Greater Than 10% Stockholder as of the Grant Date, the exercise price per share of the Shares subject to the Option shall not be less than 110% of the Fair Market Value of a Share on the Grant Date.

2.3 <u>Consideration to the Company</u>. In consideration of the grant of the Option by the Company, the Participant agrees to render faithful and efficient services to the Company or any Subsidiary. Nothing in the Plan or this Agreement shall confer upon the Participant any right to continue in the employ or service of the Company or any Subsidiary or shall interfere with or restrict in any way the rights of the Company and its Subsidiaries, which rights are hereby expressly reserved, to discharge or terminate the services of the Participant at any time for any reason whatsoever, with or without cause, except to the extent expressly provided otherwise in a written agreement between the Company or a Subsidiary and the Participant.

ARTICLE 3.

PERIOD OF EXERCISABILITY

3.1 Commencement of Exercisability.

(a) Subject to Sections 3.2, 3.3, 5.11 and 5.17 hereof, the Option shall become vested and exercisable in such amounts and at such times as are set forth in the Grant Notice.

(b) No portion of the Option which has not become vested and exercisable at the date of the Participant's Termination of Service shall thereafter become vested and exercisable, except as may be otherwise provided by the Administrator or as set forth in a written agreement between the Company and the Participant.

(c) Notwithstanding Section 3.1(a) hereof and the Grant Notice, but subject to Section 3.1(b) hereof, in the event of a Change in Control the Option shall be treated pursuant to Sections 9.2 and 9.3 of the Plan.

3.2 <u>Duration of Exercisability</u>. The installments provided for in the vesting schedule set forth in the Grant Notice are cumulative. Each such installment which becomes vested and exercisable pursuant to the vesting schedule set forth in the Grant Notice shall remain vested and exercisable until it becomes unexercisable under Section 3.3 hereof.

3.3 Expiration of Option. The Option may not be exercised to any extent by anyone after the first to occur of the following events:

(a) The Expiration Date set forth in the Grant Notice, which shall in no event be more than ten (10) years from the Grant Date;

(b) If this Option is designated as an Incentive Stock Option and the Participant, at the time the Option was granted, was a Greater Than 10% Stockholder, the expiration of five (5) years from the Grant Date;

(c) The expiration of three (3) months from the date of the Participant's Termination of Service, unless such termination occurs by reason of the Participant's death or Disability or Cause;

(d) The expiration of one (1) year from the date of the Participant's Termination of Service by reason of the Participant's death or Disability; or

(e) The Participant's Termination of Service for Cause.

3.4 <u>Special Tax Consequences</u>. The Participant acknowledges that, to the extent that the aggregate Fair Market Value (determined as of the time the Option is granted) of all Shares with respect to which Incentive Stock Options, including the Option (if applicable), are exercisable for the first time by the Participant in any calendar year exceeds \$100,000, the Option and such other options shall be Nonqualified Stock Options to the extent necessary to comply with the limitations imposed by Section 422(d) of the Code. The Participant further acknowledges that the rule set forth in the preceding sentence shall be applied by taking the Option and other "incentive stock options" into account in the order in which they were granted, as determined under Section 422(d) of the Code and the Treasury Regulations thereunder. The Participant also acknowledges that an Incentive Stock Option exercised more than three (3) months after the Participant's Termination of Employment, other than by reason of death or Disability, will be taxed as a Nonqualified Stock Option.

3.5 Tax Indemnity.

(a) The Participant agrees to indemnify and keep indemnified the Company, any Subsidiary and the Participant's employing company, if different, from and against any liability for or obligation to pay any Tax Related Items that is attributable to (1) the grant or exercise of, or any benefit derived by the Participant from, the Option, (2) the acquisition by the Participant of the Shares on exercise of the Option or (3) the disposal of any Shares.

(b) The Option cannot be exercised until the Participant has made such arrangements as the Company may require for the satisfaction of any Tax Related Items that may arise in connection with the exercise of the Option and/or the acquisition of the Shares by the Participant. The Company shall not be required to issue, allot or transfer Shares until the Participant has satisfied this obligation.

(c) The Participant hereby acknowledges that the Company (i) makes no representations or undertakings regarding the treatment of any Tax Related Items in connection with any aspect of the Option and (ii) does not commit to and is under no obligation to structure the terms of the grant or any aspect of any Award, including the Option, to reduce or eliminate the Participant's liability for Tax Related Items or achieve any particular tax result. Furthermore, if the Participant becomes subject to tax in more than one jurisdiction between the date of grant of an Award, including the Option, and the date of any relevant taxable event, the Participant acknowledges that the Company may be required to withhold or account for Tax Related Items in more than one jurisdiction.

ARTICLE 4.

EXERCISE OF OPTION

4.1 <u>Person Eligible to Exercise</u>. Except as provided in Section 5.3 hereof, during the lifetime of the Participant, only the Participant may exercise the Option or any portion thereof, unless it has been disposed of pursuant to a DRO. After the death of the Participant, any exercisable portion of the Option may, prior to the time when the Option becomes unexercisable under Section 3.3 hereof, be exercised by the deceased Participant's personal representative or by any person empowered to do so under the deceased Participant's will or under the then applicable laws of descent and distribution.

4.2 <u>Partial Exercise</u>. Any exercisable portion of the Option or the entire Option, if then wholly exercisable, may be exercised in whole or in part at any time prior to the time when the Option or portion thereof becomes unexercisable under Section 3.3 hereof. However, the Option shall not be exercisable with respect to fractional Shares.

4.3 <u>Manner of Exercise</u>. The Option, or any exercisable portion thereof, may be exercised solely by delivery to the Secretary of the Company (or any third party administrator or other person or entity designated by the Company; for the avoidance of doubt, delivery shall include electronic delivery), during regular business hours, of all of the following prior to the time when the Option or such portion thereof becomes unexercisable under Section 3.3 hereof:

(a) An exercise notice in a form specified by the Administrator, stating that the Option or portion thereof is thereby exercised, such notice complying with all applicable rules established by the Administrator. The notice shall be signed by the Participant or other person then entitled to exercise the Option or such portion of the Option;

(b) The receipt by the Company of full payment for the Shares with respect to which the Option or portion thereof is exercised, including payment of any applicable Tax Related Items, which shall be made by deduction from other compensation payable to the Participant or in such other form of consideration permitted under Section 4.4 hereof that is acceptable to the Company;

(c) Any other written representations or documents as may be required in the Administrator's sole discretion to evidence compliance with the Securities Act, the Exchange Act or any other applicable law, rule or regulation; and

(d) In the event the Option or portion thereof shall be exercised pursuant to Section 4.1 hereof by any person or persons other than the Participant, appropriate proof of the right of such person or persons to exercise the Option.

Notwithstanding any of the foregoing, the Company shall have the right to specify all conditions of the manner of exercise, which conditions may vary by country and which may be subject to change from time to time.

4.4 <u>Method of Payment</u>. Payment of the exercise price shall be by any of the following, or a combination thereof, at the election of the Participant:

(a) Cash or check;

(b) With the consent of the Administrator, surrender of Shares (including, without limitation, Shares otherwise issuable upon exercise of the Option) held for such period of time as may be required by the Administrator in order to avoid adverse accounting consequences and having a Fair Market Value on the date of delivery equal to the aggregate exercise price of the Option or exercised portion thereof; or

(c) Other legal consideration acceptable to the Administrator (including, without limitation, through the delivery of a notice that the Participant has placed a market sell order with a broker with respect to Shares then issuable upon exercise of the Option, and that the broker has been directed to pay a sufficient portion of the net proceeds of the sale to the Company in satisfaction of the Option exercise price; *provided* that payment of such proceeds is then made to the Company at such time as may be required by the Company, but in any event not later than the settlement of such sale).

4.5 <u>Conditions to Issuance of Shares</u>. The Shares deliverable upon the exercise of the Option, or any portion thereof, may be either previously authorized but unissued Shares or issued Shares which have then been reacquired by the Company. Such Shares shall be fully paid and nonassessable. The Company shall not be required to issue or deliver any Shares purchased upon the exercise of the Option or portion thereof prior to fulfillment of all of the conditions in Section 10.7 of the Plan and following conditions:

(a) The admission of such Shares to listing on all stock exchanges on which such Shares are then listed;

(b) The completion of any registration or other qualification of such Shares under any state or federal law or under rulings or regulations of the Securities and Exchange Commission or of any other governmental regulatory body, which the Administrator shall, in its absolute discretion, deem necessary or advisable;

(c) The obtaining of any approval or other clearance from any state or federal governmental agency which the Administrator shall, in its absolute discretion, determine to be necessary or advisable;

(d) The receipt by the Company of full payment for such Shares, including payment of any applicable Tax Related Items, which may be in one or more of the forms of consideration permitted under Section 4.4 hereof; and

(e) The lapse of such reasonable period of time following the exercise of the Option as the Administrator may from time to time establish for reasons of administrative convenience.

4.6 <u>Rights as Stockholder</u>. The holder of the Option shall not be, nor have any of the rights or privileges of, a stockholder of the Company, including, without limitation, voting rights and rights to dividends, in respect of any Shares purchasable upon the exercise of any part of the Option unless and until such Shares shall have been issued by the Company and held of record by such holder (as evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company). No adjustment will be made for a dividend or other right for which the record date is prior to the date the Shares are issued, except as provided in Article IX of the Plan.

ARTICLE 5.

OTHER PROVISIONS

5.1 <u>Administration</u>. The Administrator shall have the power to interpret the Plan and this Agreement and to adopt such rules for the administration, interpretation and application of the Plan as are consistent therewith and to interpret, amend or revoke any such rules. All actions taken and all interpretations and determinations made by the Administrator in good faith shall be final and binding upon the Participant, the Company and all other interested persons. No member of the Committee or the Board shall be personally liable for any action, determination or interpretation made in good faith with respect to the Plan, this Agreement or the Option.

5.2 Whole Shares. The Option may only be exercised for whole Shares.

5.3 Transferability.

(a) Subject to Section 4.1 hereof, the Option may not be sold, pledged, assigned or transferred in any manner other than by will or the laws of descent and distribution or, subject to the consent of the Administrator, pursuant to a DRO, unless and until the Option has been exercised and the Shares underlying the Option have been issued, and all restrictions applicable to such Shares have lapsed. Neither the Option nor any interest or right therein shall be liable for the debts, contracts or engagements of the Participant or his or her successors in interest or shall be subject to disposition by transfer, alienation, anticipation, pledge, hypothecation, encumbrance, assignment or any other means whether such disposition be voluntary or involuntary or by operation of law by judgment, levy, attachment, garnishment or any other legal or equitable proceedings (including bankruptcy) unless and until the Option has been exercised, and any attempted disposition thereof prior to exercise shall be null and void and of no effect, except to the extent that such disposition is permitted by the preceding sentence.

(b) During the lifetime of the Participant, only the Participant may exercise the Option (or any portion thereof), unless it has been disposed of pursuant to a DRO; after the death of the Participant, any exercisable portion of the Option may, prior to the time when such portion becomes unexercisable under the Plan or this Agreement, be exercised by the Participant's personal representative or by any person empowered to do so under the deceased Participant's will or under the then-applicable laws of descent and distribution.

(c) Notwithstanding any other provision in this Agreement, the Participant may, in the manner determined by the Administrator, designate a beneficiary to exercise the rights of the Participant and to receive any distribution with respect to the Option upon the Participant's death. A beneficiary, legal guardian, legal representative, or other person claiming any rights pursuant to the Plan is subject to all terms and conditions of the Plan and this Agreement, except to the extent the Plan and this Agreement otherwise provide, and to any additional restrictions deemed necessary or appropriate by the Administrator. If the Participant is married or a domestic partner in a domestic partnership qualified under Applicable Law and resides in a community property state, a designation of a person other than the Participant's spouse or domestic partner, as applicable, as his or her beneficiary with respect to more than 50% of the Participant's interest in the Option shall not be effective without the prior written consent of the Participant's spouse or domestic partner. If no beneficiary has been designated or survives the Participant, payment shall be made to the person entitled thereto pursuant to the Participant's will or the laws of descent and distribution. Subject to the foregoing, a beneficiary designation may be changed or revocked by the Participant at any time provided the change or revocation is filed with the Administrator prior to the Participant's death.

5.4 <u>Tax Consultation</u>. The Participant understands that the Participant may suffer adverse tax consequences as a result of the grant, vesting and/or exercise of the Option, and/or with the purchase or disposition of the Shares subject to the Option. The Participant represents that the Participant has consulted with any tax consultants the Participant deems advisable in connection with the purchase or disposition of such Shares and that the Participant is not relying on the Company for any tax advice.

5.5 <u>Binding Agreement</u>. Subject to the limitation on the transferability of the Option contained herein, this Agreement will be binding upon and inure to the benefit of the heirs, legatees, legal representatives, successors and assigns of the parties hereto.

5.6 <u>Adjustments Upon Specified Events</u>. The Administrator may accelerate the vesting of the Option in such circumstances as it, in its sole discretion, may determine. In addition, upon the occurrence of certain events relating to the Shares contemplated by Article IX of the Plan (including, without limitation, an extraordinary cash dividend on such Shares), the Administrator shall make such adjustments the Administrator deems appropriate in the number of Shares subject to the Option, the exercise price of the Option and the kind of securities that may be issued upon exercise of the Option. The Participant acknowledges that the Option is subject to adjustment, modification and termination in certain events as provided in this Agreement and Article IX of the Plan.

5.7 <u>Notices</u>. Any notice to be given under the terms of this Agreement to the Company shall be addressed to the Company in care of the Secretary of the Company at the Company's principal office, and any notice to be given to the Participant shall be addressed to the Participant at the Participant's last address reflected on the Company's records. By a notice given pursuant to this Section 5.7, either party may hereafter designate a different address for notices to be given to that party. Any notice which is required to be given to the Participant shall, if the Participant is then deceased, be given to the person entitled to exercise his or her Option pursuant to Section 4.1 hereof by written notice under this Section 5.7. Any notice shall be deemed duly given when sent via email or when sent by certified mail (return receipt requested) and deposited (with postage prepaid) in a post office or branch post office regularly maintained by the United States Postal Service (or similar foreign entity).

5.8 Titles. Titles are provided herein for convenience only and are not to serve as a basis for interpretation or construction of this Agreement.

5.9 <u>Governing Law</u>. The laws of the State of Delaware shall govern the interpretation, validity, administration, enforcement and performance of the terms of this Agreement regardless of the law that might be applied under principles of conflicts of laws.

5.10 <u>Conformity to Securities Laws</u>. The Participant acknowledges that the Plan and this Agreement are intended to conform to the extent necessary with all provisions of the Securities Act and the Exchange Act and any and all Applicable Law and regulations and rules promulgated by the Securities and Exchange Commission thereunder, and state securities laws and regulations. Notwithstanding anything herein to the contrary, the Plan shall be administered, and the Option is granted and may be exercised, only in such a manner as to conform to such Applicable Law. To the extent permitted by Applicable Law, the Plan and this Agreement shall be deemed amended to the extent necessary to conform to such Applicable Law.

5.11 <u>Amendment, Suspension and Termination</u>. To the extent permitted by the Plan, this Agreement may be wholly or partially amended or otherwise modified, suspended or terminated at any time or from time to time by the Administrator or the Board; *provided, however*, that, except as may otherwise be provided by the Plan, no amendment, modification, suspension or termination of this Agreement shall adversely affect the Option in any material way without the prior written consent of the Participant.

5.12 <u>Successors and Assigns</u>. The Company may assign any of its rights under this Agreement to single or multiple assignees, and this Agreement shall inure to the benefit of the successors and assigns of the Company. Subject to the restrictions on transfer herein set forth in Section 5.3 hereof, this Agreement shall be binding upon the Participant and his or her heirs, executors, administrators, successors and assigns.

5.13 Notification of Disposition. If this Option is designated as an Incentive Stock Option, the Participant shall give prompt notice to the Company of any disposition or other transfer of any Shares acquired under this Agreement if such disposition or transfer is made (a) within two (2) years from the Grant Date with respect to such Shares or (b) within one (1) year after the transfer of such Shares to the Participant. Such notice shall specify the date of such disposition or other transfer and the amount realized, in cash, other property, assumption of indebtedness or other consideration, by the Participant in such disposition or other transfer.

5.14 Limitations Applicable to Section 16 Persons. Notwithstanding any other provision of the Plan or this Agreement, if the Participant is subject to Section 16 of the Exchange Act, the Plan, the Option and this Agreement shall be subject to any additional limitations set forth in any applicable exemptive rule under Section 16 of the Exchange Act (including any amendment to Rule 16b-3 of the Exchange Act) that are requirements for the application of such exemptive rule. To the extent permitted by applicable law, this Agreement shall be deemed amended to the extent necessary to conform to such applicable exemptive rule.

5.15 Not a Contract of Service Relationship. Nothing in this Agreement or in the Plan shall confer upon the Participant any right to continue to serve as an employee or other service provider of the Company or any of its Subsidiaries or interfere with or restrict in any way with the right of the Company or any of its Subsidiaries, which rights are hereby expressly reserved, to discharge or to terminate for any reason whatsoever, with or without cause, the services of the Participant's at any time, except to the extent expressly provided otherwise by Applicable Law or in a written agreement between the Company or a Subsidiary and the Participant.

5.16 <u>Entire Agreement</u>. The Plan, the Grant Notice and this Agreement constitute the entire agreement of the parties and supersede in their entirety all prior undertakings and agreements of the Company and the Participant with respect to the subject matter hereof, provided that the Option shall be subject to any accelerated vesting provisions in any written agreement between the Participant and the Company (or any Subsidiary that is the employer of the Participant) or Company plan pursuant to which the Participant participates, in each case, in accordance with the terms therein.

5.17 Section 409A. This Option is not intended to constitute "nonqualified deferred compensation" within the meaning of Section 409A of the Code (together with any Department of Treasury regulations and other interpretive guidance issued thereunder, including without limitation any such regulations or other guidance that may be issued after the date hereof, "Section 409A"). However, notwithstanding any other provision of the Plan, the Grant Notice or this Agreement, if at any time the Administrator determines that the Option (or any portion thereof) may be subject to Section 409A, the Administrator shall have the right in its sole discretion (without any obligation to do so or to indemnify the Participant or any other person for failure to do so) to adopt such amendments to the Plan, the Grant Notice or this Agreement, or adopt other policies and procedures (including amendments, policies and procedures with retroactive effect), or take any other actions, as the Administrator determines are necessary or appropriate either for the Option to be exempt from the application of Section 409A or to comply with the requirements of Section 409A.

5.18 Limitation on the Participant's Rights. Participation in the Plan confers no rights or interests other than as herein provided. This Agreement creates only a contractual obligation on the part of the Company as to amounts payable and shall not be construed as creating a trust. Neither the Plan nor any underlying program, in and of itself, has any assets. The Participant shall have only the rights of a general unsecured creditor of the Company with respect to amounts credited and benefits payable, if any, with respect to the Option, and rights no greater than the right to receive the Shares as a general unsecured creditor with respect to options, as and when exercised pursuant to the terms hereof.

FRONTIER GROUP HOLDINGS, INC. **2021 INCENTIVE AWARD PLAN RESTRICTED STOCK AWARD GRANT NOTICE**

Frontier Group Holdings, Inc., a Delaware corporation, (the "Company"), pursuant to its 2021 Incentive Award Plan, as amended from time to time (the "Plan"), hereby grants to the holder listed below (the "Participant") the number of shares of the Company's Common Stock set forth below (the "Shares") subject to all of the terms and conditions as set forth herein and in the Restricted Stock Award Agreement attached hereto as Exhibit A (the "Agreement") (including without limitation the Restrictions on the Shares set forth in the Agreement) and the Plan, each of which is incorporated herein by reference. Unless otherwise defined herein, the terms defined in the Plan shall have the same defined meanings in this Restricted Stock Award Grant Notice (the "Grant Notice") and the Agreement. If the Company uses an electronic capitalization system and the fields below are blank or the information is otherwise provided in a different format electronically, the blank fields and other information shall be deemed incorporated herein from the electronic capitalization system and considered part of this Grant Notice.

Participant:	[]
Grant Date:	[]
Total Number of Shares of Restricted Stock:	[] Shares
Vesting Commencement Date:	[]
Vesting Schedule:	[]
Termination:	If the Participant experiences a Termination of Service, any Shares that have not become ve on or prior to the date of such Termination of Service will thereupon be automatically forfe

ested ly forfeited of such Termination of Service will thereupon be automatical by the Participant, and the Participant's rights in such Shares shall thereupon lapse and expire.

By the Participant's signature below or by electronic acceptance or authentication in a form authorized by the Company, the Participant agrees to be bound by the terms and conditions of the Plan, the Agreement and this Grant Notice. The Participant has reviewed the Agreement, the Plan and this Grant Notice in their entirety, has had an opportunity to obtain the advice of counsel prior to the Participant's execution of this Grant Notice or electronic acceptance or authentication and fully understands all provisions of this Grant Notice, the Agreement and the Plan. The Participant hereby agrees to accept as binding, conclusive and final all decisions or interpretations of the Administrator upon any questions arising under the Plan, this Grant Notice or the Agreement. In addition, the Participant also agrees that the Company, in its sole discretion, may satisfy any Tax Related Items in accordance with Section 2.2(c) of the Agreement by (i) withholding shares of Common Stock otherwise issuable to the Participant upon vesting of the Shares, (ii) instructing a broker on the Participant's behalf to sell Shares upon vesting and submit the proceeds of such sale to the Company, or (iii) using any other method permitted by Section 2.2(c) of the Agreement or the Plan.

FRONTIER GROUP HOLDINGS, INC.: HOLDER:

PARTICIPANT:

By:	
Print Name:	
Title:	
Address:	

Bv: Print Name:

Address:

EXHIBIT A TO RESTRICTED STOCK AWARD GRANT NOTICE

RESTRICTED STOCK AWARD AGREEMENT

Pursuant to the Restricted Stock Award Grant Notice (the "<u>Grant Notice</u>") to which this Restricted Stock Award Agreement (this "<u>Agreement</u>") is attached, Frontier Group Holdings, Inc., a Delaware corporation, (the "<u>Company</u>") has granted to the Participant the number of shares of Restricted Stock (the "<u>Shares</u>") under the Company's 2021 Incentive Award Plan, as amended from time to time (the "<u>Plan</u>"), as set forth in the Grant Notice. Capitalized terms not specifically defined herein shall have the meanings specified in the Plan and Grant Notice.

ARTICLE I.

GENERAL

1.1 <u>Incorporation of Terms of Plan</u>. The Award (as defined below) is subject to the terms and conditions of the Plan, which are incorporated herein by reference. In the event of any inconsistency between the Plan and this Agreement, the terms of the Plan shall control.

ARTICLE II.

AWARD OF RESTRICTED STOCK

2.1 Award of Restricted Stock.

(a) <u>Award</u>. Pursuant to the Grant Notice and upon the terms and conditions set forth in the Plan and this Agreement, effective as of the Grant Date set forth in the Grant Notice, the Company has granted to the Participant an award of Restricted Stock (the "<u>Award</u>") under the Plan in consideration of the Participant's past and/or continued employment with or service to the Company or any Subsidiary, and for other good and valuable consideration. The number of Shares subject to the Award is set forth in the Grant Notice. The Participant is an Employee, Director or Consultant of the Company or one of its Subsidiaries.

(b) <u>Escrow</u>. The Participant, by acceptance of the Award, shall be deemed to appoint, and does so appoint, the Secretary of the Company or such other escrow holder as the Administrator may appoint to hold the Shares in escrow as the Participant's attorney(s)-in-fact to effect any transfer of unvested forfeited Shares (or Shares otherwise reacquired by the Company hereunder) to the Company as may be required pursuant to the Plan or this Agreement and to execute such documents as the Company or such representatives deem necessary or advisable in connection with any such transfer.

(c) <u>Removal of Notations</u>. As soon as administratively practicable after the vesting of any Shares subject to the Award pursuant to Section 2.2(b) hereof, the Company shall remove the notations on any Shares subject to the Award which have vested (or such lesser number of Shares as may be permitted pursuant to Section 10.7 of the Plan). The Participant (or the beneficiary or personal representative of the Participant in the event of the Participant's death or incapacity, as the case may be) shall deliver to the Company any representations or other documents or assurances required by the Company.

2.2 Restrictions.

(a) <u>Forfeiture</u>. Notwithstanding any contrary provision of this Agreement, except as may otherwise be provided by the Administrator or as set forth in a written agreement between the Company (or any Subsidiary that is the employer of the Participant) and the Participant, upon the Participant's Termination of Service for any or no reason, any Shares subject to Restrictions shall thereupon be forfeited immediately and without any further action by the Company, and the Participant's rights in such Shares shall thereupon lapse and expire.

(b) <u>Vesting and Lapse of Restrictions</u>. As of the Grant Date, one hundred percent (100%) of the Shares shall be subject to a risk of forfeiture set forth in Section 2.2(a) hereof and the transfer restrictions set forth in Section 3.3 hereof (collectively, such risk of forfeiture and such transfer restrictions, the "<u>Restrictions</u>"). The Award shall vest and Restrictions shall lapse in accordance with the vesting schedule set forth in the Grant Notice (rounding down to the nearest whole Share).

(c) <u>Tax Withholding</u>. As set forth in Section 10.5 of the Plan, the Company shall have the authority and the right to deduct or withhold, or to require the Participant to remit to the Company, an amount sufficient to satisfy all applicable Tax-Related Items required by law to be withheld with respect to any taxable event arising in connection with the Award. The Company shall not be obligated to transfer Shares held in escrow to the Participant or the Participant's legal representative until the Participant or the Participant's legal representative until the Participant or the Participant resulting from the grant or vesting of the Award or the issuance of Shares.

(d) <u>Stop Transfer Instructions</u>. To ensure compliance with the Restrictions, the provisions of the charter documents of the Company, and/or Applicable Law and for other proper purposes, the Company may issue appropriate "stop transfer" and other instructions to its transfer agent with respect to the Restricted Stock. The Company shall notify the transfer agent as and when the Restrictions lapse.

2.3 <u>Consideration to the Company</u>. In consideration of the grant of the Award pursuant hereto, the Participant agrees to render faithful and efficient services to the Company or any Subsidiary.

ARTICLE III.

OTHER PROVISIONS

3.1 <u>Section 83(b) Election</u>. If the Participant makes an election under Section 83(b) of the Code to be taxed with respect to the Restricted Stock as of the date of transfer of the Restricted Stock rather than as of the date or dates upon which the Participant would otherwise be taxable under Section 83(a) of the Code, the Participant hereby agrees to deliver a copy of such election to the Company promptly after filing such election with the Internal Revenue Service.

3.2 <u>Administration</u>. The Administrator shall have the power to interpret the Plan and this Agreement and to adopt such rules for the administration, interpretation and application of the Plan as are consistent therewith and to interpret, amend or revoke any such rules. All actions taken and all interpretations and determinations made by the Administrator in good faith shall be final and binding upon the Participant, the Company and all other interested persons. No member of the Administrator or the Board shall be personally liable for any action, determination or interpretation made in good faith with respect to the Plan, this Agreement or the Award.

3.3 <u>Transferability</u>. Until the Restrictions hereunder lapse or expire pursuant to this Agreement and the Shares vest, the Restricted Stock (including any Shares or other securities or property received by the Participant with respect to Restricted Stock as a result of stock dividends, stock splits or any other form of recapitalization) shall be subject to the restrictions on transferability set forth in Section 10.1 of the Plan.

3.4 <u>Rights as Stockholder</u>. Except as otherwise provided herein, upon the Grant Date, the Participant shall have all the rights of a stockholder of the Company with respect to the Shares, subject to the Restrictions, including, without limitation, voting rights and rights to receive any cash or stock dividends, in respect of the Shares subject to the Award and deliverable hereunder.

3.5 <u>Tax Consultation</u>. The Participant understands that the Participant may suffer adverse tax consequences in connection with the Restricted Stock granted pursuant to this Agreement (and the Shares issuable with respect thereto). The Participant represents that the Participant has consulted with any tax consultants the Participant deems advisable in connection with the Restricted Stock and that the Participant is not relying on the Company for any tax advice.

3.6 <u>Adjustments Upon Specified Events</u>. The Administrator may accelerate the vesting of the Restricted Stock in such circumstances as it, in its sole discretion, may determine. The Participant acknowledges that the Restricted Stock is subject to adjustment, modification and termination in certain events as provided in this Agreement and Article IX of the Plan.

3.7 <u>Notices</u>. Any notice to be given under the terms of this Agreement to the Company shall be addressed to the Company in care of the Secretary of the Company at the Company's principal office, and any notice to be given to the Participant shall be addressed to the Participant at the Participant's last address reflected on the Company's records. By a notice given pursuant to this Section 3.7, either party may hereafter designate a different address for notices to be given to that party. Any notice shall be deemed duly given when sent via email or when sent by certified mail (return receipt requested) and deposited (with postage prepaid) in a post office or branch post office regularly maintained by the United States Postal Service (or similar foreign entity).

3.8 <u>Participant's Representations</u>. If the Shares issuable hereunder have not been registered under the Securities Act or any applicable state laws on an effective registration statement at the time of such issuance, the Participant shall, if required by the Company, concurrently with such issuance, make such written representations as are deemed necessary or appropriate by the Company and/or its counsel.

3.9 Titles. Titles are provided herein for convenience only and are not to serve as a basis for interpretation or construction of this Agreement.

3.10 <u>Governing Law</u>. The laws of the State of Delaware shall govern the interpretation, validity, administration, enforcement and performance of the terms of this Agreement regardless of the law that might be applied under principles of conflicts of laws.

3.11 <u>Conformity to Securities Laws</u>. The Participant acknowledges that the Plan and this Agreement are intended to conform to the extent necessary with all provisions of the Securities Act and the Exchange Act, and any and all Applicable Law. Notwithstanding anything herein to the contrary, the Plan shall be administered, and the Award is granted, only in such a manner as to conform to such Applicable Law. To the extent permitted by Applicable Law, the Plan and this Agreement shall be deemed amended to the extent necessary to conform to such Applicable Law.

3.12 <u>Amendment, Suspension and Termination</u>. To the extent permitted by the Plan, this Agreement may be wholly or partially amended or otherwise modified, suspended or terminated at any time or from time to time by the Administrator or the Board; *provided, however*, that, except as may otherwise be provided by the Plan, no amendment, modification, suspension or termination of this Agreement shall adversely affect the Award in any material way without the prior written consent of the Participant.

3.13 <u>Successors and Assigns</u>. The Company or any Subsidiary may assign any of its rights under this Agreement to single or multiple assignees, and this Agreement shall inure to the benefit of the successors and assigns of the Company and its Subsidiaries. Subject to the restrictions on transfer set forth in Section 3.3 hereof, this Agreement shall be binding upon the Participant and his or her heirs, executors, administrators, successors and assigns.

3.14 Limitations Applicable to Section 16 Persons. Notwithstanding any other provision of the Plan or this Agreement, if the Participant is subject to Section 16 of the Exchange Act, then the Plan, the Award and this Agreement shall be subject to any additional limitations set forth in any applicable exemptive rule under Section 16 of the Exchange Act (including any amendment to Rule 16b-3 of the Exchange Act) that are requirements for the application of such exemptive rule. To the extent permitted by applicable law, this Agreement shall be deemed amended to the extent necessary to conform to such applicable exemptive rule.

3.15 Not a Contract of Service Relationship. Nothing in this Agreement or in the Plan shall confer upon the Participant any right to continue to serve as an Employee or other service provider of the Company or any of its Subsidiaries or shall interfere with or restrict in any way the rights of the Company and its Subsidiaries, which rights are hereby expressly reserved, to discharge or terminate the services of the Participant at any time for any reason whatsoever, with or without cause, except to the extent expressly provided otherwise by Applicable Law or in a written agreement between the Company (or any Subsidiary that is the employer of the Participant) or a Subsidiary and the Participant.

3.16 Entire Agreement. The Plan, the Grant Notice and this Agreement (including all Exhibits thereto, if any) constitute the entire agreement of the parties and supersede in their entirety all prior undertakings and agreements of the Company and its Subsidiaries and the Participant with respect to the subject matter hereof, provided that the Shares shall be subject to any accelerated vesting provisions in any written agreement between the Participant and the Company (or any Subsidiary that is the employer of the Participant) or a Company plan pursuant to which the Participant participates, in each case, in accordance with the terms therein.

3.17 Limitation on the Participant's Rights. Participation in the Plan confers no rights or interests other than as herein provided. This Agreement creates only a contractual obligation on the part of the Company as to amounts payable and shall not be construed as creating a trust. Neither the Plan nor any underlying program, in and of itself, has any assets. The Participant shall have only the rights of a general unsecured creditor of the Company and its Subsidiaries with respect to amounts credited and benefits payable, if any, with respect to the Shares issuable hereunder.

FRONTIER GROUP HOLDINGS, INC. 2021 INCENTIVE AWARD PLAN

RESTRICTED STOCK UNIT AWARD GRANT NOTICE

Frontier Group Holdings, Inc., a Delaware corporation, (the "<u>Company</u>"), pursuant to its 2021 Incentive Award Plan, as amended from time to time (the "<u>Plan</u>"), hereby grants to the holder listed below (the "<u>Participant</u>"), an award of restricted stock units ("<u>Restricted Stock Units</u>" or "<u>RSUs</u>"). Each vested Restricted Stock Unit represents the right to receive, in accordance with the Restricted Stock Unit Award Agreement attached hereto as <u>Exhibit A</u> (the "<u>Agreement</u>"), one share of Common Stock ("<u>Share</u>"). This award of Restricted Stock Units is subject to all of the terms and conditions set forth herein and in the Agreement and the Plan, each of which are incorporated herein by reference. Unless otherwise defined herein, the terms defined in the Plan shall have the same defined meanings in this Restricted Stock Unit Award Grant Notice (the "<u>Grant Notice</u>") and the Agreement. If the Company uses an electronic capitalization system and the fields below are blank or the information is otherwise provided in a different format electronically, the blank fields and other information shall be deemed incorporated herein from the electronic capitalization system and considered part of this Grant Notice.

Participant:	[]
Grant Date:	[]
Total Number of RSUs:	[]
Vesting Commencement Date:	[]
Vesting Schedule:	[]
Termination:	If the Participant experiences a Te the date of such Termination of Se

If the Participant experiences a Termination of Service, all RSUs that have not become vested on or prior to the date of such Termination of Service will thereupon be automatically forfeited by the Participant without payment of any consideration therefor.

By the Participant's signature below or by electronic acceptance or authentication in a form authorized by the Company, the Participant agrees to be bound by the terms and conditions of the Plan, the Agreement and this Grant Notice. The Participant has reviewed the Agreement, the Plan and this Grant Notice in their entirety, has had an opportunity to obtain the advice of counsel prior to the Participant's execution of this Grant Notice or electronic acceptance or authentication and fully understands all provisions of this Grant Notice, the Agreement and the Plan. The Participant hereby agrees to accept as binding, conclusive and final all decisions or interpretations of the Administrator upon any questions arising under the Plan, this Grant Notice or the Agreement. In addition, by signing below, the Participant also agrees that the Company, in its sole discretion, may satisfy any Tax Related Items in accordance with Section 2.6(b) of the Agreement by (i) withholding shares of Common Stock otherwise issuable to the Participant upon vesting of the RSUs, (ii) instructing a broker on the Participant's behalf to sell shares of Common Stock otherwise issuable to the Participant upon vesting of the RSUs and submit the proceeds of such sale to the Company, or (iii) using any other method permitted by Section 2.6(b) of the Agreement or the Plan.

FRONTIER GROUP HOLDINGS, INC.:

DADTI	CIPANT:
PARII	UIPANI:

y:	
rint Name:	
tle:	
ddress:	

By:	
Print Name:	

Address:

EXHIBIT A TO RESTRICTED STOCK UNIT AWARD GRANT NOTICE

RESTRICTED STOCK UNIT AWARD AGREEMENT

Pursuant to the Restricted Stock Unit Award Grant Notice (the "<u>Grant Notice</u>") to which this Restricted Stock Unit Award Agreement (this "<u>Agreement</u>") is attached, Frontier Group Holdings, Inc., a Delaware corporation (the "<u>Company</u>"), has granted to the Participant the number of restricted stock units ("<u>Restricted Stock Units</u>" or "<u>RSUs</u>") set forth in the Grant Notice under the Company's 2021 Incentive Award Plan, as amended from time to time (the "<u>Plan</u>"). Each Restricted Stock Unit represents the right to receive one share of Common Stock (a "<u>Share</u>") upon vesting. Capitalized terms not specifically defined herein shall have the meanings specified in the Plan and Grant Notice.

ARTICLE I.

GENERAL

1.1 <u>Incorporation of Terms of Plan</u>. The RSUs are subject to the terms and conditions of the Plan, which are incorporated herein by reference. In the event of any inconsistency between the Plan and this Agreement, the terms of the Plan shall control.

ARTICLE II.

GRANT OF RESTRICTED STOCK UNITS

2.1 <u>Grant of RSUs</u>. Pursuant to the Grant Notice and upon the terms and conditions set forth in the Plan and this Agreement, effective as of the Grant Date set forth in the Grant Notice, the Company hereby grants to the Participant an award of RSUs under the Plan in consideration of the Participant's past and/or continued employment with or service to the Company or any Subsidiaries and for other good and valuable consideration.

2.2 <u>Unsecured Obligation to RSUs</u>. Unless and until the RSUs have vested in the manner set forth in Article 2 hereof, the Participant will have no right to receive Common Stock under any such RSUs. Prior to actual payment of any vested RSUs, such RSUs will represent an unsecured obligation of the Company, payable (if at all) only from the general assets of the Company.

2.3 <u>Vesting Schedule</u>. Subject to Section 2.5 hereof, the RSUs shall vest and become nonforfeitable with respect to the applicable portion thereof according to the vesting schedule set forth in the Grant Notice (rounding down to the nearest whole Share).

2.4 <u>Consideration to the Company</u>. In consideration of the grant of the award of RSUs pursuant hereto, the Participant agrees to render faithful and efficient services to the Company or any Subsidiary.

2.5 <u>Forfeiture, Termination and Cancellation upon Termination of Service</u>. Notwithstanding any contrary provision of this Agreement or the Plan, upon the Participant's Termination of Service for any or no reason, all Restricted Stock Units which have not vested prior to or in connection with such Termination of Service shall thereupon automatically be forfeited, terminated and cancelled as of the applicable termination date without payment of any consideration by the Company, and the Participant, or the Participant's beneficiary or personal representative, as the case may be, shall have no further rights hereunder. No portion of the RSUs which has not become vested as of the date on which the Participant incurs a Termination of Service shall thereafter become vested, except as may otherwise be provided by the Administrator or as set forth in a written agreement between the Company (or any Subsidiary that is the employer of the Participant) and Participant.

2.6 Issuance of Common Stock upon Vesting.

(a) As soon as administratively practicable following the vesting of any Restricted Stock Units pursuant to Section 2.3 hereof, but in no event later than thirty (30) days after such vesting date (for the avoidance of doubt, this deadline is intended to comply with the "short term deferral" exemption from Section 409A of the Code), the Company shall deliver to the Participant (or any transferee permitted under Section 3.2 hereof) a number of Shares equal to the number of RSUs subject to this Award that vest on the applicable vesting date. Notwithstanding the foregoing, in the event Shares cannot be issued pursuant to Section 10.7 of the Plan, the Shares shall be issued pursuant to the preceding sentence as soon as administratively practicable after the Administrator determines that Shares can again be issued in accordance with such Section.

(b) As set forth in Section 10.5 of the Plan, the Company shall have the authority and the right to deduct or withhold, or to require the Participant to remit to the Company, an amount sufficient to satisfy all applicable Tax-Related Items required by law to be withheld with respect to any taxable event arising in connection with the Restricted Stock Units. The Company shall not be obligated to deliver any Shares to the Participant or the Participant's legal representative unless and until the Participant or the Participant's legal representative shall have paid or otherwise satisfied in full the amount of all Tax-Related Items applicable to the taxable income of the Participant resulting from the grant or vesting of the Restricted Stock Units or the issuance of Shares.

2.7 <u>Conditions to Delivery of Shares</u>. The Shares deliverable hereunder may be either previously authorized but unissued Shares, treasury Shares or issued Shares which have then been reacquired by the Company. Such Shares shall be fully paid and nonassessable. The Company shall not be required to issue Shares deliverable hereunder prior to fulfillment of the conditions set forth in Section 10.7 of the Plan.

2.8 <u>Rights as Stockholder</u>. The holder of the RSUs shall not be, nor have any of the rights or privileges of, a stockholder of the Company, including, without limitation, voting rights and rights to dividends, in respect of the RSUs and any Shares underlying the RSUs and deliverable hereunder unless and until such Shares shall have been issued by the Company and held of record by such holder (as evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company). No adjustment shall be made for a dividend or other right for which the record date is prior to the date the Shares are issued, except as provided in Article IX of the Plan.

ARTICLE III.

OTHER PROVISIONS

3.1 <u>Administration</u>. The Administrator shall have the power to interpret the Plan and this Agreement and to adopt such rules for the administration, interpretation and application of the Plan as are consistent therewith and to interpret, amend or revoke any such rules. All actions taken and all interpretations and determinations made by the Administrator in good faith shall be final and binding upon the Participant, the Company and all other interested persons. No member of the Administrator or the Board shall be personally liable for any action, determination or interpretation made in good faith with respect to the Plan, this Agreement or the RSUs.

3.2 Transferability. The RSUs shall be subject to the restrictions on transferability set forth in Section 10.1 of the Plan.

3.3 <u>Tax Consultation</u>. The Participant understands that the Participant may suffer adverse tax consequences in connection with the RSUs granted pursuant to this Agreement (and the Shares issuable with respect thereto). The Participant represents that the Participant has consulted with any tax consultants the Participant deems advisable in connection with the RSUs and the issuance of Shares with respect thereto and that the Participant is not relying on the Company for any tax advice.

3.4 <u>Binding Agreement</u>. Subject to the limitation on the transferability of the RSUs contained herein, this Agreement will be binding upon and inure to the benefit of the heirs, legatees, legal representatives, successors and assigns of the parties hereto.

3.5 <u>Adjustments Upon Specified Events</u>. The Administrator may accelerate the vesting of the RSUs in such circumstances as it, in its sole discretion, may determine. The Participant acknowledges that the RSUs are subject to adjustment, modification and termination in certain events as provided in this Agreement and Article IX of the Plan.

3.6 <u>Notices</u>. Any notice to be given under the terms of this Agreement to the Company shall be addressed to the Company in care of the Secretary of the Company at the Company's principal office, and any notice to be given to the Participant shall be addressed to the Participant at the Participant's last address reflected on the Company's records. By a notice given pursuant to this Section 3.6, either party may hereafter designate a different address for notices to be given to that party. Any notice shall be deemed duly given when sent via email or when sent by certified mail (return receipt requested) and deposited (with postage prepaid) in a post office or branch post office regularly maintained by the United States Postal Service (or similar foreign entity).

3.7 <u>Participant's Representations</u>. If the Shares issuable hereunder have not been registered under the Securities Act or any applicable state laws on an effective registration statement at the time of such issuance, the Participant shall, if required by the Company, concurrently with such issuance, make such written representations as are deemed necessary or appropriate by the Company and/or its counsel.

3.8 Titles. Titles are provided herein for convenience only and are not to serve as a basis for interpretation or construction of this Agreement.

3.9 <u>Governing Law</u>. The laws of the State of Delaware shall govern the interpretation, validity, administration, enforcement and performance of the terms of this Agreement regardless of the law that might be applied under principles of conflicts of laws.

3.10 <u>Conformity to Securities Laws</u>. The Participant acknowledges that the Plan and this Agreement are intended to conform to the extent necessary with all provisions of the Securities Act and the Exchange Act and any other Applicable Law. Notwithstanding anything herein to the contrary, the Plan shall be administered, and the RSUs are granted, only in such a manner as to conform to Applicable Law. To the extent permitted by Applicable Law, the Plan and this Agreement shall be deemed amended to the extent necessary to conform to such Applicable Law.

3.11 <u>Amendment, Suspension and Termination</u>. To the extent permitted by the Plan, this Agreement may be wholly or partially amended or otherwise modified, suspended or terminated at any time or from time to time by the Administrator or the Board; *provided, however*, that, except as may otherwise be provided by the Plan, no amendment, modification, suspension or termination of this Agreement shall adversely affect the RSUs in any material way without the prior written consent of the Participant.

3.12 <u>Successors and Assigns</u>. The Company may assign any of its rights under this Agreement to single or multiple assignees, and this Agreement shall inure to the benefit of the successors and assigns of the Company. Subject to the restrictions on transfer herein set forth in Section 3.2 hereof, this Agreement shall be binding upon the Participant and his or her heirs, executors, administrators, successors and assigns.

3.13 <u>Limitations Applicable to Section 16 Persons</u>. Notwithstanding any other provision of the Plan or this Agreement, if the Participant is subject to Section 16 of the Exchange Act, then the Plan, the RSUs and this Agreement shall be subject to any additional limitations set forth in any applicable exemptive rule under Section 16 of the Exchange Act (including any amendment to Rule 16b-3 of the Exchange Act) that are requirements for the application of such exemptive rule. To the extent permitted by Applicable Law, this Agreement shall be deemed amended to the extent necessary to conform to such applicable exemptive rule.

3.14 <u>Not a Contract of Service Relationship</u>. Nothing in this Agreement or in the Plan shall confer upon Participant any right to continue to serve as an employee or other service provider of the Company or any of its Subsidiaries or interfere with or restrict in any way with the right of the Company or any of its Subsidiaries, which rights are hereby expressly reserved, to discharge or to terminate for any reason whatsoever, with or without cause, the services of the Participant's at any time, except to the extent expressly provided otherwise by Applicable Law or in a written agreement between the Company or a Subsidiary and the Participant.

3.15 Entire Agreement. The Plan, the Grant Notice and this Agreement (including all Exhibits thereto, if any) constitute the entire agreement of the parties and supersede in their entirety all prior undertakings and agreements of the Company and the Participant with respect to the subject matter hereof, provided that the RSUs shall be subject to any accelerated vesting provisions in any written agreement between Participant and the Company (or any Subsidiary that is the employer of the Participant) or Company plan pursuant to which Participant participates, in each case, in accordance with the terms therein.

3.16 Section 409A. This Award is not intended to constitute "nonqualified deferred compensation" within the meaning of Section 409A of the Code (together with any Department of Treasury regulations and other interpretive guidance issued thereunder, including without limitation any such regulations or other guidance that may be issued after the date hereof, "Section 409A"). However, notwithstanding any other provision of the Plan, the Grant Notice or this Agreement, if at any time the Administrator determines that this Award (or any portion thereof) may be subject to Section 409A, the Administrator shall have the right in its sole discretion (without any obligation to do so or to indemnify Participant or any other person for failure to do so) to adopt such amendments to the Plan, the Grant Notice or this Agreement, or adopt other policies and procedures (including amendments, policies and procedures with retroactive effect), or take any other actions, as the Administrator determines are necessary or appropriate for this Award either to be exempt from the application of Section 409A or to comply with the requirements of Section 409A.

3.17 Limitation on Participant's Rights. Participation in the Plan confers no rights or interests other than as herein provided. This Agreement creates only a contractual obligation on the part of the Company as to amounts payable and shall not be construed as creating a trust. Neither the Plan nor any underlying program, in and of itself, has any assets. The Participant shall have only the rights of a general unsecured creditor of the Company and its Subsidiaries with respect to amounts credited and benefits payable, if any, with respect to the RSUs, and rights no greater than the right to receive the Common Stock as a general unsecured creditor with respect to RSUs, as and when payable hereunder.

FRONTIER GROUP HOLDINGS, INC.

NON-EMPLOYEE DIRECTOR COMPENSATION PROGRAM

Non-employee members of the board of directors (the "<u>Board</u>") of Frontier Group Holdings, Inc. (the "<u>Company</u>") shall be eligible to receive cash and equity compensation as set forth in this Non-Employee Director Compensation Program (this "<u>Program</u>"). The cash and equity compensation described in this Program shall be paid or be made, as applicable, automatically and without further action of the Board, to each member of the Board who is not an employee of the Company or any parent or subsidiary of the Company (each, a "<u>Non-Employee Director</u>") who may be eligible to receive such cash or equity compensation, unless such Non-Employee Director declines the receipt of such cash or equity compensation by written notice to the Company. This Program shall remain in effect until it is revised or rescinded by further action of the Board. This Program may be amended, modified or terminated by the Board at any time, without advance notice, in its sole discretion. The terms and conditions of this Program shall supersede any prior cash and/or equity compensation arrangements for service as a member of the Board between the Company and any of its Non-Employee Directors. This Program shall become effective on the date of the pricing of the initial public offering of Company common stock (the "<u>Effective Date</u>").

1. Cash Compensation.

(a) <u>Annual Retainers</u>. Each Non-Employee Director shall be eligible to receive an annual retainer of \$80,000 for service on the Board.

(b) Additional Annual Retainers. In addition, a Non-Employee Director shall receive the following annual retainers, as applicable:

(i) <u>Chair of the Audit Committee</u>. A Non-Employee Director serving as Chairperson of the Audit Committee shall receive an additional annual retainer of \$20,000 for such service.

(ii) <u>Chair of the Compensation Committee</u>. A Non-Employee Director serving as Chairperson of the Compensation Committee shall receive an additional annual retainer of \$15,000 for such service.

(iii) <u>Chair of the Nominating and Corporate Governance Committee</u>. A Non-Employee Director serving as Chairperson of the Nominating and Corporate Governance Committee shall receive an additional annual retainer of \$15,000 for such service.

(iv) <u>Chair of the Finance Committee</u>. A Non-Employee Director serving as Chairperson of the Finance Committee shall receive an additional annual retainer of \$15,000 for such service.

(c) <u>Payment of Retainers</u>. The annual retainers described in Sections 1(a) and 1(b) shall be earned on a quarterly basis based on a calendar quarter and shall be paid by the Company in arrears not later than the fifteenth day following the end of each calendar quarter. In the event a Non-Employee Director does not serve as a Non-Employee Director, or in the applicable positions described in Section 1(b), for an entire calendar quarter, the retainer paid to such Non-Employee Director shall be prorated for the portion of such calendar quarter actually served as a Non-Employee Director, or in such position, as applicable.

2. <u>Equity Compensation</u>. Non-Employee Directors shall be granted the equity awards described below. The awards described below shall be granted under and shall be subject to the terms and provisions of the Company's 2021 Incentive Award Plan or any other applicable Company equity incentive plan then-maintained by the Company (the "<u>Equity Plan</u>") and shall be evidenced by the execution and delivery of award agreements, including attached exhibits, in substantially the forms previously approved by the Board. All applicable terms of the Equity Plan apply to this Program as if fully set forth herein, and all grants of Restricted Stock Units hereby are subject in all respects to the terms of the Equity Plan.

(a) Initial Awards. Each person who is initially elected to the Board as a Non-Employee Director shall be granted, automatically and without necessity of any action by the Board or any committee thereof, on the date of such initial election Restricted Stock Units with respect to that number of shares of Company common stock (the "<u>Common Stock</u>") calculated by dividing (i) the product of (A) \$120,000 multiplied times (B) a fraction, the numerator of which is the number of days remaining until either (1) the first anniversary of the annual meeting of the Company's stockholders that immediately preceded such Non-Employee Director's election or appointment or (2) in the event such election or appointment occurs prior to the first annual meeting of the Company's stockholders that occurs after the Effective Date, the projected date of such first annual meeting and the denominator of which is 365, by (ii) the per share Fair Market Value (as defined in the Equity Plan) of the Common Stock as of the date of appointment or election and rounding down to the nearest whole number. The awards described in this Section 2(a) shall be referred to as "<u>Initial Awards</u>." No Non-Employee Director shall be granted more than one Initial Award.

(b) <u>Subsequent Awards</u>. On the date of each annual meeting of the Company's stockholders, each Non-Employee Director who will continue to serve as a Non-Employee Director immediately following such annual meeting shall be granted, automatically and without necessity of any action by the Board or any committee thereof, on the date of such annual meeting Restricted Stock Units with respect to that number of shares of Common Stock calculated by dividing (i) \$120,000 by (ii) the per share Fair Market Value of the Common Stock on the date of grant ("<u>Subsequent Award</u>"). For the avoidance of doubt, a Non-Employee Director elected for the first time to the Board at an annual meeting of the Company's stockholders shall only receive an Initial Award having a value of \$120,000 in connection with such election, and shall not receive any Subsequent Award on the date of such meeting.

(c) Terms of Awards Granted to Non-Employee Directors

(i) <u>Vesting</u>. Each Initial Award and each Subsequent Award shall vest in full on the earlier of (A) the first anniversary of the date of grant or (B) immediately prior to the next annual meeting of the Company's stockholders after the date of grant, subject to the Non-Employee Director continuing to provide services to the Company through such vesting date.

(ii) <u>Change in Control Acceleration</u>. All of a Non-Employee Director's Initial Awards and Subsequent Awards, and any other equitybased awards outstanding and held by the Non-Employee Director, shall vest and, if applicable, become exercisable and all restrictions thereon shall lapse with respect to one hundred percent (100%) of the shares subject thereto immediately prior to the occurrence of a Change in Control (as defined in the Equity Plan), to the extent outstanding at such time.

3. <u>Reimbursements</u>. The Company shall reimburse each Non-Employee Director for all reasonable, documented, out-of-pocket travel and other business expenses incurred by such Non-Employee Director in the performance of his or her duties to the Company in accordance with the Company's applicable expense reimbursement policies and procedures as in effect from time to time.

4. <u>Flight Benefits</u>. Each Non-Employee Director shall be eligible to receive flight benefits on Frontier Airlines in the form of a Universal Air Travel Plan, Inc. ("<u>UATP</u>") card made available once per 12-month period that provides for travel solely on Frontier Airlines in the amount of \$5,500, for each Non-Employee Director other than the Chairman of the Board, and \$13,750, for the Chairman of the Board, in each case, that must be used, if at all, within 12 months of the date the UATP card is issued.

* * * *

CODE OF ETHICS OF FRONTIER GROUP HOLDINGS, INC.

ADOPTED AS OF _____, 2021

INTRODUCTION

In accordance with the requirements of the Securities and Exchange Commission (the "<u>SEC</u>") and of the National Association of Securities Dealers Automated Quotations Stock Market ("<u>Nasdaq</u>") Listing Standards, the Board of Directors (the "<u>Board</u>") of Frontier Group Holdings, Inc., a Delaware corporation (the "<u>Company</u>"), has adopted this Code of Ethics (the "<u>Code</u>"). This Code of Ethics contains general guidelines for conducting the business of the Company, its subsidiaries and controlled affiliates (collectively, "<u>Frontier</u>") consistent with the highest standards of business ethics. To the extent our Code sets a higher standard than required by commercial practice or applicable laws, rules or regulations, Frontier adheres to these higher standards. It is your responsibility know and understand the expectations and requirements set forth in this code, and to meet those standards.

Low Fares Done Right

Frontier seeks to serve our passengers and stakeholders through our "Low Fares Done Right" philosophy. Safety is our top priority as we provide a dependable, positive customer experience. In addition, we obtain low unit costs as we offer reasonably priced fare options to our passengers, which in turn will drive growth throughout the organization. To achieve our results, we promote the right people with a culture of winning and professionalism. The standards of that professionalism and integrity are set forth in this Code.

Who Code Applies To

This Code applies to all of the members of our board of directors, our officers and other employees. We refer to all Frontier officers and other employees covered by this Code as "employees," unless the context otherwise requires. In this Code, we refer to our principal executive officer, principal financial officer principal accounting officer or controller, or persons performing similar functions, as our "principal financial officers." If any terms of this Code conflict with applicable law, benefit plan provisions, or collective bargaining agreements, then the law, provision or bargaining agreement will prevail.

Obtaining Help and Information about the Code

This Code is not meant to be a comprehensive rulebook and cannot address every situation you may face. If you feel uncomfortable about a situation or have any doubts about whether it is consistent with Frontier's ethical standards, seek help. We encourage you to contact your supervisor for help first. If your supervisor cannot answer your questions or if you do not feel comfortable contacting your supervisor, contact the Legal Department, the Vice President of Human Resources or Frontier's Whistleblower Hotline (the "Hotline").

The Hotline is available 24 hours a day, seven days a week. By using the Hotline, you may report suspected illegal or unethical activity while having the option to remain anonymous. The Hotline number is 1-844-263-9404. Employees can also report possible violations online through the URL link listed on Frontier's intranet site under the "My links" tab. The Hotline is administered by an independent third party. If you choose to report anonymously via the Hotline, please provide sufficient detail so we can address your concern. Refer to our "Whistleblower Policy" for additional information.

Duty to Report Violations of the Code

All employees and directors have a duty to report any known or suspected violation of this Code, including violations of the laws, rules, regulations or policies that apply to Frontier. If you know of or suspect a violation of this Code, immediately report the conduct to your supervisor, the Legal Department, or via the Hotline number or URL link. The appropriate management personnel will work with you (and your supervisor, as applicable) to investigate your concern. All reports of known or suspected violations of the law or this Code will be handled sensitively and with discretion. Your supervisor, the Legal Department and Frontier will protect your confidentiality to the extent possible, consistent with law and Frontier's need to investigate your concern. Reports may also be made anonymously.

Policy Against Retaliation

Frontier prohibits retaliation against any employee or director who, in good faith, seeks help or reports known or suspected violations. Any reprisal or retaliation against you because you, in good faith, sought help or filed a report will be subject to disciplinary action, including potential termination of employment.

Waivers of the Code

Before our directors, executive officers, other principal financial officers or employees, or an immediate family member of any such director, executive officer, other principal financial officer or employee, engages in any activity that would be otherwise prohibited by this Code, he or she is strongly encouraged to obtain a written waiver. Any waiver of this Code for our directors, executive officers or other principal financial officers, or an immediate family member of a director, executive officer or other principal financial officer, may be made only by our Board of Directors and such waiver, along with the reasons for granting such waiver, will be disclosed to the public as required by law. Waivers of this Code for other employees may be made only by Frontier's Chief Executive Officer or General Counsel and will be reported to our Audit Committee.

Accountability; Discipline for Violations of this Code

If your conduct as a representative of Frontier does not comply with this Code, it can result in serious consequences for you and Frontier. Frontier will subject any employee or director who violates this Code to appropriate discipline, which may include termination of employment or removal from the Board of Directors, as appropriate. This determination would be based upon the facts and circumstances of each particular situation. If you are accused of violating this Code, you will be given an opportunity to present your version of the events at issue prior to any determination of appropriate discipline. Employees and directors who violate applicable law may expose themselves to substantial civil damages, criminal fines and prison terms. Frontier may also face substantial fines and penalties and may incur damage to its reputation and standing in the community.

CONFLICTS OF INTEREST

Identifying Potential Conflicts of Interest

A conflict of interest can occur when an employee or director's private interest interferes, or appears to interfere, with the interests of Frontier as a whole. You should avoid any private interest that influences your ability to act in the interests of Frontier or that makes it difficult to perform your work objectively and effectively.

Employment of Relatives and Personal Relationships in the Workplace

The employment of relatives or individuals involved in a personal or dating relationship may cause potential or actual conflicts of interest or favoritism, or the appearance of a conflict or favoritism. Frontier recognizes that even the appearance of potential conflict or favoritism can damage credibility, impair department morale, and undermine an employee's effectiveness. Accordingly, Frontier does not permit employees who are related or involved in a dating or personal relationship to have direct supervisory authority over one another. Frontier considers "direct supervisory authority" to include situations in which one individual has direct influence over another individual's employment through decisions, recommendations, or judgment related to such matters as assessment of performance, work responsibilities, salary, career growth, or discipline.

Frontier permits the employment of relatives or individuals involved in a personal relationship if their employment does not, in the opinion of the Vice President of Human Resources and the Vice President of the applicable department, create an actual, potential, or perceived conflict of interest. If you have a relative seeking employment at Frontier or are involved in a dating or personal relationship with individuals seeking employment at Frontier that might violate this policy, you must disclose the relationship, in writing, to the Vice President of the applicable department and the Vice President of Human Resources.

For the purpose of this guideline, a relative is defined as spouse, domestic partner (with an affidavit on file), parent, grandparent, child, grandchild, brother, sister, brother-in-law, sister-in-law, mother-in-law, stepparent, stepporther, stepsister and stepchild.

Other Potential Conflicts

permitted.

A conflict of interest that interferes with your Frontier responsibilities and duties or is detrimental to Frontier's image or interest, is not

- <u>Outside Employment</u>. No employee should be engage in, by employed by, serve as a director of, or provide any services to a company that the employee knows or suspects is a supplier or competitor of Frontier. Independent consulting services that compete with Frontier are prohibited.
- <u>Improper Personal Benefits</u>. No employee should obtain or receive any material personal benefits or favors because of his or her position with Frontier. See "Gifts and Entertainment" below for additional guidelines.

- <u>Financial Interests</u>. No employee should have a significant financial interest (ownership or otherwise) in or with any company that the employee knows or suspects is a supplier or competitor of Frontier. A "significant financial interest" means: (i) ownership of greater than 1% of the equity of a supplier or competitor or (ii) an investment in a supplier or competitor that represents more than 5% of the total assets of the employee.
- <u>Service on Boards and Committees</u>. No employee should serve on a board of directors or trustees or on a committee of any entity (whether profit or not-for- profit) whose interests reasonably would be expected to conflict with those of Frontier.

Customer transactions conducted on an arms-length basis, on standard commercially available terms, are not considered conflicts, such as with banks, brokerage firms, other financial institutions, air carriers, travel agents or tour operators.

Disclosure of Conflicts of Interest

Employees and directors must disclose any situations that reasonably would be expected to give rise to a conflict of interest. If you suspect that you have a conflict of interest, or something that others could reasonably perceive as a conflict of interest, you must report it in writing to your supervisor, the Legal Department or the Vice President of Human Resources. The appropriate personnel will work with you to determine whether you have a conflict of interest and, if so, how best to address it. Although conflicts of interest are not automatically prohibited, they are not desirable and may only be waived as described in "Waivers of the Code" above.

CONFIDENTIAL INFORMATION

Employees and directors have access to a variety of confidential information regarding Frontier and our business. Confidential information includes all non-public information that might be of use to competitors, or, if disclosed, harmful to Frontier. Frontier considers all non-public information about Frontier, our business and our employees to be highly confidential. Employees have a duty to safeguard all confidential information of Frontier or third parties with which Frontier conducts business, except when disclosure is authorized or legally mandated. An employee's obligation to protect confidential information continues after he or she leaves Frontier. Unauthorized disclosure of confidential information could cause competitive harm to Frontier and could result in legal liability to you and Frontier.

Frontier information includes, but is not limited to operating and financial information; compensation data; marketing strategies; pending projects and proposals; and operations or maintenance manuals. Employee information includes, but is not limited to the following regarding any employee or that employee's dependents: names, addresses, phone numbers, social security numbers, dates of birth, medical or disability information, or any other personally identifiable information.

Employees who manage or use this type of information must protect this information from unauthorized modification, disclosure, and destruction. The means of protection used must be commensurate with the risk of exposure, the value of the information and the available computing resources. Employees are not authorized to gather, maintain, or disseminate personally identifiable information for any purpose or remove such information from Frontier's premises without written permission, as permitted by applicable law, except when required for a business purpose or in the ordinary daily course of performing duties on behalf of Frontier. Personally identifiable information should not be transmitted to anyone outside Frontier without proper safeguards to protect the data during transit. Personally identifiable information should not be stored on an unencrypted laptop or on CD, DVD, memory stick, or any other form of portable media.

Also see our "Social Media Usage Policy," in the Employee Handbook.

MEDIA CONTACTS AND OFFICIAL STATEMENTS ON BEHALF OF FRONTIER

Frontier employees and directors have access to a significant amount of information about Frontier. Frontier also values and works hard to maintain an appropriate image within the aviation, business, and consumer markets through positive relationships with the media. Accordingly, employees must obtain approval from the Corporate Communications department prior to speaking with the media about Frontier. This allows Frontier to maintain its public image, and provide a consistent and appropriate message to the media and general public. If you have any questions regarding media or external communications, please contact the Vice President of your department, the Vice President of Investor Communications and/or Corporate Communications at *CorpComm@flyfrontier.com*.

COMPETITION AND FAIR DEALING

All employees should endeavor to deal fairly with fellow employees and with Frontier's customers, suppliers and competitors. Employees should not take unfair advantage of anyone through manipulation, concealment, abuse of privileged information, misrepresentation of material facts or any other unfair-dealing practice.

Relationships with Customers

Our business success depends upon our ability to foster positive relationships with our customers. Frontier is committed to dealing with customers fairly, honestly and with integrity. Information you provide to customers should be accurate and complete to the best of your knowledge. Employees should not deliberately misrepresent information to customers.

Relationships with Suppliers

Frontier deals fairly and honestly with its suppliers. This means that our relationships with suppliers are based on value, price, quality, service and reputation, among other factors. Employees dealing with suppliers should carefully guard their objectivity. Specifically, no employee should accept or solicit any personal benefit from a supplier or potential supplier that might compromise, or appear to compromise, his or her objective assessment of the supplier's products and prices. Employees can give or accept promotional items of nominal value or moderately scaled entertainment within the limits of responsible and customary business practice. See "Gifts and Entertainment" below.

Relationships with Competitors

Frontier is committed to free and open competition in the marketplace. Employees should avoid actions that would be contrary to laws governing competitive practices in the marketplace, including federal and state antitrust laws. Such actions include misappropriation and/or misuse of a competitor's confidential information or making false statements about the competitor's business and business practices. For further discussion of appropriate and inappropriate business conduct with competitors, see "Compliance with Antitrust Laws" below.

CORPORATE OPPORTUNITIES

All employees and directors owe a duty to Frontier to advance the legitimate interests of Frontier when the opportunity to do so arises. You are prohibited from directly or indirectly (a) taking personally for themselves opportunities that are discovered through the use of Frontier property, information or positions; (b) using Frontier property, information or positions for personal gain; and (c) competing with Frontier.

GIFTS AND ENTERTAINMENT

The giving and receiving of gifts is a common business practice. Appropriate business gifts and entertainment are welcome courtesies designed to build relationships and understanding among business partners. Gifts and entertainment, however, should not compromise, or appear to compromise, your ability to make objective and fair business decisions.

It is your responsibility to use good judgment in this area. As a general rule, you may give or receive gifts or entertainment to or from customers or suppliers only if the gift or entertainment would not be viewed as an inducement to or reward for any particular business decision. The following specific examples may be helpful:

 <u>Meals and Entertainment</u>. You may occasionally accept or give meals, refreshments or other entertainment if the items are of reasonable value, a primary purpose of the meeting or attendance at the event is business related, and the expenses would be paid by Frontier as a reasonable business expense if not paid for by another party.

Entertainment of reasonable value may include food and tickets for sporting and cultural events if they are generally offered to other customers, suppliers or vendors.

- <u>Advertising and Promotional Materials</u>. You may occasionally accept or give advertising or promotional materials of nominal value.
- <u>Personal Gifts</u>. You may accept or give personal gifts of reasonable value related to recognized special occasions such as a graduation, promotion, new job, wedding, retirement or a holiday. A gift is acceptable if it is based on a family or personal relationship and unrelated to the business involved between the individuals.

<u>Gifts Rewarding Service or Accomplishment</u>. You may accept a gift from a civic, charitable or religious organization specifically related to your service or accomplishments.

You should make every effort to refuse or return a gift that is beyond these permissible guidelines. If it would be inappropriate to refuse a gift or you are unable to return a gift, you must promptly report the gift to your supervisor. Your supervisor will consult with the Legal Department, which may require you to donate the gift to an appropriate community organization.

If you conduct business in other countries, you must be particularly careful that gifts and entertainment are not construed as bribes, kickbacks or other improper payments. See "The U.S. Foreign Corrupt Practices Act" below. In addition, please see our "Anti-Corruption Compliance Policy."

Note: Gifts and entertainment may not be offered or exchanged under any circumstances to or with any U.S. federal, state or local government employees. For a more detailed discussion of special considerations applicable to dealing with federal, state and local governments, see "Interactions with the Government" below.

FRONTIER RECORDS

Accurate and reliable records are crucial to our business. Our records are the basis of our earnings statements, financial reports and many other aspects of our business and guide our business decision-making and strategic planning. Frontier records include financial records, personnel records, records relating to our development of services and products and all other records maintained in the ordinary course of our business.

All Frontier records must be complete, accurate and reliable in all material respects. Each employee and director must follow Frontier's document retention policy with respect to Company records within such employee's or director's control.

PROTECTION AND USE OF FRONTIER ASSETS

Employees should protect Frontier's assets and ensure their efficient use for legitimate business purposes only. Theft, carelessness and waste have a direct impact on Frontier profitability. The use of Frontier's funds or assets, whether or not for personal gain, for any unlawful or improper purpose is prohibited.

To ensure the protection and proper use of Frontier assets, each employee should:

- Exercise reasonable care to prevent theft, damage or misuse of Frontier property;
- Report the actual or suspected theft, damage or misuse of Frontier property to a supervisor;
- Use Frontier's electronic and telephone systems, written materials and other property primarily for business-related purposes; and

Use Frontier property only for legitimate business purposes, as authorized in connection with your job responsibilities.

Employees should be aware that Frontier property includes all data and communications transmitted or received to or by, or contained in, Frontier's electronic or telephonic systems. Frontier property also includes all written communications. Employees and other users of this property should have no expectation of privacy with respect to these communications and data. For more information, see our "Information Technology -Acceptable Use Policy," in the Employee Handbook.

ACCURACY OF PUBLIC COMMUNICATIONS AND FINANCIAL REPORTS

Frontier is subject to various securities and other laws, regulations and reporting obligations. Both federal law and our policies require that all public communications, including all reports and documents filed with or submitted to the SEC, must be full, fair, accurate, timely and understandable. Inaccurate, incomplete or untimely reporting will not be tolerated and can severely damage Frontier and result in legal liability.

To ensure we meet this standard, all employees and directors (to the extent they are involved in Frontier's disclosure processes) are required to maintain familiarity with the disclosure requirements, processes and procedures applicable to Frontier commensurate with their job positions. Such employees and directors are prohibited from knowingly misrepresenting, omitting or causing others to misrepresent or omit, material facts about Frontier to others, including Frontier's independent auditors, governmental regulators and self-regulatory organizations.

The Chief Financial Officer and other employees working in the Finance Department have a special responsibility to ensure that all of our financial disclosures are full, fair, accurate, timely and understandable. These employees must understand and strictly comply with generally accepted accounting principles and all standards, laws and regulations for accounting and financial reporting of transactions, estimates and forecasts.

Also see our "Whistleblower Policy."

COMPLIANCE WITH LAWS AND REGULATIONS

Each employee and director has an obligation to comply with all laws, rules and regulations applicable to Frontier's operations. These include, without limitation, laws covering bribery and kickbacks, the development, testing, approval, marketing and sale of our services, copyrights, trademarks and trade secrets, information privacy, insider trading, illegal political contributions, antitrust prohibitions, foreign corrupt practices, offering or receiving gratuities, environmental hazards, employment discrimination or harassment, occupational health and safety, false or misleading financial information or misuse of corporate assets. You are expected to understand and comply with all laws, rules and regulations that apply to your job position. If any doubt exists about whether a course of action is lawful, you should seek advice from your supervisor or the General Counsel.

Frontier employees with responsibilities in the areas governed by the FAA and other applicable laws and regulations must understand and comply with these laws and regulations. These employees are expected to have a thorough understanding of, and to comply with, the laws, regulations and other relevant standards applicable to their job positions. Frontier has developed standard operating procedures and provides regular training to aid employees in understanding and complying with the requirements of the FAA or the DOT.

Also see our "Whistleblower Policy" and "Anti-Corruption Compliance Policy."

INTERACTIONS WITH THE GOVERNMENT

Frontier may conduct business with the U.S., state and local governments and the governments of other countries. Frontier is committed to conducting its business with all governments and their representatives with the highest standards of business ethics and in compliance with all applicable laws and regulations, including the special requirements that apply to communications with governmental bodies that may have regulatory authority over our services and operations, such as the FAA, DOT, government contracts and government transactions. In your interactions with any government, you should:

- Be forthright and candid at all times. No employee or director should intentionally misstate or omit any material information from any written or oral communication with the government.
- Ensure that all required written submissions are made to the government and are timely, and that all written submissions, whether voluntary or required, satisfy applicable laws and regulations.
- Do not offer or exchange any gifts, gratuities or favors with, or pay for meals, entertainment, travel or other similar expenses for, government employees.

If your job responsibilities include interacting with the government, you are expected to understand and comply with the special laws, rules and regulations that apply to your job position as well as with any applicable standard operating procedures that Frontier has implemented. Also see our "Anti-Corruption Compliance Policy."

POLITICAL CONTRIBUTIONS AND ACTIVITIES

Frontier encourages its employees and directors to participate in the political process as individuals and on their own time. However, federal and state contribution and lobbying laws severely limit the contributions Frontier can make to political parties or candidates. It is Frontier policy that its funds or assets not be used to make a political contribution to any political party or candidate, unless prior approval has been given by Frontier's Chief Executive Officer or General Counsel.

The following guidelines are intended to ensure that any political activity you pursue complies with this policy:

- <u>Contribution of Funds</u>. You may contribute your personal funds to political parties or candidates.
- <u>Volunteer Activities</u>. You may participate in volunteer political activities during non-work time. You may not participate in political activities during working hours.
- <u>Use of Company Name</u>. When you participate in political affairs, you should be careful to make it clear that your views and actions are your own, and not made on behalf of Frontier. For instance, Frontier letterhead should not be used to send out personal letters in connection with political activities.

These guidelines are intended to ensure that any political activity you pursue is done voluntarily and with your own resources and time. Please contact the General Counsel if you have any questions about this policy; also see our "Anti-Corruption Compliance Policy."

COMPLIANCE WITH ANTITRUST LAWS

Antitrust laws of the United States and other countries are designed to protect consumers and competitors against unfair business practices and to promote and preserve competition. Our policy is to compete vigorously and ethically while complying with all antitrust, monopoly, competition or cartel laws in all countries, states or localities in which Frontier conducts business. Violations of antitrust laws may result in severe penalties against Frontier and its employees, including potentially substantial fines and criminal sanctions. You are expected to maintain basic familiarity with the antitrust principles applicable to your activities. The following is a summary of actions that are violations of applicable antitrust laws:

- <u>Price Fixing</u>. Frontier may not agree with its competitors to raise, lower or stabilize prices or any element of price, including discounts and credit terms.
- <u>Limitation of Supply</u>. Frontier may not agree with its competitors to limit or restrict the supply of its services.
- <u>Allocation of Business</u>. Frontier may not agree with its competitors to divide or allocate markets, territories or customers.
- <u>Monopolies</u>. Frontier may not engage in any behavior that can be construed as an attempt to monopolize.
- <u>Boycott</u>. Frontier may not agree with its competitors to refuse to sell or purchase products from third parties.

Meetings with Competitors

Employees should exercise caution in meetings with competitors. Any meeting with a competitor may give rise to the appearance of impropriety. You should try to meet with competitors in a closely monitored, controlled environment for a limited period of time. You should create and circulate agendas in advance of any such meetings, and the contents of your meeting should be fully documented.

If you participate in a meeting with a competitor in which any prohibited topics are broached, you should affirmatively end the discussion, and you should state your reasons for doing so. During meetings with competitors, avoid sharing or obtaining confidential information from the competitor. Also avoid statements that could be construed as unfair acts such as harassment, threats or interference with the competitors' existing contractual relationships.

Professional Organizations and Trade Associations

Employees should be cautious when attending meetings of professional organizations and trade associations at which competitors are present. Attending meetings of professional organizations and trade associations is legal and proper, if such meetings have a legitimate business purpose and are conducted in an open fashion, adhering to a proper agenda. At such meetings, you should not discuss the restricted topics listed above, Frontier's pricing policies or other competitive terms or any other proprietary, competitively sensitive information.

COMPLIANCE WITH INSIDER TRADING LAWS

In the event Frontier becomes a publicly-traded company, employees and directors would be prohibited from trading in Frontier's stock or other securities while in possession of material, non-public information about Frontier. Material information is information of such importance that it can be expected to affect the judgment of investors as to whether or not to buy, sell, or hold the securities in question. In addition, employees and directors would be prohibited from recommending, "tipping" or suggesting that anyone else buy or sell Frontier's stock or other securities on the basis of material, non-public information. Employees and directors who obtain material non-public information about another company in the course of their duties are prohibited from trading in the stock or securities of the other company while in possession of such information or "tipping" others to trade on the basis of such information. Violation of insider trading laws can result in severe fines and criminal penalties, as well as disciplinary action by Frontier, up to and including termination of employment.

The principal executive officer, principal financial officer principal accounting officer or controller (or persons performing similar functions) of Frontier are also required to promote compliance by all employees with the Code and to abide by our standards, policies and procedures.

Please contact the General Counsel if you have any questions about this policy; also see our "Insider Trading Compliance Policy" and, if applicable, "Section 16 Compliance Policy".

COMPLIANCE WITH REGULATION FD

In the event Frontier becomes a publicly-traded company, in connection with its public communications, Frontier would be required to comply with a rule under the federal securities laws referred to as Regulation FD. Regulation FD provides that, when Frontier discloses material, non-public information about Frontier to securities market professionals or stockholders (where it is reasonably foreseeable that the stockholders will trade on the information), Frontier must also disclose the information to the public. "Securities market professionals" generally include analysts, institutional investors and other investment advisors. Please contact the General Counsel if you have any questions about this policy; also see our "Guide for Corporate Disclosure (Regulation FD Policy)."

THE FOREIGN CORRUPT PRACTICES ACT

The Foreign Corrupt Practices Act prohibits Frontier and its employees, directors and agents from offering or giving money or any other item of value to win or retain business or to influence any act or decision of any government official, political party, candidate for political office or official of a public international organization. Stated more concisely, the FCPA prohibits the payment of bribes, kickback or other inducements to foreign officials. This prohibition also extends to payments to a sales representative or agent if there is reason to believe that the payment will be used indirectly for a prohibited payment to foreign officials. Violation of the FCPA can result in severe fines and criminal penalties, as well as disciplinary action by Frontier, up to and including termination of employment. For more information, see our "Anti-Corruption Compliance Policy."

ENVIRONMENT, HEALTH AND SAFETY

Frontier is committed to providing a safe and healthy working environment for its employees and to avoiding adverse impact and injury to the environment and the communities in which it does business. Frontier employees and directors must comply with all applicable environmental, health and safety laws, regulations, policies and Frontier standards. It is your responsibility to understand and comply with the laws, regulations and policies that are relevant to your job. Failure to comply with environmental, health and safety laws, regulations and policies can result in civil and criminal liability against you and Frontier, as well as disciplinary action by Frontier, up to and including termination of employment.

Environment

All Frontier employees and directors should strive to conserve resources and reduce waste and emissions through recycling and other energy conservation measures. You have a responsibility to promptly report any known or suspected violations of environmental laws or any events that may result in a discharge or emission of hazardous materials.

Health and Safety

Frontier is committed not only to comply with all relevant health and safety laws, but also to conduct business in a manner that protects the safety of its employees. All employees and directors are required to comply with all applicable health and safety laws, regulations and policies relevant to their positions.

EMPLOYMENT PRACTICES

Frontier pursues fair employment practices in every aspect of its business. The following is intended only to be a summary of certain of our employment policies. Copies of Frontier's detailed policies, including the Employee Handbook, are available on Frontier's intranet and from the Human Resources Department. Frontier employees must comply with all applicable labor and employment laws, including anti-discrimination laws and laws related to freedom of association and privacy. It is your responsibility to understand and comply with the laws, regulations and policies that are relevant to your job. Failure to comply with labor and employment laws can result in civil and criminal liability against you and Frontier, as well as disciplinary action by Frontier, up to and including termination of employment. For more information about Frontier's employment policies, including procedures for specific situations, please consult the Employee Handbook.

Harassment and Discrimination

Frontier is committed to providing equal opportunity and fair treatment to all individuals on the basis of merit, without discrimination because of race, color, religion, national origin, sex (including pregnancy), sexual orientation, age, disability, veteran status or other characteristic protected by law.

Alcohol and Drugs

All Frontier employees must comply strictly with Frontier policies and applicable law regarding the abuse of alcohol and the possession, sale and use of illegal (at the federal or state level) substances. It is your responsibility to understand and comply with the laws, regulations and policies that are relevant to your job. Failure to comply with drug and alcohol laws and regulations can result in civil and criminal liability against you and Frontier, as well as disciplinary action by Frontier, up to and including termination of employment. Also see our "Drug and Alcohol Free Workplace" policy in the Employee Handbook.

Violence Prevention and Weapons

The safety and security of Frontier employees is vitally important. Frontier will not tolerate violence or threats of violence in, or related to, the workplace. For more information, including procedures for specific situations, please consult the Employee Handbook.

CONCLUSION

This Code contains certain fundamental principles, policies and procedures that govern the Frontier employees and directors in the conduct of the Frontier's business consistent with the highest standards of business ethics. It is not intended to and does not create any rights in any employee, customer, client, visitor, supplier, competitor, shareholder or any other person or entity. It is Frontier's belief that this Code is robust and covers most conceivable situations. If you have any questions about these guidelines, please contact your supervisor or the General Counsel. Frontier expects all of its employees and directors to adhere to these standards.

This Code, as applied to Frontier's principal financial officers, shall be our "code of ethics" within the meaning of Section 406 of the Sarbanes-Oxley Act of 2002 and the rules promulgated thereunder.

This Code and the matters contained herein are neither a contract of employment nor a guarantee of continuing Frontier policy. Frontier reserves the right to amend, supplement or discontinue this Code and the matters addressed herein, without prior notice, at any time.

* * * * *

Acknowledgment of Receipt of Compliance

Return By: _____, 20____

To: General Counsel

From:

Re: Frontier Group Holdings, Inc. Code of Ethics

I have received, reviewed, and understand the above-referenced Code of Ethics and hereby undertake, as a condition to my present and continued employment at Frontier Group Holdings, Inc., to comply fully with the policies and procedures contained therein.

Signature

Date

Name

Title